# DISSERTATION

# MODELING SOIL ORGANIC MATTER: THEORY, DEVELOPMENT, AND APPLICATIONS IN BIOENERGY

# **CROPPING SYSTEMS**

Submitted by

Eleanor Elizabeth Campbell

Graduate Degree Program in Ecology

In partial fulfillment of the requirements

For the Degree of Doctor of Philosophy

Colorado State University

Fort Collins, Colorado

Summer 2015

Doctoral Committee:

Advisor: Keith Paustian

William J. Parton M. Francesca Cotrufo Kenneth F. Reardon Copyright by Eleanor Elizabeth Campbell 2015

All Rights Reserved

#### ABSTRACT

# MODELING SOIL ORGANIC MATTER: THEORY, DEVELOPMENT, AND APPLICATIONS IN BIOENERGY CROPPING SYSTEMS

Soil organic matter (SOM) is a complex, dynamic, and highly variable soil constituent that is of fundamental importance to many soil functions, terrestrial ecosystem processes, and biogeochemical cycles. Its importance extends across scales, ranging from site-specific impacts on soil fertility to the global net exchange of carbon between terrestrial systems and the atmosphere. Soil organic matter is impacted by human activities, as seen most directly in agricultural systems. In this context, SOM models play an important role in integrating the understanding of complex, interacting soil processes across temporal and spatial scales, contributing to land use decision making by providing comparative evaluation of soil impacts associated with different management practices. Crop-based bioenergy feedstock productions systems are an emerging area for these types of SOM model applications. However, model evaluations are dependent on the theoretical basis of a given SOM model, as well as the quality of data used to drive the model for a given system or management scenario. This study therefore explores linkages between advances in the theoretical understanding of SOM dynamics, the development of SOM models to reflect these advances, and the application of SOM models to assess crop-based bioenergy production systems.

First, five emerging areas in SOM research were reviewed in the context of SOM models, including SOM stabilization mechanisms, saturation kinetics, temperature sensitivity, dynamics in deep soils, and incorporation into earth system models. These reviews demonstrated the importance of identifying where SOM model development and applications are most limited, whether in theoretical understanding, in model implementation, or in data availability. For example, SOM saturation kinetics is

ii

theoretically well understood but remains difficult to implement in SOM models, only yielding improvements in a narrow set of ecological conditions. SOM temperature sensitivity and deep soil dynamics, however, are more limited by poor data availability in addition to poor theoretical understanding of interacting processes.

A selection of shortfalls in SOM modeling were then addressed and explored with the Litter Decomposition and Leaching (LIDEL) model, a litter decomposition model that incorporates dynamic microbial carbon use efficiency (CUE) and yields dissolved organic carbon (DOC) as one of the byproducts of litter decomposition. In this analysis a hierarchical Bayesian statistical approach was used to test model performance and estimate unknown model parameters using experimental data. While this analysis showed the LIDEL model successfully integrates hypotheses for litter nitrogen and lignin controls on dynamic microbial CUE and the generation of DOC from litter decomposition, there remains a great deal of uncertainty in the rate of microbial biomass turnover as well as the proportioning of biomass from microbial turnover between solid versus soluble microbial products. Targeted experimental evaluation of the generation of DOC from microbes versus litter would support greater certainty in these model parameters and further model development for more general applications.

Finally, the performance of the DAYCENT ecosystem model was evaluated in simulating US corn residue removal and Brazilian sugarcane production, two types of crop-based bioenergy feedstocks. DAYCENT is a process-based ecosystem model that integrates a soil organic carbon model to simulate carbon and nitrogen cycling processes through plant-soil interactions. The results of DAYCENT corn residue removal simulations highlighted several DAYCENT model biases, such as low corn yield estimates in dry regions and an overestimation of soil carbon loss with conventional tillage. Despite these biases, the results showed the importance of considering interactive effects between corn residue removal and other crop management practices in this type of bioenergy feedstock production system. The results suggest corn residue removal is ideally paired with management practices—such as reduced tillage—to

iii

maintain or improve soil carbon stocks. The analysis of Brazilian sugarcane management practices also highlighted management practices poorly simulated by DAYCENT, in particular identifying the need to improve DAYCENT simulations of high N<sub>2</sub>O emission conditions observed in mechanically-harvested sugarcane, perhaps by adding simulation of DOC movement across the soil profile. However, this analysis also identified a need for more accurate and consistent daily precipitation data to drive DAYCENT simulations of N<sub>2</sub>O emissions from Brazilian sugarcane management practices, particularly as there is interest in regionally-scaled analyses of direct greenhouse gas emissions from sugarcane production in Brazil.

Taken together, the results of this study show the importance of a close connection between emerging areas in SOM theory, SOM model developments, and SOM model applications in crop-based bioenergy feedstock production systems. This connection allows for the identification of specific areas in need of further research, whether developing new modeling approaches or gathering additional data to parameterize, drive, and evaluate model simulations. This connection should remain a central emphasis as SOM models are increasingly incorporated into crop-based bioenergy policy and land management decision making.

#### ACKNOWLEDGEMENTS

I would first like to recognize the guidance of my advisor, Dr. Keith Paustian. His support and mentorship had an enormous influence on every aspect of my doctoral work. I also greatly valued the opportunity to collaborate with and receive advice from my committee members, Dr. Francesca Cotrufo, Dr. William Parton, and Dr. Kenneth Reardon. They challenged me and pushed me to explore new ideas.

The graduate students, researchers and faculty within NREL were a great resource for support, advice, friendship, and inspiration. I could not have done the computational side of my research without the help of Ty Boyack and Mark Easter. John Field was an important source of many great discussions. Steve Williams, Cindy Keough, Melannie Hartman were gracious enough to answer my many questions. I appreciated being a part of shared efforts with Jeff Kent, Yao Zhang, Ram Gurung, Trung Nguyen, Amy Swan, Ernie Marx, and the rest of the Paustian group. Barbara Fricks, Ian Blackburn, Beth Wittmann, and Nycole Echeverria provided invaluable research assistance. Many thanks to Jocelyn Lavallee and Jacklyn Hatala Mathes for comments on sections of the review chapter. Funding for the SOM modeling review chapter, the LIDEL project, and the Brazilian sugarcane project was provided by Shell Global Solutions (US), Inc. The LIDEL project was also supported by NSF DEB # 0918482, NSF-GRFP, NSF RET- Kathleen Guilbert, and NSF DEB #1256754. Funding for the corn stover project was provided by the USDA-ARS (part of the USDA-ARS-Renewable Energy Assessment Project (REAP), renamed USDA-ARS-Resilient Economic Agricultural Practices (REAP)), and by the North Central Regional SunGrant Center at SD State University through the US DOE – Office of Biomass Programs, award number DE-FC36-05G085041.

Finally, I had the benefit of a wonderful network of friends and family who were an important part of this effort. Katie Renwick and Kelly Hopping kept me motivated to write. Ryan Garcia kept me convinced that I could always find a way. Sarah Fulton-Smith kept my spirits up. My husband, Tom Ebert, was the daily enduring support that kept me sustained through to the end.

# TABLE OF CONTENTS

ABST	TRACT	ii
ACKI	NOWLEDGEMENTS	v
1 II	INTRODUCTION	1
2 S	SOIL ORGANIC MATTER MODELING- PAST APPROACHES AND NEW CHALLENGES	4
2.	.1 Introduction	4
2.	.2 SOM modeling, past and present	7
	2.2.1 Basic modeling concepts and terminology	7
	2.2.2 History overview of SOM modeling	8
	2.2.3 SOM modeling approaches	10
	2.2.3.1 First order kinetics – dominant multi-compartment models	10
	2.2.3.2 Organism-oriented and non-1st order kinetic	11
	2.2.3.3 Conceptual vs. measurable SOM pools	12
	2.2.3.4 Data-model integration in SOM research	14
2.	.3 New directions in SOM model development	16
	2.3.1 Recent ideas for SOM stabilization mechanisms	16
	2.3.2 First order model versus saturation kinetics	25
	2.3.3 Temperature controls on decomposition	29
	2.3.4 Deep soil organic matter dynamics	
	2.3.5 Representation of SOM in earth system models (ESMs)	46

	2.4 Conclusions	52
3	TRACKING THE FATE OF LITTER C INTO CO $_2$ , MICROBIAL PRODUCTS AND DISSOLVED ORGANIC	
C/	ARBON BY THE LITTER DECOMPOSITION AND LEACHING (LIDEL) MODEL USING A HIERARCHICHAL	
B	AYESIAN APPROACH TO ESTIMATE PARAMETERS AND VARIABILITY	63
	3.1 Introduction	63
	3.2 Material and methods	67
	3.2.1 LIDEL model description	67
	3.2.1.1 Linking hypotheses for microbial CUE and litter decay	69
	3.2.1.2 Hypotheses from experimental data: DOC versus mass loss	71
	3.2.1.3 LIDEL initial conditions	73
	3.2.2 Statistical Approach	74
	3.2.2.1 Data	74
	3.2.2.2 Hierarchical Bayesian Analyses	75
	3.3 Results	77
	3.4 Discussion	79
	3.5 Conclusions	82
4	ASSESSING THE SOIL CARBON, BIOMASS PRODUCTION, AND NITROUS OXIDE EMISSION IMPACT (	ЭF
C	DRN STOVER MANAGEMENT FOR BIONERGY FEEDSTOCK PRODUCTION USING DAYCENT	93
	4.1 Introduction	93
	4.2 Materials and Methods	97
	4.2.1 Experimental Data	97

	4.2.2 DAYCENT Model Overview	97
	4.2.3 Statistical Analyses	99
	4.3 Results	100
	4.3.1 Biomass	100
	4.3.2 Soil C Change	100
	4.3.3 N <sub>2</sub> O emission	101
	4.4 Discussion	102
	4.4.1 Biomass	102
	4.4.2 Soil C change	103
	4.4.3 N <sub>2</sub> O emission	105
,	4.5 Conclusion	108
5	EVALUATING DAYCENT MODEL PERFORMANCE ESTIMATING SOIL CARBON AND NITROUS OXIDI	E
ΕN	IISSIONS IN SOUTH CENTRAL BRAZILIAN SUGARCANE PRODUCTION SYSTEMS	118
	5.1 Introduction	118
	5.2 Materials and Methods	121
	5.2.1 Experimental Data	121
	5.2.2 DAYCENT model overview	123
	5.2.3 Climate data modifications for simulating $N_2O$ emissions	125
	5.2.4 Statistical Analyses	126
	5.3 Results	127
	5.4 Discussion	130

6	CONCLUSION	146
RE	FERENCES	149
A	PPENDIX 1	191
	A1.1 Definitions	191
	A1.2 Data models	192
	A1.3 Process models	193
	A1.4 Parameter models	193
	A1.5 Monte Carlo Markov chain algorithm	195
A	PPENDIX 2	199
	A2.1 Definitions	199
	A2.2 Data models	200
	A2.3 Parameter models	200
	A2.4 Monte Carlo Markov chain algorithm	201

#### **1** INTRODUCTION

Soil organic matter (SOM) is highly complex, often mysterious, and offers endless challenges to the understanding of its underlying dynamics. Soil organic matter also plays a fundamental role in natural processes across scales. These scales range from soil microsites where water, organic matter, nutrients, and oxygen, and living organisms interact to determine the fate of litter inputs (Davidson et al., 2014), to globally-scaled processes such as carbon (C) cycling between terrestrial systems and the atmosphere (Heimann and Reichstein, 2008). Soil organic matter is therefore an important consideration in land management, both in sustaining its value as a resource—e.g. supporting productive soils used for agricultural production—and in mitigating direct and indirect ecosystem impacts of human activities—e.g. the potential for loss of soil C due to land use conversion from native ecosystems into agricultural production (Searchinger et al., 2008). The understanding of SOM dynamics is therefore often linked to efforts to soil and ecosystem management, whether to improve soil health and productivity or to preserve key functions such as C storage.

In this context, SOM models play a key role in integrating understanding of complex, interacting soil processes across scales. Soil organic matter model development has continually advanced, with recent emphasis on testing hypotheses for the impacts of climate change on SOM dynamics and ecosystem processes. The climate impact assessment of crop-based bioenergy production systems is an example of an area where SOM models are seeing increasing application, in the context of greenhouse gas emission management and the development of sustainable land management practices. In these applications, SOM models provide a tool to project across temporal and spatial areas, contributing to decision making by providing comparative evaluation of GHG impacts associated with crop management practices. However the validity of these model evaluations are only as good as their basis in the scientific understanding of SOM dynamics.

The first part of this work, presented in chapter two, provides an extensive review of SOM modeling, discussing past developments, history, and giving an overview of the 'toolbox' common model formulations. Chapter two also provides a series of five short reviews on rapidly advancing areas in SOM modeling. These short reviews are on the topics of 1) recent ideas in SOM stabilization mechanisms, 2) saturation kinetics, 3) temperature controls on decomposition, 4) deep SOM dynamics, and finally concludes with 5) a discussion of SOM representation in earth system models.

Looking to address some of the shortfalls in SOM modeling discussed in chapter two, chapter three presents the Litter Decomposition and Leaching (LIDEL) model, a new modeling approach to litter decomposition that incorporates dynamic microbial carbon use efficiency and yields dissolved organic carbon (DOC) as one of the byproducts of litter decomposition. Other SOM models have been criticized for poorly accommodating either of these important components of litter decomposition. The LIDEL model provides a first step to a new approach to litter decomposition that would strongly advance simulation of DOC dynamics through the soil profile, recognized as an important component of deep soil C processes (Rumpel and Kögel-Knabner, 2010). Chapter three additionally presents a hierarchical Bayesian analysis for data-model integration, using this approach to estimate unknown LIDEL model parameters and compare the performance of four different LIDEL model formulations. This analytical approach accommodates the high variability and uncertainty in both measured data and model formulation that are commonly characteristic in SOM research.

The final two chapters then step out to a larger scale to explore the use of the DAYCENT model—a full ecosystem model that integrates the CENTURY soil organic carbon model to simulate C and N cycling processes in plant-soil interactions (Parton, 1987)—in simulating two types of crop-based bioenergy feedstocks. Chapter four provides a validation analysis of DAYCENT performance in simulating corn stover harvest for cellulosic ethanol in the United States. Chapter five then presents a

parameterization and validation analysis of DAYCENT model performance in South Central Brazilian sugarcane production systems.

Taken together, these four chapters contribute towards a more integrated connection between emerging areas in SOM theory, SOM model developments, and SOM model applications in crop-based bioenergy feedstock production systems. This connection is an important area of consideration as SOM models are increasingly incorporated into crop-based bioenergy policy and land management decision making. This study aims to demonstrate how connecting across SOM theory, SOM model development, and SOM model applications can allow for better identification of specific areas in need of further research, whether developing new modeling approaches or gathering additional data to parameterize, drive, and evaluate model simulations.

### 2 SOIL ORGANIC MATTER MODELING- PAST APPROACHES AND NEW CHALLENGES<sup>1</sup>

## 2.1 Introduction

Many discussions of soil organic matter (SOM) research begin with a statement to the effect that soils are complex, challenging to study, and do not easily part with their mysteries. A new favorite term is 'the Gemish effect' (Davidson et al., 2014), referring to the challenge of identifying fundamental drivers in the hodgepodge of soil minerals, organisms, and organic matter cycling dynamically through space and time in the belowground environment. Apart from being recognized as the "most complicated biomaterial on the planet" (Young & Crawford, 2004, pg 1634), SOM is also a key component of ecosystem processes and biogeochemical cycles, notably including the responses to and sustainability of human impacts on the earth. The study of SOM further requires wrestling with the logistical challenges of soil opacity and observer effects—where what is observed is changed by the process of observation as well as the conceptual challenges of integrating interactions between multiple factors across scales. It is with the latter that simulation models for SOM dynamics play a crucial role in SOM research.

In soil science and ecology, simulation models serve many purposes ranging from mathematical formalization of hypotheses to projecting processes across space and time. Models are meant to provide a useful simplification of reality. Yet even after over a century of study there are few hard rules in the understanding SOM dynamics. Some of this is driven by the logistical challenges of measuring soils. Soils are difficult to characterize in ways that account for their complexity and heterogeneity across systems, scales, and depths. SOM model testing is therefore routinely data-poor, particularly across large scales and long periods of time. However other uncertainties arise from using measurements of operationally defined (e.g., by mesh size, chemical extraction, density) soil fractions to evaluate and compare SOM

<sup>&</sup>lt;sup>1</sup> In preparation with K. Paustian for a chapter titled "Simulation Modeling" in Methods of Soil Analysis: Soil Organic Matter, edited by R.F. Turco.

dynamics. Soil fractionation methods are recognized to have variable linkages to mechanisms that affect SOM and drive its ecological behavior (von Lützow et al., 2007; Wander, 2004). Bulk soil dynamics are not easily reconstructed from measurement of its parts.

Simulation models of SOM therefore play a key role in SOM research and application by supporting hypothesis testing and predictions that are often infeasible by direct measurements, and more importantly, by providing an explicit mathematical framework for considering the structure and dynamics of SOM. The linkage between SOM measurements and conceptual understanding of SOM dynamics is accordingly at the foundation of SOM model structure. For example, some SOM models are formulated with conceptual, kinetically-defined SOM pools, that are not directly measurable per se, but have been proven through decades of testing as general models that work reasonably well to simulate bulk soil dynamics, particularly over the long-term (e.g., Jenkinson & Rayner, 1977; Parton et al., 1987). Other emerging SOM models take different approaches to SOM dynamics—e.g. defining SOM pools by specific SOM mechanisms or as analytically measureable SOM fractions. However, many of these models currently remain more limited in their applications, either due to simplification of factors that are important at larger scales, or due to limited data availability (Davidson et al., 2014; Segoli et al., 2013; Tipping et al., 2012). Ideally, a SOM model would be based on mechanistic understanding of SOM dynamics, use SOM pools that can be informed by measured data, and apply across multiple scales. At this point in time, however, no single SOM model yet fits this ideal and indeed inherent tradeoffs between model attributes (e.g., generality, predictive capacity, complexity) relative to their intended purpose suggest that no such ideal model can exist (Levins, 1966; Sharpe, 1990).

Jenny's now classic factors of soil formation, informed by even earlier work (Hilgard, 1906; Dokuchaev, 1883), provides a useful framework to examine the complexities SOM dynamics, presenting the interaction of soil forming factors as;

$$S = f(cl, o, r, p, t, ...)$$
Eqn 1

such that a particular soil attribute (S) is a function of climate (cl), organisms (o), topography (r), parent material (p), time (t) and other factors that often include human activities (Bidwell and Hole, 1965; Jenny, 1941). This approach has formed the conceptual underpinning of many process-based soil and ecosystem models (Vitousek, 1994). We argue that it continues to act as an overarching framework in SOM modeling, providing a common thread to evaluate how current hypotheses for SOM dynamics are reflected in SOM models applied across scales.

Development of SOM models moves hand-in-hand with conceptual understanding of SOM dynamics and their linkages to SOM measurements. In recent years, new hypotheses for SOM persistence have led to reformulation of SOM models to explore explicit simulation of microbial biomass and more mechanistically-based microbial decomposition mechanisms. Researchers have also examined the mechanism of SOM saturation as a limitation on soil carbon storage. Expanding efforts to aggregate SOM data and complete large-scale multi-model testing are also pushing SOM model development forward. These efforts have been driven in part by the potential role of SOM as a sink or source of atmospheric CO<sub>2</sub>, with implications for global climate change. Temperature sensitivity of SOM decomposition with warming climates has consequently been brought under greater scrutiny, alongside increased interest in simulation of SOM dynamics in deep soil layers. Global and earth system models, which are now incorporating SOM models of increasing complexity, are an area for soil scientists to both evaluate SOM hypotheses at a global scale, as well as support better model development and accuracy through model-data integration.

In this review we first provide some background and history of SOM model development, as well as brief summaries of modeling approaches and their linkages to SOM measurements and the conceptual understanding of SOM dynamics (Section 2.2). We will then review a selection of recent advances in SOM modeling that we believe are particularly relevant to ongoing SOM research. These include: new ideas in SOM stabilization, modeling SOM saturation kinetics, temperature controls on

decomposition, SOM dynamics in deep soil layers, and finally a review of SOM representation in earth system models (ESMs) (Section 2.3).

# 2.2 SOM modeling, past and present

#### 2.2.1 Basic modeling concepts and terminology

There are several general differences in approaches to mathematical modeling. The first is the difference between an **empirical** versus a **mechanistic** model. An empirical model is data-driven, making predictions based on observations of a relationship or interaction, without explicitly defining the underlying causal mechanisms. A mechanistic model, on the other hand, is based on a representation of mechanisms that drive dynamics in a system, using a 'first principles' approach (i.e. starting with the most basic assumptions) to build a predictive model. The latter is often a more defensible foundation for an explanatory model, but is hard to come by in ecology and biogeochemistry where system dynamics typically result from complex, heterogeneous, interactive factors that are often not fully understood.

Whether empirically- or mechanistically-based, models can be **static** or **dynamic**, the former referring to a model of a system at a fixed point or interval of time while the latter explicitly includes changes through time. Models can also be either **stochastic** or **deterministic**. Stochastic models include random variation in model simulations, while deterministic models do not. In a deterministic model the same initial conditions and model parameters will always generate the same result, while stochastic model results will have some degree of random variability. Many processes in ecology and biogeochemistry are variable and change through time. Stochastic and dynamic models are often a closer representation of these types of systems. However, stochastic and dynamic models are also typically more complex and computationally intensive, which may limit their use. Models are ideally as simple as possible, while still serving as an effective representation of reality.

As commonly used terminology, a model **pool** refers to a quantity of material. For the purposes of this review model pools are generally quantities of OM or C. A **flux** is the rate at which material

moves from one pool to another or into a larger **sink** or **source** external to the system being simulated. A **parameter** controls the dynamics of the system and can either be a constant or variable, depending on whether it is a set value or informed by other internal (e.g. pools or fluxes) or external (e.g. driving variables) components of the system. Figure 2-1 gives a visual representation of a set of dynamic, deterministic models with structures commonly used in SOM models, representing pools as boxes and fluxes as arrows.

#### 2.2.2 History overview of SOM modeling

The origins of mathematical modeling of SOM processes, we would argue, largely stem from two main areas of inquiry. First, several early models were developed to describe the rate of mass loss of plant residues and the relationship between plant residue inputs and soil organic matter levels (Hénin and Depuis, 1945; Jenny et al., 1949; Salter and Green, 1933), including the observed loss of SOM following conversion of native ecosystems to cropland (Jenny, 1941). These were formulated as single pool models with first-order decay kinetics for mass loss, vis,

$$\frac{dC}{dt} = I - kC$$
 Eqn 2

at equilibrium(C<sup>\*</sup>),  $\frac{dC}{dt} = 0$ , then,

$$C^* = \frac{I}{k'}$$
 Eqn 3

where C is SOC (or plant litter C), I is organic C input and k is the specific rate of decay. This simple formulation expresses the fundamental concept that (in a stable environment) SOM content (or plant litter stock) tends towards an equilibrium state given by the balance between plant carbon litter inputs (I) and SOM C decay, which can be defined by an overall mean residence time (equal to  $\frac{1}{k}$ ) of C in soil (Olson, 1963).

A second area of early mathematical modeling concerned the dynamics of nitrogen, given its many recognized states in soil and particularly with the advent of <sup>15</sup>N as a tracer, models were

formulated to describe and interpret the results of isotopic mixing and N transformations between different forms in soils (Jansson, 1958; Kirkham and Bartholomew, 1954). For early models of both soil C and N dynamics, models needed to be solvable using analytical methods and thus representation of SOM was highly simplified, usually as a single 'homogeneous' pool. Although the heterogeneity of SOM, in the sense that it is comprised of many different forms of organic matter decomposing at different rates, was clearly recognized (e.g Waksman, 1926; Waksman & Stevens, 1930), this key aspect of SOM was not addressed in early models, likely due to the practical mathematical reasons stated above.

With the widespread availability of digital computers and programming languages beginning in the late 1960s, along with the establishment of the International Biological Programme (IBP) and its goal of computer modeling of whole ecosystem dynamics (Innis, 1978), more complex models, which included soil organic matter dynamics formulated with multiple SOM pools, continuous driving variables (e.g., temperature, moisture) and a more mechanistic treatment of controls on decomposition and stabilization factors, were developed (e.g., Hunt, 1977; Smith, 1979; McGill et al., 1981). Arguably, one of the motivations affecting the formulation of this first generation of computerized models was a desire for comprehensiveness and detail in the components and processes known to be of importance. Hence, in several models, explicitly including microbial biomass components as explicit state variables controlling decomposition and transformation of non-living litter and SOM pools was axiomatic. While useful as explanatory and research models, the strong feedbacks between microbial biomass and organic matter pools and high sensitivity of the models to poorly-understood parameters controlling microbial growth and mortality, yielded model stability problems and made the models less suitable for predictive purposes. Subsequently, the vast majority of ecosystem-scale models of SOM dynamics developed during the 1980s and 90s (e.g. Smith et al., 1997) retained a multiple-pool approach and a relatively comprehensive inclusion of dynamic environmental drivers, but reverted back to a first-order kinetics paradigm. In these models the role of microbes and other decomposer organisms are certainly

implicit, and their responses are represented by rate modifying functions determined by dynamic environmental and management conditions, but the actual quantity or biomass of decomposer organisms are not included as a rate control. This mainstream approach largely continues to this day, although as discussed in the third section of this review, there is an ongoing reexamination of this dominant paradigm and an exploration of ways to more explicitly represent microbial controls on SOM transformations.

#### 2.2.3 SOM modeling approaches

The 'toolbox' of mathematical approaches to SOM modeling has seen relatively little recent change or expansion (Parton et al., 2015). Rather, advancements in SOM modeling are largely derived from new linkages between mathematical approaches and conceptual understanding of SOM dynamics. In this section we present a basic summary of mathematical modeling and applications to SOM, as background to more recent developments in SOM models reviewed in Section 2.3.

#### 2.2.3.1 First order kinetics – dominant multi-compartment models

1<sup>st</sup> order decomposition kinetics are a modeling approach where the flux of material from a pool is linearly related to the quantity of material in that pool (Figure 2-1, A & B). 1<sup>st</sup> order kinetics are a mathematically simple expression of decomposition that continues to dominate SOM modeling. It is also a mathematical approach that lends itself more easily to analysis of steady-state conditions, particularly when applied to a single pool model where the steady state is proportional to the input divided by the decomposition rate (Eqn's 2 & 3, detailed derivation in Stewart *et al.*, 2007).

As described above, a single-pool SOM model using 1<sup>st</sup> order kinetics is one of the simplest mathematical approaches to simulate a dynamic relationship between SOM and decomposition processes (Figure 2-1, A). A single-pool approach has been used to simulate SOM dynamics (e.g. Giardina & Ryan, 2000). However, the most common SOM model structure uses multiple pools to simulate SOM dynamics, each pool acting under 1<sup>st</sup> order kinetics but with different rates of change and

input movement from one compartment to another (Figure 2-1, B) (Paustian, 1994). The widely-used CENTURY model reflects this approach, using three compartments that reflect pools containing fast, slow, and passive SOM as well as additional pools representing fresh plant residues (Parton et al., 1987). As will be discussed in Section 2.3.5, this type of structure is particularly common within models applied at high spatial scales (e.g. Earth System Models (ESMs)), by serving as a relatively simple mathematical approach that dynamically represents decomposition across heterogeneous SOM. However, it is increasingly critiqued as having limited ability to reflect new understanding of mechanisms driving SOM dynamics (Todd-Brown et al., 2012).

#### 2.2.3.2 Organism-oriented and non-1st order kinetic

There are currently two main areas of development in SOM models exploring alternative approaches to 1<sup>st</sup> order kinetics. The first centers on an old debate described above, specifically: how are soil organisms represented in SOM models? Decomposition processes are obviously dependent on microorganisms. However, 1<sup>st</sup> order decomposition models like CENTURY implicitly model microbes through controls on decomposition rates while lumping microbial biomass within kinetically-defined SOM pools. This approach is based on the assumption that the soil microbial community can respond quickly to changes, and their biomass will therefore not limit the rate at which decomposition processes occur (Paustian, 1994). Early models like PHOENIX took a more organism-oriented approach, explicitly simulated microbial biomass as a direct control on decomposition rates (McGill et al., 1981). These types of models were more commonly applied to examine soil food webs and energy flow through the soil environment (Paustian, 1994). However, 1<sup>st</sup> order kinetic models represented a simpler approach shown to perform well simulating a wider array of SOM responses, and therefore gained more widespread acceptance and use than organism-oriented models for ecosystem-scale soil C dynamics (Stockmann et al., 2013).

Recent ideas in SOM persistence have brought explicit microbial models back into focus, where microbial biomass is included as a SOM pool that affects SOM decay rates (Figure 2-1, C). It is not yet clear if returning to these approaches yields improvements in model simulations across scales, but there are rapidly developing efforts exploring these mathematical approaches. This will be discussed in more detail in Section 2.3.1.

A perhaps less contentious topic is the implementation of saturation kinetics within SOM models, where the quantity of material in a pool—relative to a maximum amount—affects the flow of material into that pool (Figure 2-1, D). This is a reasonably straightforward, defensible SOM concept. However it adds mathematical complexity to SOM models relative to 1<sup>st</sup> order kinetics. It is also not yet certain how to quantify saturation capacity for a given SOM pool. Despite repeated observation of saturation kinetics, this is a dynamic that has had little successful implementation in SOM models. Saturation kinetics in SOM modeling will be discussed in more detail in Section 2.3.2.

#### 2.2.3.3 Conceptual vs. measurable SOM pools

Linking measurable SOM fractions to modeled SOM pools has been long recognized as an important area of development in SOM modeling (Elliott et al., 1996; Paustian, 1994; Stockmann et al., 2013). In the absence of this linkage, the dynamics of individual SOM pools cannot be directly compared against measured pool quantities to evaluate their accuracy in SOM simulations. This creates a strong potential for equifinality, where the same results can occur through different means (Luo et al., 2012; Tang and Zhuang, 2008), often referred to by the more colloquial phrase of 'getting the right answer for the wrong reasons'.

An additional benefit of linking SOM measured fractions to modeled SOM pools would be to reduce the uncertainty inherent in 'spin-up' equilibrium simulations needed to initialize conceptual (non-measured) SOM pools. The spinup method of SOM model initialization is based on the assumption that the underlying SOM pools are distributed with reasonable accuracy if the equilibrium simulations

match the initial conditions of the experiment being simulated, and therefore model pools 'self-initialize' as a function of their mathematically-defined turnover rates. This is the best method available to use a multi-pool model when, as in the case of most SOM models, the initial conditions of the pools are not measurable. Spin-up simulations often need to be run for tens, hundreds, or even thousands of simulated years, and therefore drastically increases the time required to complete experimental model simulations. In the case of large-scale analyses this can place a logistical limit on simulation complexity and applications (Xia et al., 2012). Direct measurements of SOM pools would remove the need for spin up simulations entirely, by making starting conditions equal to those known to exist in soils directly. On the other hand, determining initial pool sizes based on direct-measurement of the modeled pools would necessarily require that the soil measurements are in fact made, which could be a major constraint for large-scale regional and global model applications.

However, the fact that this topic continues to be recognized as a challenge in SOM modeling is expressive of the remaining uncertainty linking measurement of SOM fractions to conceptual understanding of SOM dynamics. Benchmarking data have been suggested to reduce equifinality (Luo et al., 2012). Increases in computational capacity has also allowed for innovations in data assimilation and the increasing use of Bayesian statistical approaches to constrain model parameters (Braakhekke et al., 2013; Luo et al., 2011). Until a SOM measurement method yields consistently SOM fractions uniform in their kinetic or biochemical behavior, the linkages between SOM measurements and modeled SOM pools will remain unclear (Dungait et al., 2012; Wander, 2004). Linking modeled SOM pools to measurable soil fractions remains an important area in the development of SOM model-data integration. In the meantime the type of measurements being analyzed as well as the scope and scale of the question being asked needs to guide model selection, as some models are more suited to certain types of analyses than others (Manzoni et al., 2012).

#### 2.2.3.4 Data-model integration in SOM research

A detailed discussion linking specific measurement methods to pools and dynamics within SOM models is beyond the scope of this work, with the exception of examples in topics reviewed in Section 2.3. However, we suggest exploring previous research examining this linkage between SOM measurements, conceptual understanding of SOM dynamics, and SOM models in greater detail (Dungait et al., 2012; Simpson and Simpson, 2012; von Lützow et al., 2007; Wander, 2004). Here we will instead step back and make a few points about the integration between data and models in SOM research and SOM model development.

First, there are several ways SOM models and SOM data interact. These can be grouped into three general categories: using data to *formulate* a SOM model, using data to *drive* a SOM model, or using data to *evaluate* a SOM model. Data used to *formulate* SOM models are tied to the hypotheses that a model represents. An example: using incubation data from warming experiments to mathematically define a SOM decomposition temperature response curve (e.g. Parton et al., 1987). Empirical, data-driven relationships are used in models when factors are known or hypothesized to impact SOM processes, but underlying mechanisms are either unknown or not fully understood. Data can also be used to parameterize components of an SOM model, optimizing model performance by 'tuning' parameter values to data when parameters are not measured directly. Data to *drive* a SOM model, on the other hand, are typically based on external factors known or hypothesized to force SOM behavior. These data, depending on their scale and variation, can link spatial and temporal heterogeneity to simulated SOM dynamics. These data often tie back to Jenny's factors of soil formation (Eqn 1), for example with air temperature and precipitation (climate), soil texture (parent material), and vegetation (organisms) commonly used in simulating SOM dynamics in ecosystem, global, or earth system models. Finally, data to *evaluate* SOM models are fairly self-explanatory, used to validate model

performance, evaluate uncertainty, and finally support hypothesis testing through the comparison of SOM simulations with measured results.

These categories help clarify some of the pitfalls in model-data integration. From the *formulation* side, data link to the hypotheses mathematically represented by a model. New data may therefore prompt model changes, for example if new data alter the shape of an empirical relationship built into model structure, or if new measurement methods yield entirely new hypotheses for SOM processes. In any case, data used in model *formulation* require careful consideration to ensure a model accurately represents hypotheses for SOM dynamics. The information to *drive* SOM models tends to be more sensitive to data limitations, particularly across scales. Data limitations for *drivers* affect how spatial and temporal heterogeneity are represented in the simulation of SOM dynamics, potentially leading to biased or erroneous results. Model *evaluation* experiences similar limitations determined by the scale at which an SOM model is being applied. Data availability for model *evaluation* will affect assessment of model accuracy, uncertainty, as well as its ability to support hypothesis testing.

The full suite of data for model formulation, driving, and evaluation are not always easy to identify or compare between SOM models, particularly when they are integrated into large-scale ecosystem or global models. Measurement uncertainty is also a consideration across all three types of data-model integration. Neither challenge is new (Keenan et al., 2011). However, in recent years researchers have targeted these challenges using rapidly advancing computational capacity, developing networks that bridge between SOM data and models. These efforts are aimed to support an iterative cycle of data-model integration. Components of this cycle includes data standardization, increasing data acquisition and accessibility to models, identifying and targeting data-poor areas of research (e.g. tropical soils), model benchmarking, and ideally multi-model comparisons (e.g. Luo *et al.*, 2012). These efforts provide a better environment for model development, and hypothesis testing. Large scale data-model integration is increasingly recognized as a multidisciplinary effort with high value for SOM

research. This is an area of development that will likely continue to advance, potentially dramatically and in the near future.

#### 2.3 New directions in SOM model development

We now present a series of short reviews on selected areas in SOM research of particular relevance for the use and development of SOM models. We emphasize areas where SOM modeling and experimental research can greatly benefit by integration and collaboration. A recurring theme is the challenge of integrating understanding across scales.

#### 2.3.1 Recent ideas for SOM stabilization mechanisms

Researchers have studied SOM dynamics for over a century, understanding with increasing sophistication how the quantity, quality, and cycling of SOM play important roles in soil functions (e.g. hydrology and water holding capacity), soil fertility (e.g. nutrient availability and cycling), and ecosystem processes (e.g. net exchange of carbon and energy flow) (Manlay et al., 2007; Wander, 2004). However, a fundamental area for SOM research, including SOM modeling, are the linkages between SOM cycling processes and SOM persistence. In other words: what controls the persistence of macromolecules in the soil against the pervasive presence and activity of organic matter mineralizing microorganisms and microbial enzymes?

For decades, stabilization mechanisms and SOM persistence have been integral components of research on SOM cycling and the development of SOM models (Campbell *et al.*, 1967; Jenkinson & Rayner, 1977; Parton et al., 1987; Oades, 1988). It continues to be a predominant research topic (Dungait et al., 2012; Kleber, 2010; Schmidt et al., 2011), of particular importance to understand the role of soils in the global C cycle and to determine their potential to either contribute or mitigate atmospheric greenhouse gas (GHG) emissions (Friedlingstein et al., 2001; Jobbagy and Jackson, 2000; Schimel, 1995). Soil organic matter turnover and stabilization mechanisms have been subject to numerous recent reviews (Table 2-1) and we will not tackle an exhaustive update to these previous

works. Rather we will focus this section on a major shift in one of the dominant conceptual paradigms for SOM stabilization, equating SOM chemical complexity with resistance to enzymatic attack and thus its longevity in soils. New observations of SOM chemical characteristics and behavior have provided evidence against inherent SOM chemical complexity as a dominant mechanism for long-term SOM stabilization (Dungait et al., 2012; Kleber et al., 2011; Schmidt et al., 2011; von Lutzow, 2008). New hypotheses have led to re-evaluating SOM modeling approaches as well as increasing exploration of more organism-oriented models explicitly linking microbial biomass and microbial mechanisms to SOM dynamics.

Stepping back, we would first like to link SOM persistence more generally to our previous discussion connecting SOM fractionation methods, modeled SOM pools, and conceptual understanding of SOM dynamics. Organic matter is a high-energy organization of molecules created and maintained by living processes that—upon death and particularly in the presence of the ubiquity of microbes and microbial enzymes—should decompose, forming new bonds at lower energy states (Hedges et al., 2000; Kleber, 2010; Schmidt et al., 2011). The persistence of OM in soils indicates widespread operation of protective mechanisms that slow or prevent OM decomposition from occurring in the soil environment.

Linking protective mechanisms to SOM persistence is, however, challenged by their temporal variability and tendency to interact. For example, some OM inputs are accessed quickly by soil microbes, mineralized within minutes, hours, or days. However, OM that becomes protected from microbial activity, e.g. tightly bound to soil minerals within micro-aggregates or more temporarily stabilized by roots or fungal hyphae within macro-aggregates (Golchin et al., 1994; Oades, 1984; Pronk et al., 2012; Six et al., 2004; Six and Paustian, 2014; Tisdall and Oades, 1982), can remain in soils for years, decades, centuries or even millennia depending on factors driving the dynamics of protective mechanisms (e.g. Oades, 1988; Jastrow, 1996). The net result of interacting stabilization mechanisms are soils containing

SOM that is a mixture of ages at different stages of decomposition, and that potentially react differently to change (e.g. land use or climate) (Paul et al., 1997; Schimel et al., 1994; Trumbore, 2000).

Experimentally isolating the mechanism or interaction of mechanisms that determine SOM persistence also remains difficult, as some mechanisms operate over extended time periods, and experimentally studying mechanisms often requires changing some of the conditions that affect them. Laboratory soil incubation studies, for example, must account for changes in soil structure and the effects of isolation from the *in situ* plant-soil system. Due to logistics and the development of increasingly unrealistic conditions in laboratory incubations through time, there is also predominance of short-term laboratory experiments evaluating more rapid SOM cycling mechanisms (Conant et al., 2011). Long-term field experiments begin to account for slower soil mechanisms and interaction between multiple factors in situ (Paustian et al., 1995), but these experiments can only cover so much ground and are still relatively 'short-term' for some soils processes, even with the longest (at Rothamsted, UK) having recently surpassed 170 years (Jenkinson and Rayner, 1977). Isotopic analyses using <sup>14</sup>C 'bomb carbon' labeling from atmospheric nuclear testing has high value for estimating responses of slow SOM mechanisms (Trumbore, 2009, 2000). It is also possible to evaluate longer-term SOM dynamics using other isotopic C labels, or natural changes in <sup>13</sup>C abundance when vegetation shifts between a dominance of C3 and C4 plants (Kuzyakov and Domanski, 2000). A combination of approaches and the use of non-C tracers can also be revealing in a given soil environment (Braakhekke et al., 2013; Kuzyakov and Domanski, 2000).

Regardless of experimental approach, bulk SOM dynamics are generally not fully informative of underlying mechanisms and can even mask important short-term dynamics when, as is common, total SOM mass is dominated by older, more slowly cycling material (Trumbore, 2000). Therefore SOM stabilization, turnover rates and persistence are typically evaluated in association with SOM fractionation methods. The age of bulk SOM ranges on average from decades to centuries, associated

with kinetic dynamics across soil fractions that exhibit patterns most consistent at kinetic extremes, e.g. the slowest turnover rates of hundreds to thousands of years associated with tightly mineral-bound OM, while the fastest turnover rates of weeks to a few years associated with fresh litter inputs (Balesdent, 1987; Campbell et al., 1967; Scharpenseel and Becker-Heidmann, 2006; Trumbore, 2009; von Lützow et al., 2007). Unfortunately SOM cycling at either extreme tend to make up a relatively small quantity of SOM, with the remainder cycling somewhere in a decade-to-century continuum that is, as discussed above, difficult to separate into fractions with uniform kinetic or biochemical characteristics.

Simulation models for SOM dynamics therefore play a key role integrating understanding in a mathematical framework that allows for broader and more rapid hypothesis testing than feasible through direct experimentation and measurement alone (Schimel et al., 1994). Models such as CENTURY and RothC—both still widely used today—are based on multiple 1<sup>st</sup> order dynamic SOM pools assigned different kinetics, or rates of change through time (Jenkinson and Rayner, 1977; Parton, 1987). While the multi-pool 1<sup>st</sup> order kinetic approach has been shown to successfully represent the heterogeneity of SOM decomposition and improve model predictions (Carpenter, 1981; Trumbore, 2000), these pools remain conceptually defined. There have been efforts to link modeled SOM pools to specific measured SOM fractions (e.g. Zimmermann *et al.*, 2007), but this is difficult particularly for slowly cycling SOM pools where the linkages between measure pools, stabilization mechanisms, and decomposition rates remain unclear.

It is within this context that a new understanding of SOM chemical complexity and its role as a mechanism for SOM persistence can be more thoroughly explored. Until recently mechanisms for SOM stabilization were generally grouped into three categories: 1) physical protection from microbial processes (e.g. through aggregate formation), 2) mineral associations that limit exposure to lytic enzymes, and 3) increasing OM recalcitrance by selective preservation of less biodegradable litter inputs and the formation of complex, stable humic molecules (Six et al., 2002; Sollins et al., 1996; von Lutzow,

2008). Mineral associations and physical separation continue to be recognized as important SOM stabilization mechanisms. They can be explored in greater detail in the reviews listed above (Table 2- 1). However the hypothesis that a main mechanism for long-term SOM persistence is the gradual transformation of primary biomolecules into complex secondary 'humic' molecules resistant to decomposition has been largely refuted.

The origin of the hypothesis linking inherent chemical complexity to SOM stability—what has been termed 'biochemical protection' or simply 'recalcitrance' (Six et al., 2002; Sollins et al., 1996) traces back to early research on soil humus. We refer to a historical accounting of SOM conceptual development by Manley et al. (2007) for greater detail, but here will summarize a few key points. Humus saw its first use as scientific term in the late 1700's/early 1800s, referring to the dark material accumulating in soils with the decay of dead plants and animals. One of the earliest linkages of humus to SOM as a soil constituent was in Albrecht Thaër's *Principles of rational agriculture* (1809). The connection between humus and soil fertility drove subsequent research to understand its development and behavior, particularly in the context of plant nutrition and soil nutrient availability (Manlay et al., 2007). This included exploring various measurement methods to separate humus into meaningful fractions, notably including acid and alkali extraction that yielded fractions with varying decomposition behaviors in soil incubation. It is worth noting that as early as 1938 an argument was made against the use of these chemical fractionation methods—at least to characterize humus—due to their dubious linkages to underlying soil processes:

"Practically all the methods suggested at various times for determining the amounts of "humus" or of the "humified" portions of the organic matter in soils, peats, and composts are unsuitable for this purpose, primarily because they were developed with little or no regard to the processes that take place when fresh organic matter undergoes decomposition...One is justified in speaking of humus only in regard to the sum total of the organic matter in soils, peats or composts. A study of the nature of this humus involves an investigation of the chemical nature of the plant and animal residues from which it originated, of the chemical processes of decomposition involved in its formation, of the nature and activities of the microorganisms bringing about the decomposition, as well as of the environmental conditions under which the decomposition took place." (Waksman, 1938, pg 87-88)

Some of the above should sound familiar from our introduction and Section 2.2. Clearly the challenge of linking measured OM fractions to conceptual understanding of underlying processes has taxed many generations of soil scientists!

The emergence of ecosystem ecology and the increasing focus on SOM as a component of biogeochemical cycles saw a gradual shift away from the use of humus as a scientific term in these fields, as concepts behind humus formation and dynamics fit within the purview of SOM processes and dynamics. Humus and humification came to be viewed more as classic soil concepts, referring to microbially and chemically processed organic matter resistant to further decomposition and the process that creates it, respectively (Lützow et al., 2006). However the chemical fractionation approach to characterize SOM by acid and alkali extraction, originating in humus theory and tied to the concept of isolating the large, complex organic molecules inherently resistant to decomposition (Schnitzer and Khan, 1972), largely remained, supported by radiocarbon dating that showed its isolation of SOM pools that cycled at different rates (Leavitt et al., 2006; Paul, 1984; Paul et al., 1997). This was generally referred to as the Humic Polymer Model (Brady and Weil, 2007).

Arguments against this theoretical approach and application of fractionation methods clearly are not new (e.g. Waksman, 1938). However, it is largely in the last decade that the weight of evidence against this hypothesis accumulated (Simpson et al., 2007; Stockmann et al., 2013), particularly based on new measurements for SOM chemical forms *in situ* as well as re-examination of the alkali- and acidbased chemical SOM fractionation method. Closer examination of the molecular nature of humic molecules have lent more support to the hypothesis that they have supramolecule structures—clusters of small molecules stabilized by hydrogen bonds and hydrophobic interactions, capable of forming micelles in aqueous solution and changing structure under different chemical conditions—rather than large polymer structures (Piccolo, 2002). Analyses also support that the behavior of humic molecules links more to the underlying properties that characterize molecular interactions, rather than the

chemical composition of humic molecules themselves (Sutton and Sposito, 2005). The link between acid and alkali extraction and SOM fractions with varying resistance to decomposition were shown to be largely a result of the extraction method, rather than indicative of underlying SOM chemical differences (Kelleher and Simpson, 2006). These extraction methods were further recognized to yield data with poor explanatory power for underlying SOM dynamics (Lützow et al., 2006; Wander, 2004). Within the soil *in situ*, evidence supported the characterization of SOM as a complex mixture of smaller biopolymers with complex spatial distribution in the soil matrix, rather than as large complex humic molecules (Lehmann et al., 2008). Ultimately, evidence supported decoupling inherent SOM chemical complexity from SOM persistence. As summarized by Kleber (2010); "recent investigations...revealed that significant quantities of organic material in soil may persist in spite of being chemically labile, unprotected, accessible, and decomposable" (pg 325). New hypotheses have emerged, centered on linking microbial processes with OM input chemistry and properties of the soil matrix (Cotrufo et al., 2013; Dungait et al., 2012; Gleixner, 2013; Schmidt et al., 2011). These hypotheses have invigorated an enduring debate in SOM modeling: to what degree should microbes and microbial mechanisms be explicitly represented in SOM simulations?

As mentioned in Section 2.2, the implicit versus explicit simulation of microbes and microbial processes in SOM models is a debate that dates back to the 1970's, when explicit microbially-based models such as PHOENIX emerged alongside models, such as RothC and CENTURY (Jenkinson et al., 1987; Jenkinson and Rayner, 1977; McGill et al., 1981; Parton, 1987). The latter models performed more successfully simulating long-term SOM changes, based on the assumption that litter quality and environmental factors (i.e. temperature, moisture, and soil texture) are the main drivers of SOM turnover. However, while these types of models have been shown to predict long-term SOM dynamics well they perform more poorly over transient dynamics and short-term change, both of which are important in the context of predicting impacts from climate change (Lawrence et al., 2009). Interest in responses to climate change combined with rapidly advancing methods to study microbial processes

have resulted in a renewed interest to explicitly connect microbial mechanisms to models of SOM dynamics (Stockmann et al., 2013).

Hypotheses have been developed linking variable microbial growth efficiency (MGE) to decomposition processes (e.g. Moorhead & Sinsabaugh, 2006; Sinsabaugh *et al.*, 2013). Previously some of these concepts were explored in the Q- model, which explicitly included microbial growth rate and assimilation efficiency effects on SOM turnover (Bosatta and Ågren, 1991), tied to measures of litter chemistry to model C decomposition and SOM accumulation (Ågren and Bosatta, 1996). A recent synthesis of theory is provided by the Microbial Efficiency-Matrix Stabilization (MEMS) hypothesis, based on evidence that stable SOM is predominantly comprised of microbially-processed organic matter. The MEMS hypothesis suggests the potential for greater stabilization of more labile plant material, as high microbial growth efficiency would reduce OM lost to respiration during decomposition, increasing the creation of microbial products that could then persist by interactions with the soil matrix (Cotrufo et al., 2013). There have been some successes with representing variable MGE within SOM models in large-scale analyses (Wieder et al., 2013).

Other modeling approaches are exploring the connection between microbial community structure and SOM stabilization. This might be implemented by modeling the presence or absence of 'narrow' microbial community functions, such as N fixation or lignin degradation that are only accomplished by certain types of enzymes created by a small subset of microbes (McGuire and Treseder, 2010). The guild-based decomposition model (GDM) provides one such model of this type, with a structure based on microbial affinity for different types of substrates, as regulated by changing litter lignin and N characteristics over the course of decomposition (Moorhead and Sinsabaugh, 2006). The Microbial-Mineral Carbon Stabilization (MIMICS) is another example, linking microbial community functional groups (specifically, r- versus K- strategists), litter chemistry, and SOM stabilization through dynamic microbial growth efficiency. They hypothesize that the greatest SOM stabilization in sandy soils

occurs with low quality litter inputs due to biochemical protection versus the greatest SOM stabilization in fine-textured soils with high quality litters due to physical protection of microbial products (W. R. Wieder et al., 2014). In both cases, these models are largely theoretical and difficult to test in complex soil environments, as measurements of microbial community functional groups are not well established. Measurement methods, such as the use of genetic markers to quantify soil microbial community functions, is an area of active development that will allow for these types of hypotheses to be tested more explicitly.

The role of mycorrhizal fungi in SOM stabilization is another area of recent interest. Researchers have shown that plants in symbiotic relationships with mycorrhizal fungi can compete directly with freeliving microbes for nitrogen (N) (Chapman et al., 2006). Mycorrhizal fungi, in particular ectomycorrhizae and ericoid mycorrhizae (EEM), secrete enzymes that allow them—and their plant symbiotes—to access organic N directly. However not all mycorrhizal fungal associations are equal in this regard, with arbuscular mycorrhizae providing more limited access to organic N. These differences led to the hypothesis that plants with different types of symbiotes would compete differently with free-living microbes for N, leading to varying degrees of microbial decomposition repression with concordant impacts on total SOM (i.e. better mycorrhizal-based competition for resources = more repression of free-living microbes = more SOM). This hypothesis was evaluated and supported at a global scale, across climate, clay content, and biomes (Averill et al., 2014). Additionally, a study of SOM accumulation in boreal forests due to root-fungal interactions demonstrated long-term continual SOM accumulation in the absence of disturbance. This suggested an alternative to SOM stabilization from aboveground litter inputs whereby SOM layers "grow from below" (Clemmensen et al., 2013, pg 1617). Cumulatively these results suggest mycorrhizal fungal associations are an important consideration in SOM stabilization across scales, and may be a fruitful avenue to consider in refining SOM models.

In the midst of these developing areas of research, it remains difficult to determine whether microbial mechanisms improve SOM predictions, largely due to lack of data to either drive or validate new models (see Section 2.3.5). For example, a review of conventional versus microbially-based model performance was largely inconclusive, since microbial models with promise were limited by lack of data from larger scale application and benchmarking (Treseder et al., 2012). This will not always be the case, as ongoing studies target filling in these data gaps. In terms of SOM stabilization, the new understanding of SOM chemical characteristics and interactions with microbial processes have created many new testable hypotheses that will continue to develop as data are accumulated (Rillig et al., 2007). We suggest an emphasis on testing new hypotheses across temporal and spatial scales, recognized by others as an important focus for SOM model development (Manzoni and Porporato, 2009). Testing across scales is necessary to determine if the added complexity of explicit microbial mechanisms leads to an improvement in SOM simulations. The degree to which specific microbial mechanisms are integrated into SOM models will likely depend on the scale at which a given model is being applied (Stockmann et al., 2013).

# 2.3.2 First order model versus saturation kinetics

As shown (Figure 2-1, A & B) and discussed both above and in Section 2.3.5, single and multipool 1<sup>st</sup> order decomposition kinetics are commonly used to simulate SOM dynamics. This simple approach performs reasonably well across a diversity of soils and land use changes (Paustian, 1994). Mathematically, however, 1<sup>st</sup> order kinetics implies a linear proportional relationship between OM inputs and the quantity of SOM stocks when a soils system is at equilibrium (Eqn's 2 & 3). This is a reasonable mathematical approach for some theorized mechanisms of SOM stabilization. The zonal structure theory of organo-mineral interactions is one example, suggesting the thickness of OM layers in the kinetic zones of organo-mineral complexes are predominantly determined by OM inputs (Kleber et al., 2007). However, field research repeatedly demonstrates the capacity for soils to become saturated,

in which case SOM stocks become relatively insensitive to increases in OM inputs or show a diminishing rate of SOM accumulation with increasing rates of OM input (Gulde et al., 2008; K. Paustian et al., 1997; Stewart et al., 2007). First-order kinetics may therefore overestimate OM gains in soils where SOM storage capacity is near a saturation point (Figure 2-2), i.e., where the rate of OM input into a pool may be controlled by the quantity already contained in the pool relative to its maximum holding capacity (Figure 2-1, D) (Hassink and Whitmore, 1997). Given the possibility of land use changes that lead to high OM inputs—such as fertilized, highly productive crops—and given interest in increasing soil C storage as a strategy to reduce atmospheric C concentrations (Follett et al., 2012), the potential for soil saturation is an important consideration as setting an upper limit to SOM stabilization (Baldock and Skjemstad, 2000). In the past this subject was given extensive review (Six et al., 2002). In this section we will update this review with advances in understanding and recent applications in SOM models.

In the Six et al. (2002) review, the three 'standard' categories of SOM stabilization mentioned in 2.3.1 were used, of mineral-associations (termed 'chemical protection'), physical protection through aggregate compartmentalization, and protection through inherent recalcitrance of SOM (termed 'biochemical protection'), with the remaining OM in the 'unprotected' pool. These pools were then linked to measurable SOM fractions in order to evaluate the potential for saturation within each fraction, as well as link these mechanisms to a maximum capacity for total SOM protection (Six et al., 2002). The 'biochemically protected' conceptual pool and its associated acid hydrolysis-based measurements have been largely dropped from studies evaluating saturation kinetics. However, the potential for saturation kinetics to affect both bulk soil OM storage as well as dynamics between mineral-associated, aggregate protected, and unprotected OM fractions in soils have continued to be developed and recognized as an important consideration in recent research on SOM and stabilization (Dungait et al., 2012).
Several key studies following the Six et al. (2002) review supported the SOM saturation concept and laid more of the groundwork for its current application. In bulk soils, for example, studies of longterm agricultural experimental sites confirmed the possibility of soil saturation behavior and suggest a distinction between the absolute maximum capacity of soils to stabilize C versus a soil's 'effective stabilization capacity', given external factors such as disturbance by tillage (Balesdent et al., 2000; Stewart et al., 2007). A subsequent laboratory incubation study of soils that varied in their 'saturation deficit' (i.e., proximity to a texture-dependent maximum SOC content), showed that 'saturation deficit' impacted total SOC stabilized with increasing C inputs. This study supported saturation behavior as impacting the total SOM stabilization capacity of soils, and further suggested that management practices to increase SOM storage in soils should target soils with greater starting SOM deficits (Stewart et al., 2008a). Several studies used the Six et al. (2002) framework to show that soil fractions could be evaluated for saturation behavior (Stewart et al., 2009, 2008b). Another study used an alternative measurement approach to evaluate physically separated fractions (e.g. macro and micro aggregates vs. silt- and clay- associated OM), and suggested the possibility of a hierarchy in SOM saturation across mineral-associated and aggregate-protected OM, increasing OM concentrations in more labile, faster cycling fractions when stable fractions are saturated (Gulde et al., 2008). These latter studies suggest that fractions within soils can become saturated even if bulk soils are not, with consequences for how SOM is distributed between pools that vary in turnover times.

The saturation concept has continued to be incorporated into studies of bulk soil OM dynamics. For example, bulk soils have continued to be evaluated for linear versus non-linear relationships between experimental OM additions and SOM content, in order to determine how close experimental soils are to their saturation capacity (Heitkamp et al., 2012; W. J. Zhang et al., 2010). A laboratory incubation study analyzing a gradient of degraded agricultural soils showed that, counter to Stewart et al. (2008a), the greatest C gains were in soils with intermediate rather than maximum saturation deficits

due to destruction of OM storage capacity in the most degraded soils. In this study researchers suggest management practices to increase OM storage need to intervene before soils reach a degradation threshold when stabilization mechanisms begin to decline (Kimetu et al., 2009).

The use of physically separated soil fractions to evaluate mineral –associated, aggregate protected, and 'unprotected' OM applied by Gulde et al. (2008) also continues to be a useful conceptual and measurement protocol in current research. One study linked soil C saturation theory to mineral N dynamics and potential for saturation, highlighting concern for mechanisms that saturate stable OM pools and concentrate OM in the more labile OM pools that are at greater risk for loss (Castellano et al., 2012). Another study of long-term agricultural experiments in Virginia uses a similar framework to show saturation in the silt+clay fraction, suggesting further OM accumulation at these sites will occur in particulate organic matter (POM) and aggregates and be more susceptible to loss unless management practices continuously minimize soil disturbance (Stewart et al., 2012). A recent study expands the mineral-associated fraction from only considering the saturation capacity of silt+clay to also include the saturation capacity of Fe and Al oxides. In some soils, such as the Andisols reviewed in the study, aluminum and iron oxides may contribute more to SOM stabilization than clay, therefore should be evaluated for saturation behavior (Matus et al., 2014).

In summary, experimental data increasingly support saturation kinetics as an important mechanism in SOM dynamics, particularly in soils with high SOM and/or increasingly high OM inputs. Therefore, saturation is conceptually well defined, well supported by experiment evidence, and simple (on its own) to express mathematically. It is notable, then, that saturation kinetics have had limited incorporation in SOM models since the Six et al. (2002) review. In one study exploring changes to the RothC-26.3 model to simulate physical protection with aggregate formation and hierarchy, saturation kinetics were not included but suggested as an improvement to include non-linear relationship between fresh organic matter and SOC dynamics (Malamoud et al., 2009). Another study with the RothC model

did add simulation of C saturation, in order improve simulation of high OM input treatments that deviated increasingly from model predictions based on 1<sup>st</sup> order kinetics. However, the modified model performed poorly for all treatments except those exhibiting saturation behavior, suggesting the need for better implementation of saturation dynamics within the model (Heitkamp et al., 2012).

In conclusion, this is a clear area in need of development in SOM modeling. At the very least, high OM soils in systems with high OM inputs should be recognized as vulnerable to overestimation of SOM storage and accumulation when simulated with conventional first-order decay SOM models.

## 2.3.3 Temperature controls on decomposition

Temperature is an important driver of SOM dynamics, recognized alongside moisture as key components of climate as a soil-forming factor (Eqn. 1, Jenny, 1941). This fact is neither new nor in dispute. However, in the last two decades controversy and debate have surrounded how, exactly, SOM dynamics respond to temperature change, motivated by the importance of understanding whether soils will become a stronger sink or source of CO<sub>2</sub> as temperatures increase under global climate change (Kirschbaum, 1995; Trumbore et al., 1996). Soils contain far more C than the atmosphere. Even a small percentage change in SOC with increasing temperatures can have a substantial impact on atmospheric CO<sub>2</sub> concentrations. However, the linkages between SOM dynamics and temperature-sensitive soil processes remain poorly understood. Behaviors observed in controlled laboratory incubations are often less consistent or less discernible under more realist *in situ* experiments. Temperature sensitivity is also extremely difficult to isolate from confounding effects of moisture and plant productivity in large scale analyses (Kirschbaum, 2000). While SOM temperature responses are far from resolved, in this section we will present the basis of past debate as well as summarize progress in the context of SOM model development.

First we will clarify our use of the terms 'temperature response' and 'temperature sensitivity'. **Temperature response** is a general term referring to changes in a unit of measure ( $CO_2$  flux, mean

resident time, etc.) over a range of incrementally increasing or decreasing temperatures. Temperature sensitivity refers to the magnitude of the response given the magnitude of the change, and is often used as a comparative term. For example, two soils incubated at the same interval of increasing temperatures might both have a positive temperature response and release more CO<sub>2</sub> per unit of temperature, but the magnitude of the increase in  $CO_2$  released per temperature increment is greater in the second soil (e.g. Figure 2-3,  $R_2$ ) than the first (e.g. Figure 2-3,  $R_1$ ). The second soil ( $R_2$ ) therefore has higher temperature sensitivity. Temperature responses are often exponential, so a common term used to express experimental differences in temperature sensitivity (as well as model temperature responses in SOM decomposition) is  $Q_{10}$ , or the proportional change in respiration with 10 Celsius change in temperature. A  $Q_{10}$  of ~2 (i.e. a doubling of respiration with 10°C temperature increase) is common across biologically meaningful temperatures, but values can vary from less than 0.5 to more than 300 and differ across soil fractions (Hamdi et al., 2013; Leifeld and Fuhrer, 2005). An Arrhenius type equation (Eqn. 4) and a modified version referred to as the Lloyd and Taylor equation (Eqn. 5) are both also common approaches to modeling temperature effects on respiration, connecting temperature effects to the activation energy of chemical reactions (Lloyd and Taylor, 1994). In these equations the rate of respiration given temperature, R(T) is calculated using the general structure of either;

$$R(T) = de^{E_a T^{-1}}, \text{ or }$$
Eqn 4

$$R(T) = de^{E_a(b+T)^{-1}},$$
 Eqn 5

where d is a constant,  $E_a$  is the activation energy, and b is a parameter used to correct temperature bias.

A central challenge in this area of research is disentangling the abundance of known and suspected temperature responses across the multitude of interacting processes that affect SOM dynamics. At a small scale, for example, many microbial decomposition processes are understood to be temperature sensitive, ranging from the strong temperature sensitivity of enzyme kinetics to potential temperature sensitivities of microbial growth and substrate use efficiency (Frey et al., 2013; Hagerty et al., 2014). Microbes can also acclimatize or change community characteristics under varying temperature environments through time (Luo *et al.*, 2001, Figure 2-4). The soil matrix can further modulate temperature responses, for example by affecting soil water dynamics or directly interacting with and suppressing microbial enzyme activity. The temperature sensitivity of decomposition also potentially varies by the chemical nature of the SOM substrate being decomposed (Davidson and Janssens, 2006), although as will be discussed this is an area of extensive debate.

On longer timescales and at a landscape level, other factors vary with temperature and cause difficulty with conclusively identifying temperature-related SOM mechanisms. For example, temperature responses of primary productivity occurs over short and long timeframes- e.g. immediate impacts on photosynthesis and transpiration as well as potential long-term impacts on vegetation structure and community dynamics. Primary production responses affect litter inputs into soils as well as soil water dynamics and the decomposition environment experienced by microbes (Bardgett et al., 2008). Ultimately, small-scale, isolated temperature responses and sensitivities in SOM dynamics may have a positive, negative, or no feedback with other ecosystem components in the soil environment when integrated over longer time scales, larger areas, or in interaction with other factors (Figure 2-4).

In this context, experimental results and SOM models are often combined to explore potential impacts of individual mechanisms, with an eye towards understanding global SOM changes with increasing temperatures (Jones et al., 2003; Kirschbaum, 1995). As a consequence, the underlying conceptual model of SOM structure then becomes the foundation on which mechanistic understanding of temperature responses are integrated and applied to larger scales. As will be presented below, this can result in the same data supporting opposite SOM responses to temperature increases. Advances in the understanding of SOM persistence (discussed in Section 2.3.1), have also led to re-interpretation of

some linkages between observed temperature responses and mechanisms within SOM models. The remainder of this section will focus on these developments.

Since the topic emerged, concern with SOM responses to global warming focused research on evaluating temperature sensitivities across temperature ranges in bulk soils as well as among individual SOM fractions with different turnover rates (i.e. 'labile' versus 'recalcitrant' SOM). Ecosystem models supported the potential for increasing temperatures to increase C inputs into soils via the stimulation of decomposition rates, particularly by increasing the release of mineral N (Schimel et al., 1994). An early synthesis of incubation studies noted variability in Q<sub>10</sub> values that suggested greater sensitivity of SOM responses at lower temperatures, which could place the large stores of SOM in cold boreal and arctic regions at greater risk for loss (Kirschbaum, 1995). A different study using repeated measures of soils across an elevation transect showed that more 'labile' SOM could contribute to rapid SOM response with temperature increases (Trumbore et al., 1996), but that longer term consequences are of greater concern for the often larger pool of SOM cycling more slowly.

Several topics then became subject to extensive debate, one centered on the hypothesis that 'labile' and 'recalcitrant' SOM fractions had different temperature sensitivities, with the latter being more sensitive than the former. At the time, this hypothesis was connected to a model linking SOM substrate quality to temperature sensitivity of decomposition. This model, now referred to as the carbon quality-temperature theory (Fierer et al., 2005), was based on the hypothesis that SOM decomposition dynamics were determined by substrate quality via the number of enzymatic steps- and therefore the total free energy change- required to mineralize organic matter carbon. The carbon quality-temperature theory predicted greater temperature sensitivity in low quality substrates compared to high quality substrates, as well as greater temperature sensitivity at low temperatures versus high temperatures (Bosatta and Ågren, 1999). Based on the assumption that 'old' SOM is more chemically complex and a poorer microbial substrate than 'new' SOM, this made a logical linkage to the

hypothesis that 'older' (i.e. more stable, 'recalcitrant') SOM will be more sensitive to temperature than 'newer' (i.e. less stable, 'labile') SOM. Liski et al (1999) then explicitly argued against this hypothesis, presenting results that suggested old SOM is less temperature sensitive than newer litter, leading to less C loss and even some C gain with increasing temperatures than predicted by other models (Liski et al., 1999). However, the SOM model underlying Liski et al.'s (1999) conclusions was criticized, specifically the assumption of fixed residence times in the pools used to simulate SOM and temperature effects on respiration (Ågren, 2000). Subsequent studies continued to examine this hypothesis, yielding results that supported, showed no difference, or suggested the opposite relationship between 'labile' versus 'recalcitrant' SOM, maintaining a high level of uncertainty on this topic (Benbi et al., 2014; Conant et al., 2008; Fang et al., 2005; Hartley and Ineson, 2008; Melillo et al., 2002). This theory has been argued as largely irrelevant for mineral-associated SOM and SOM that is cycling more slowly. As discussed in Section 2.3.1, 'recalcitrance' does not have a basis in chemical complexity (Kleber, 2010), and older SOM is not necessarily more thermodynamically stable or chemically different than 'newer' SOM (Kleber et al., 2011). The activation energy-based temperature sensitivity in the carbon quality-temperature theory, therefore, is not necessarily a mechanism with logical application to mineral-associated or slowly cycling OM pools (Conant et al., 2011). However, it should be noted a study that looked to address the shortfall of short-term incubation studies in the comparison of 'labile' and 'stable' SOM temperature sensitivities—specifically by evaluating initial versus final soils from long-term bare fallow agricultural experiments, where labile SOM has been shown to be progressively depleted through time—used apparent activation energy as part of strong evidence supporting greater temperature sensitivity in more stable SOM (Lefèvre et al., 2014).

A second topic of controversy in temperature effects on SOM emerged from a study that questioned the temperature sensitivity of mineral soils as a whole, suggesting they are largely insensitive to temperature and therefore may not be as important in the context of climate change as

hypothesized at that time. This study based their conclusions on experimental results from mineral soils in temperate forested systems (Giardina and Ryan, 2000). A theoretical model exploring the separation of physico-chemical stabilization (e.g. stabilization by adsorption to soil minerals) from microbial processes was able to replicate this effect simulating SOM at equilibrium, if physico-chemical stabilization increased in excess of microbial respiration (Thornley and Cannell, 2001). However, as with the Liski et al. (1999) study, results from Giardina & Ryan (2000) were also critiqued on the basis of the underlying SOM model, which in this case made the assumption of using single SOM pool model to calculate turnover times. It was suggested that a multi-pool SOM modeling approach was needed to calculate turnover times, to prevent obscuring temperature sensitivities of individual SOM pools (Davidson et al., 2000; Powlson, 2005). The Giardina & Ryan (2000) data were eventually re-analyzed by Knorr et al. (2005) using a multi-pool approach, which showed that cumulative temperature insensitivity in the short term *could* be exhibited by aggregating responses of individually temperature-sensitive SOC pools, but ultimately would lead to a strong positive feedback in the long term as more slowly cycling pools persist in releasing CO<sub>2</sub>. This reverses the conclusion from Giardina & Ryan (2000) and fell in line with other studies, by supporting a strongly positive feedback between warming and SOM decomposition in mineral soils. However, Reichstein et al. (2005) recommended precaution towards the Knorr et al. (2005) conclusions. Though they supported the use of the multi-pool approach, Reichstein et al. (2005) noted that Knorr et al.'s (2005) results—showing that slowly cycling SOM is more sensitive to temperature than more rapidly cycling SOM—could be due to a statistical anomaly (Reichstein et al., 2005).

Even as research has moved past these earlier controversies, identifying mechanistic linkages between SOM dynamics and temperature remain a challenging experimental effort. As mentioned earlier, it is difficult to isolate temperature-sensitive mechanisms in SOM dynamics, particularly across scales and within slowly cycling SOM pools. It has been suggested that the latter is in part due to the

relatively short duration of most climate change experiments (Conant et al., 2011). Regardless, evidence for positive feedbacks between increasing temperatures and SOM is often strongest with incubation studies due to minimization of confounding effects, such as concurrent changes in primary productivity (Janssens et al., 2001). At larger scales temperature sensitivity often seems reduced, suggesting a weaker or slower positive feedback between warming temperatures, SOM, and atmospheric  $CO_2$  (Ise and Moorcroft, 2006; Kirschbaum, 2000). One large scale study did support a linkage between SOM loss and warming climate conditions through direct measurements at a national scale, although they were not able to distinguish loss through respiration versus loss through leaching (Bellamy et al., 2005). In their highly cited review on the topic, Davidson and Janssens (2006) highlight the relative ease with which environmental factors can obscure temperature sensitivities, suggesting separation between 'intrinsic' and 'apparent' temperature sensitivity due to the interaction with environmental constraints that could amplify or suppress temperature responses. Ecosystem respiration has shown a strong convergence in 'intrinsic' temperature sensitivity when high resolution FLUXNET CO<sub>2</sub> data were used to exclude confounding ecosystem properties and seasonal dynamics (Mahecha et al., 2010). However, this analysis also suggested mechanisms that impact C stabilization versus C supply—operating both aboveand belowground on longer time scales—may potentially have complex impacts on ecosystem respiration and C cycling behavior (Mahecha et al., 2010). Studies support the potential for increased rates of C cycling in soils under increasing temperature, but the resulting impact on SOM storage remains uncertain (Bond-Lamberty and Thomson, 2010; Giardina et al., 2014; Smith and Fang, 2010). Cumulatively these studies support the importance of identifying mechanisms that drive SOM responses to warming temperatures, as well as incorporating these mechanisms explicitly into SOM models.

Many widely used SOM models cannot easily reflect specific temperature-sensitive mechanisms, due to their basis on kinetically-defined SOC pools that are not clearly linked to measurable soil fractions (Dungait et al., 2012). Representations of temperature controls on decomposition within these models,

and the data used to inform them, must therefore be considered carefully. For example, many widely used models initially had their temperature functions parameterized using incubation data. Examples include the CENTURY and the RothC models (Jenkinson and Rayner, 1977; Kirschbaum, 1995; Parton, 1987). These models have obviously undergone considerable change and development since initial parameterization (e.g. Del Grosso et al., 2005). However, given observations that temperature sensitivity tends to change in strength across spatial scales (e.g. Kirschbaum, 2000; Ise & Moorcroft, 2006), incubation-based model parameterizations of temperature response should be examined carefully when models are applied at larger scales. Additionally kinetically-based SOC models are commonly implemented in Earth System Models (ESMs), often using Q<sub>10</sub> functions to simulate temperature sensitivity (Todd-Brown et al., 2013). Q<sub>10</sub> functions have been shown to perform poorly at temperature extremes, making this approach potentially poorly suited to simulate tropical and arctic ecosystems where climate change is of particular concern (e.g. Koven et al., 2011). The use of a static Q<sub>10</sub> function has also been strongly criticized (Davidson et al., 2006; Tang and Riley, 2015). A Gaussian model has been suggested as a better representation of temperature sensitivity across a wider range of temperatures (Tuomi et al., 2008). Using multiple temperature response models for measured-modeled comparison can also be more revealing than the use of a single model (Benbi et al., 2014). Implementing functions to better simulate temperature sensitivity in SOM models based on kinetically-defined SOC pools, as well parameterizing temperature sensitivity across scales, is a logical and reasonably straightforward area to improve SOM model development. However kinetically-defined models will always be limited by the lack of mechanistic linkages between SOM dynamics and temperature response, by the inability to link SOC pools to mechanistically defined and measurable SOC fractions.

As an alternative, researchers are beginning to develop and explore models to simulate temperature-sensitive respiration responses more mechanistically, based on understanding of factors such as substrate diffusion, enzyme activity, and membrane transport (Davidson et al., 2006). The

DAMM model, as one example, simulates the interaction between soil water, soluble substrates and temperature as they affect enzyme kinetics (Davidson et al., 2012). Other models like *ecosys* are better suited to accommodate mechanistically-based microbial responses to temperature, by explicitly simulating microbial communities (Grant et al., 2003). The substrate quality-based model by Bosatta & Ågren (1999) provides a mechanistic basis for temperature sensitivity in some SOM components. A litter incubation study supported this theory, with results showing increased temperature sensitivity in lower quality litters as well as litters as they progressed through the decomposition process (with declining quality as decomposition advanced) (Fierer et al., 2005). The carbon quality-temperature theory was also supported by an analysis of the low density (or "light fraction"; LF) fraction of soils, a SOM fraction considered to be more microbially accessible as it is not associated with soil minerals (Wagai et al., 2013).

More recent models include better mechanistic linkages to a wider array of SOM pools. One conceptual model differentiates between processes that make SOM available for decomposition (e.g. physical protection and aggregate turnover) versus processes that decompose SOM once it is available (e.g. microbial enzyme dynamics, depolymerization, Figure 2-4) (Conant et al., 2011). Microbial interactions with SOC dynamics under warming scenarios have suggested a diversity of possible pathways for SOM dynamics and interactions with global carbon cycles (Bardgett et al., 2008), and also are being more explicitly incorporated into SOM models. Microbial substrate use efficiency has been shown to be sensitive to temperature, but with the potential for community shifts and acclimation that may reduce this sensitivity through time (Allison et al., 2010; Frey et al., 2013; Luo et al., 2001; Tucker et al., 2013). An incubation study supported a separation of microbial turnover from MGE, with temperature sensitivity of the former and temperature insensitivity in the latter leading to possible increases in SOC with temperature, a dynamic not reflected in current models (Hagerty et al., 2014). A model linking microbial biomass and enzyme kinetics to temperature and moisture aims to tease apart

these tightly linked drivers, supporting field experiment results suggesting moisture control when moisture is limiting, and temperature control when it is not (Steinweg et al., 2012). A recent model suggests CUE and abiotic mechanisms like mineral sorption vary interactively in temperature sensitivity through time, suggesting the need for temperature sensitivity to be modeled dynamically in order to gain greater accuracy through better linkages to underlying processes (Tang and Riley, 2015). Long-term multi-factorial ecosystem-level experiments are needed to clarify interactive climate change effects on microbial processes, and better inform development of these types of models (Bardgett et al., 2008).

In summary, temperature sensitivity in SOM dynamics remains an important area of continued research, both experimentally and from the standpoint of SOM model development. There has been a strong call for greater standardization and clarity in how experimental results are reported and interpreted, to support better integration with modeling efforts (Subke and Bahn, 2010). Models, in turn, need to better reflect the full spectrum of potential temperature responses, in order to more accurately simulate the implications for the timing, magnitude, direction, and geographic location of SOM changes under increasing global temperatures. Only further experimental exploration of temperature-sensitive decomposition mechanisms—particularly including large-scale studies that cross a range of ecosystem types (Giardina et al., 2014; Subke and Bahn, 2010)—integrated with SOM model development will continue advancing this area of research.

## 2.3.4 Deep soil organic matter dynamics

'Deep soil' in this discussion refers to the soils beneath the surface 20-30 cm depth layer, which has often been excluded from many ecosystem-scale models of SOM dynamics. Organic matter dynamics below the top soil layers remain, to a large extent, much less studied and one of the open frontiers in SOM research. Subsurface soils are consequently an important area of development in SOM modeling. Many models focus on the top 20-30 cm soil layer, partially due to logistics; surface soils are easier to sample, are often more uniform (particularly in tilled agricultural soils) and have much more

data available for model formulation and testing. In addition, surface soils generally contain a large fraction of the total SOM, which is younger and cycling more rapidly than SOM in deeper soil layers, and which is most responsive to management perturbations (Batjes, 1996; Jobbagy and Jackson, 2000; Paul et al., 1997; Scharpenseel et al., 1989). This has led to their emphasis in the simulation of ecosystem dynamics and soil responses to land management practices. Historically, SOM in deep soil layers was thought to consist mainly of inert complex humic material and mineral-bound OM. However while mineral-bound OM remains supported as an important stabilization mechanism in subsurface soils (Rumpel and Kögel-Knabner, 2010), more recent analyses show deep SOM to mainly consist of simple molecules—often dominated by highly processed microbial products (Erich et al., 2012)—that are also responsive to land management change (Baker et al., 2007; Follett et al., 2012; Poeplau and Don, 2013; Trumbore et al., 1995), on shorter timescales than previously understood (Koarashi et al., 2012). Therefore there is more recent focus on dynamics of the 50% or greater total SOC contained in soils below 20-30cm (Batjes, 1996; Jobbagy and Jackson, 2000), particularly in the context of subsurface soils' role, response, and management under global change scenarios (Salomé et al., 2010). Knowledge gaps in the understanding of deep SOM dynamics, as well as their connection to global cycles, are being targeted by experimental and field research, with calls to better reflect and test new hypotheses in SOM models (Schmidt et al., 2011).

Deep soil OM dynamics involve similar mechanisms to surface soils, but with the potential for time lags and differences in soil environments that may separate subsurface and surface SOM responses to change (Fierer et al., 2003a; Salomé et al., 2010; Sanaullah et al., 2011). A review of deep soil OM by Rumpel & Kögel-Knabner (2010) provides a framework for recent experimental work focusing on deep SOM dynamics. They highlight the need to clarify dominant SOM inputs as well as mechanisms that stabilize or destabilize SOM in deep soil layers. They summarized key inputs to deep SOM as 1) the movement of dissolved organic matter (DOM) with water, 2) root growth, exudates, and turnover, and

3) physical OM transport through bioturbation or physical soil processes. They suggest destabilization mechanisms include aggregate disturbance and increases in microbial access to nutrients and labile C. Stabilization of deep SOM is linked more strongly to mineral associations and the physical separation between dispersed microbes and SOM at depth. However the authors state that the relative importance of these mechanisms as well as their interactions remained in need of development, particularly with *in situ* field experiments (Rumpel and Kögel-Knabner, 2010). Moreover, the relative importance of these factors is likely variable across ecosystems (Jobbagy and Jackson, 2000; Schmidt et al., 2011). In the following section we will summarize recent research in deep SOM, expanding on stabilization and destabilization mechanisms as well as the dynamics of deep SOM inputs. We will then discuss modeling approaches to deep SOM dynamics and conclude with suggestions for future directions.

The importance of physical separation between microbes and OM in deep soils as a stabilization mechanism—specifically as opposed to the inherent chemical complexity of deep SOM—has received widespread experimental support (Rumpel and Kögel-Knabner, 2010; Salomé et al., 2010). For example, a study using 'bomb' radiocarbon from past atmospheric thermonuclear testing provided evidence that subsurface SOM can cycle rapidly on a decadal scale, turning over up to an estimated 1.6% of total annual gross terrestrial primary production across temperate grassland and deciduous forest biomes each year (Koarashi et al., 2012). In this study the chemistry of the light fraction, often theoretically used as a measure of fast-cycling SOM, was consistent across depths, only increasing in age but not changing in chemical characteristics (Koarashi et al., 2012). A different field study using <sup>13</sup>C and <sup>15</sup>N labeled wheat roots buried in litter bags at three soil depths to 90cm observed that the chemical characteristics as well as the quantity of litter decomposed at different depths was similar after a 3-year period (Sanaullah et al., 2011). The importance of physical protection from microbes in subsurface SOM stabilization is further supported by a laboratory incubation demonstrating that old SOM in deep soil layers could be mobilized simply by successive drying and rewetting, suggesting that SOM in deep soils is not

necessarily biologically inert but rather may persist through mechanisms that limit microbial access (Schimel et al., 2011). The authors of this study suggest the term 'metastable', for deep SOM that is 'unused but not unusable' (pg 1103).

Mechanistic linkages between microbial community characteristics and SOM stability remain poorly understood. Microbes clearly vary across the soil profile, and can show different sensitivities to factors such as temperature, moisture and nutrient addition (Fierer et al., 2003a). Microbes have a strong association with SOC content across soil depths (Fierer et al., 2009, 2003b), and their use of 'old' versus 'new' soil C has been shown to depend on depth as well as microbial community composition (Kramer and Gleixner, 2008). One study demonstrated that microbial communities varied as much across the vertical soil profile as they do between ecosystem in the surface soils, but without being able to link these differences to SOM dynamics (Eilers et al., 2012).

Regardless of mechanistic linkages to microbial community characteristics, the importance of SOM stabilization by physical separation from sparse microbes supports the potential importance of 'priming' as a SOM destabilization mechanism in deep soil layers. Priming occurs when inputs of 'fresh' OM - i.e. labile OM that has not been microbially-processed leads to increasing decomposition of otherwise persistent deep SOM with the stimulation of microbial activity (Fontaine et al., 2007; Kuzyakov et al., 2000). Priming could play an important role in predicting deep soil responses to increasing concentrations of atmospheric CO<sub>2</sub>, particularly in forested systems where elevated CO<sub>2</sub> has been shown to increase fine root production as well as cumulative C inputs into deeper soil layers (Iversen, 2010). The potential for priming also must be considered in land management strategies for increasing SOM, particularly with the growth of deep-rooted crop species. Deeps soils have the potential to gain and store soil C with land management to increase soil C sequestration (Poeplau and Don, 2013). However, priming may offset desired gains in SOM storage with increased belowground OM inputs (Dungait et al., 2012; Follett et al., 2012). Research aimed towards evaluating priming in deep soils has

not yet identified easily generalizable mechanisms, but rather suggests a high degree of site specificity. For example, Free-Air CO<sub>2</sub> Enrichment sites have shown accrual, loss, as well as no change in SOC from increased OM inputs to deeper soil layers (Drake et al., 2011; Iversen et al., 2012; Langley et al., 2009).

As described earlier there are several pathways for OM to move into deep soil layers, including bioturbation, root growth and turnover, and DOM movement. Of these, DOM has received particular attention. This attention is due to its rapid cycling, as well as its role linking surface OM with deep soil mineral fractions through vertical transport and thereby possibly affecting deep SOM stabilization and storage (Fröberg et al., 2009; Rumpel and Kögel-Knabner, 2010). The quantity and quality of DOM from different parts of the soil profile can also serve as a metric for cumulative behavior of sorption, desorption, decomposition and leaching processes, as they interact with soil minerals, pH, litter, and hydrology (Kalbitz et al., 2000). In earlier research DOM emerging from deeper soil layers were thought to have 'escaped' either adsorption or respiration after generation from litter decomposition (Schiff et al., 1997), an idea supported by bulk DOC measurements across the soil profile (Kalbitz et al., 2000). However more recent evidence supports the concept of repeated microbial processing as DOM is successively sorbed and desorbed down the soil profile, leading to predominance of highly processed DOM as depth increases (Kaiser and Kalbitz, 2012). Drying-rewetting can affect the depth and direction of DOM movement different in surface versus deep soil layers, with the potential for either upward or downward flow depending on the size and frequency of drying and rewetting cycles (Lopez-Sangil et al., 2013). There is strong evidence that Fe- and Al-oxide concentrations have a positive effect on DOM sorption capacity (Oren and Chefetz, 2012; Sanderman et al., 2008). However other mechanisms can affect DOM in deep soil layers. For example N addition decreased DOM release from deep soils in a temperate largely deciduous forested system. The authors linked this behavior to changes the soil chemical environment and solute behavior with the form of N addition (Hagedorn et al., 2012).

Recent research on deep SOM dynamics still only reveals pinpoints of understanding in a complex belowground system. Logistical challenges and lack of data are profoundly limiting. For example the Jobbagy & Jackson (2000) global estimate of SOM cited above required mathematically extrapolating down to 3m from datasets largely limited to 1m depths, excluded all agricultural soils, and discussed the possibility of biome-specific biases due to lack of data (Jobbagy and Jackson, 2000). 'Deep soil' research also may often not go deep enough. A study of highly weathered, very deep soil profiles demonstrated that SOC measurements limited to the top 50cm of soil would only capture as little as 14% of total SOC across the soil profile, while even measurements as deep as 5m captured at most 75% of the total in soils that averaged 21m in depth (Harper and Tibbett, 2013). Burial of soils through volcanic, loess, alluvial, or human mechanisms can create pockets of deep SOM with characteristics and behavior largely unaccounted in either global estimates or existing deep SOM model dynamics (Chaopricha and Marín-Spiotta, 2014). In this context, perhaps even more so than in other areas of SOM research, SOM model-data integration is needed to advance understanding. Simulation models can provide a framework to integrate data and test hypotheses, potentially helping direct subsequent experimental work to tease apart interacting, often co-occurring mechanisms in deep soil environments.

In the past, deep soil SOM turnover was only explicitly included in a few models and was noted as a critical area for SOM model development (Falloon & Smith, 2000). In addition to different approaches to addressing the three main deep SOM inputs summarized by Rumpel and Kögel-Knabner (2010)—roots and exudates, SOM movement with water, and vertical mixing with bioturbation and other physical processes—models developed before and since use a variety of approaches to simulate changing dynamics and SOM stability across soil layers. As one example, the DAYCENT model—a newer version of the CENTURY model—was modified to simulate deep soil C dynamics only by slowing SOM pool turnover and increasing allocation to passive soil C, without separating soil layers (Wieder et al., 2014b). Alternatively the Community Land Model (CLM) was modified to simulate deep SOM dynamics

using a vertical cascade, where SOM passes through layers in the soil profile with loss at each transition (Koven et al., 2013). The vertical cascade approach and multiple soil layers is more common (Table 2-2). The SOLVEG-II model, for example, uses 27 separate layers from the surface to 5.5m, each individually containing three CENTURY-type first order kinetic SOM pools with flows between preceding and following layers (Ota et al., 2013).

As discussed earlier, DOM is clearly an important mechanism of OM input into deep soil layers, and recent models have varied in explicitly or implicitly simulating DOM movement (Table 2-2). The RothPC-1 model, for example, includes DOC implicitly by modeling vertical C transport associated with advection (distance time<sup>-1</sup>) (Jenkinson and Coleman, 2008). The DyDOC model, on the other hand, explicitly models DOC dynamics across soil layers using measured soil C pools derived from a detailed <sup>14</sup>C labeling experiment (the Enriched Background Isotope Study (EBIS) out of Oak Ridge Reservation (Tipping et al., 2012). However the DyDOC model does not include the exchange of C with solid soil C, suggested by other researchers as a mechanism to explain both greater age and greater microbial processing of OM at deeper soil depths (Kaiser and Kalbitz, 2012; Sanderman et al., 2008). The SOVEG-II model is another model that explicitly simulates DOM, as a function diffusion, advection with water flow, plant uptake, and microbial decomposition (Ota et al., 2013). Alternatively, in the SOMPROF model DOM is implicitly simulated using a single advection rate (Braakhekke et al., 2013).

We would like to note that root inputs and bioturbation show much less variation in modeling approaches than DOM. Root distributions with depth tend to be modeled using exponential functions (Table 2-2). Modeling approaches to roots has been critiqued as insufficient to capture key dynamics, with the suggestion of modifying models to accept root distribution data directly (Iversen, 2010). Limited study of deep root dynamics is largely linked to the logistical challenges of their measurement and evaluation (Maeght et al., 2013). The limited exploration of modeling approaches to root growth in SOM models is likely a combination of the lack of data with the potential complexity involved in moving

away from a simple exponential function. However vertical root dynamics are certainly important the context of projecting SOM responses with climate change (Iversen, 2010). Therefore as deep SOM modeling approaches develop, this is an area in need of careful consideration. Bioturbation, on the other hand, has been shown to be largely inconsequential compared to other input and transport mechanisms (Braakhekke et al., 2013). It is therefore less of a priority for deep SOM model development.

The goal of these various modeling approaches to deep SOM is to clarify key mechanisms, identify patterns, and ideally simulate impacts of future climate projections and land use change scenarios with some degree of accuracy. The C-TOOL model takes a practical approach to its application, by simplifying assumptions and solely focusing on whole-soil SOC dynamics in agricultural systems. The C-TOOL model aims to encourage utility by minimizing parameters and data requirements for its use, and saw reasonable model performance when initial SOM pools were optimized using measured data (Taghizadeh-Toosi et al., 2014). In a study by Guenet et al. (2013), simplicity was also the aim, by comparing simple models that simulate vertical SOM transport using either only diffusion, only advection, or the combination of diffusion and advection as they acted on three different SOC mineralization models (1<sup>st</sup> order kinetics, only the input of fresh OM, or 2-pool 1<sup>st</sup> order kinetics) (Guenet et al., 2013). In this study the vertical transport mechanism with the best fit to data from a longterm bare fallow experiment depended on the underlying SOC model, yielding little mechanistic understanding for transport but supporting the importance of fresh SOC inputs in controlling deep soil OM dynamics (Guenet et al., 2013). Other experimental results are supported by model simulations. Numerical experimentation of rooting depth using the SOLVEG-II model resulted in SOM movement below rooting depth as well as dominance of SOM in active and slow pools at depth (Ota et al., 2013). These results confirm both as possible mechanisms leading to global observations of SOM below rooting depths (Jobbagy and Jackson, 2000), as well as supporting observations that deep SOM can cycle on

decadal scales (Koarashi et al., 2012). The BAMS1 model, which takes a more mechanistic approach to SOM and includes vertical SOM dynamics, simulated the persistence of relatively decomposable SOM across the soil profile. This model has potential to test SOM responses to global change across the soil profile. However it is highly complex and requires expanded observed data to more carefully test components of the model's internal structure (Riley et al., 2014).

Clearly there is room for iterative development of deep SOM understanding by combining experimental, field, and modeling research. We would like to note that, given the preceding short review, changes in temperature sensitivity and response to climate change are particularly poorly understood across soil depths, are not accounted in these models, and are an area in need of targeted consideration (Subke and Bahn, 2010). The models discussed in this deep soil review support some observed mechanisms but also suggest testable hypotheses in need of additional data to confirm or refute. Carbon isotope labeling and tracers are particularly important tools for deep SOM research, by allowing for OM dynamics to be observed with minimal disturbance. However there is likely value in using other, non-carbon tracers with known dynamics and interactions with SOM. For example, the use of <sup>210</sup>Pb<sub>ex</sub> was able to inform SOMPROF model parameters, although use in addition with <sup>14</sup>C or other C labeling was suggested as a more powerful approach (Braakhekke et al., 2013). Given logistical challenges in studying the deep soil realm, collaborative efforts between modelers and experimental researchers are needed identify, understand, evaluate, and predict deep SOM dynamics.

## 2.3.5 Representation of SOM in earth system models (ESMs)

The development of SOM modeling within Earth System Models (ESMs) is an advancing area of research in need of careful attention from soil scientists. Soils contain the largest terrestrial C pool, and have the potential to be either a sink or source of atmospheric greenhouse gas (GHG) emissions due to interacting factors that vary across scales, ranging from global vegetation shifts, to site-specific land management practices, to temperature effects on microbial dynamics (e.g. Section 2.3.3) (Heimann and

Reichstein, 2008). The integration of SOM models into ESMs is a critical activity for the robust characterization of feedbacks between the Earth System and future climate changes, both for the purposes of mitigation and adaptation across multiple fields that include the sciences, economics, and policy. However the inclusion of SOM models in ESMs present new challenges in SOM model development and validation, due to high uncertainty, variability, and uneven coverage in both data needed to drive SOM models and data to evaluate model performance. The use of SOM models within ESMs also requires SOM researchers to grapple with the challenge of integrating the understanding of SOM dynamics across scales (Manzoni and Porporato, 2009). Model intercomparison, benchmarking, and model-data integration projects will be an increasingly important component of scientific development in this field (Luo et al., 2012). In this section we will describe the development as well as across multi-model ESM comparisons

Global biotic C cycling components included in ESMs are often grouped in three broad categories: atmosphere, land, and ocean. As the largest terrestrial C pool (Jobbagy and Jackson, 2000), as well as by interacting dynamically with atmospheric C, soils play an important role in determining global land-based C cycling and land-atmosphere C interactions. Models of SOM are accordingly needed in ESM development to dynamically link atmospheric C, climate change effects, and land-based C storage (Falloon and Smith, 2000; William R. Wieder et al., 2014).

Earth system models are highly computationally intensive but have been developed to increase their resolution and complexity as advancing technology allowed more detailed climate modeling coupled with feedbacks from biotic and abiotic system components. ESM development began in the 1960s, but originally only included static land surface simulations to gain limited capacity in simulating water runoff and evaporation. Land surface models eventually became more dynamic, interacting with the atmosphere in model simulations. However dynamic C cycling did not appear until the 1990's, with

the advancement of simulating primary productivity in ESMs through a more mechanistic understanding of photosynthesis and stomatal conductance (e.g. implementing the model from Farquhar *et al.* (1980)). This created a dynamic linkage between the atmospheric C cycle and net primary productivity (NPP) of the biosphere, which then made C available to move through other ecosystem processes (e.g. growth in different vegetation structures, litter inputs, soil decomposition) (Pitman, 2003). For the sake of simplicity, initially ESMs did not include coupled feedbacks between climate and terrestrial C cycling, but instead simulated climate change effects on soils separately from climate change forecasting models (Jenkinson et al., 1991; Schimel et al., 1994). However the need to couple soil models with both climate and vegetation models was well recognized (Jenkinson et al., 1991).

When coupled atmospheric and terrestrial C cycling models did develop, the resulting predictions suggested the potential for soils to accelerate climate change (Friedlingstein et al., 2001). These predictions were largely driven by relatively simplistic approaches to terrestrial C cycling, wherein the effects of increased CO<sub>2</sub> concentrations on photosynthesis surpassed temperature effects on microbial respiration (Cox et al., 2000), with high uncertainty in the latter (Jones et al., 2003). Subsequent ESM simulations and multi-model comparisons included some form of SOM modeling, varying by the number of soil pools being modeled but generally using 1<sup>st</sup> order kinetics (Section 2.2, Figure 2-1) (Friedlingstein et al., 2006; Krinner et al., 2005; Sitch et al., 2003). ESMs have since been developed to link C and N cycling (long-recognized as an important component for global SOM modeling (Schimel et al., 1994)) as well as the simulation of land use and land cover changes (LULCC) with climate change (Friedlingstein and Prentice, 2010; Lawrence et al., 2011; Wang et al., 2013).

Feedbacks between nutrient cycles at global scales continue to be an important area of development in ESMs. For example, a terrestrial coupled C-N model emphasized the importance of including demand-driven biological N fixation to more accurately simulate C-N interactions in terrestrial systems at a global scale (Esser et al., 2011). Another study highlighted phosphorous (P) dynamics,

particularly in P-limited tropical ecosystems, as an important component missing from most global models (Yang et al., 2014). Spin-up simulations to initialize C pools within ESMs also remain a major bottleneck in the implementation of more complex SOM models, although there is ongoing work to decrease the computational intensity of this component of ESM simulations (Wang et al., 2013; Xia et al., 2012). In one study, global SOM model simulations have been shown to improve with the use of observed rather than internally simulated litter decomposition, with some models showing convergence over long-term projections with climate change impacts (William R. Wieder et al., 2014). More recently, ESMs have been developed to include soil C and N cycling in deeper soil layers (discussed in Section 2.3.4), which while not capturing SOC age dynamics adequately, did improve long-term simulations of SOC responses to climate change (Koven et al., 2013).

As ESMs advance and incorporate SOM models with higher levels of complexity, there are expanding opportunities to implement and test of new ideas in SOM modeling, including many discussed earlier in this chapter. Clearly the debate surrounding SOM dynamics under increasing temperatures, discussed in Section 2.3.3, can and should link to ESM simulations both for hypothesis testing as well as improving ESM simulations under climate change scenarios. The implicit versus explicit inclusion of microbial biomass and processes, discussed in Section 2.3.1, is another example. As mentioned above, 1<sup>st</sup> order kinetic SOM models remain widely used in ESMs. However microbial dynamics have many direct and indirect linkages to land-atmosphere carbon exchange, with complex interactions and potential for community shifts that can lead to positive or negative atmospheric feedbacks (Allison et al., 2010; Bardgett et al., 2008; He et al., 2010). The implicit inclusion of microbial growth efficiency (MGE) controls on SOM dynamics (Wieder et al., 2013; Xu et al., 2014), as well as explicit microbial biomass and microbial C:N stoichiometry would improve model predictions (Fujita et al., 2014; Todd-Brown et al., 2012). Simulating MGE did improve global simulations of measured SOC

variability despite poor performance in tropical and Arctic ecosystems (Wieder et al., 2013). Incorporating microbial dynamics in the Community Land Model (CLM) also changed global projections of SOC dynamics with warming temperatures and increased C inputs from CO<sub>2</sub> fertilization on plant growth, resulting in larger SOC loss from the former and minimal SOC gain from the latter as compared with prior ESM projections (Wieder et al., 2013). However adding an explicit microbial biomass pool to ESMs remains a challenge, in terms of both model complexity and data availability. For example, globally scaled distributions of microbial biomass that could be used to inform or test ESM predictions have only recently been estimated and were derived from highly variable global data unevenly distributed across biomes and hemispheres (Serna-Chavez et al., 2013; Xu et al., 2013). It has been suggested that SOM models incorporating more mechanistic approaches to microbial processes may be better suited to capture transient change in large-scale ESM simulations (Manzoni and Porporato, 2009; Schimel, 2013). There is certainly appeal to the use of microbial enzyme kinetics and stoichiometric constraints to derive a better, simpler 'first principles' approach to SOM dynamics that is analytically tractable and might emulate the early success of simulating photosynthesis and stomatal conductance in ESMs (Bonan et al., 2012; Davidson et al., 2014; Pitman, 2003). However these types of models (e.g. the DAMM model) are still in the early stages of development and testing.

Ultimately, ESMs pose a central challenge to SOM modeling and model development: that is, they test the ability of SOM models to predict accurately across temporal and spatial scales. This arises from both the varying scale of ESM application—e.g. short to long term past and future projections across regional to global spatial scale—as well as the varying scale of the processes and the data used to develop, run, and evaluate simulations (Ostle et al., 2009). Integrating SOM research from smaller-scale experimental studies and model development into coupled global models (e.g. dynamic global vegetation models, (Arneth et al., 2009)) that can then link to ESMs requires determining how best to scale up in spatial resolution while scaling down and simplifying model processes as much as is required

(Figure 2-5). Deciding how best to integrate any of these components across varying process complexities and spatial scales remains difficult. Early analyses showed that while simple versus more complex SOM models in ESMs converged over long periods of time, the complex models more accurately captured transient changes (Schimel et al., 1994). More recently, the relative uniformity in using 1<sup>st</sup> order kinetics SOM modeling in ESMs has been shown to result in better performance at large temporal and spatial scales, but poorer performance at smaller scales due to either poor data availability, high levels of uncertainty, or poor representation of ecosystem properties/dynamics important to capture finer-scale dynamics (Todd-Brown et al., 2013). It still remains unclear whether explicit modeling of microbial biomass and processes instead of using 1<sup>st</sup> order kinetic SOM modeling approaches in ESMs lead to improvement in predictive accuracy across scales (Todd-Brown et al., 2012).

Answering these questions requires multi-model ESM performance comparison with validation datasets that range from regional to global scales, a need that has been recognized but remains in early developmental stages (Luo et al., 2012). For example, data from the C4MIP multi-model comparison suggest the potential for increased C storage in high northern latitudes due to increased primary production and rates of litter turnover, but recognizes that these estimations are missing a substantial list of potentially important interactive factors (e.g. C in permafrost and peatlands, impacts of land use disturbance, N cycling) (Qian et al., 2010). Multi-model comparisons (Cramer et al., 2001; Friedlingstein et al., 2006; Qian et al., 2010) and simulations that interchange key submodels to test model sensitivity have been a useful tool for ESM development (Meehl et al., 2004). As a high profile example, the Intergovernmental Panel on Climate Change (IPCC) developed ensembles of Earth system Models of Intermediate Complexity (EMICs) for a more robust approach to simulate past and future climate change (Randall et al., 2007). SOM model development could continue to benefit from involvement in these efforts (C4MIP now incorporated into CMIP5, for example, Table 2-3). SOM model development at these scales need to consider multiple model structures and variance in input data to reduce either

model-specific or data aggregation biases (Paustian et al., 1997), alongside comparisons between different modeling approaches to better understand the range of potential soil and ecosystem responses to climate change (Esser et al., 2011; Friedlingstein et al., 2006; Qian et al., 2010).

A number of ESM model comparisons, parameter validation, and benchmarking projects are ongoing (Table 2-3), with organizations like the World Climate Research Programme providing resources to promote project development. Researchers have repeatedly recognized that, for individual submodels or in its entirety, ESMs are fundamentally an interdisciplinary effort (Bonan et al., 2002; Pitman, 2003). Given the importance of climate change prediction across temporal and spatial scales, alongside the increasing sophistication in developing and evaluating climate change scenarios for mitigation and adaptive measures (Moss et al., 2010), we would like to emphasize the importance of this area of research and the need for collaboration between ESM researchers, SOM model developers, and SOM field and laboratory researchers alike to advance predictive accuracy through model-data integration. Expanding the network of scientists involved in either providing data for or advancing projects like the coupled model inter-comparison project (CMIP5) could lead to more rapid advances in understanding climate-soil interactions in the context of the global carbon cycle, in addition supporting to better predictions of future climate change.

## 2.4 Conclusions

We began this chapter by providing a brief background to SOM modeling, giving an overview of its history, philosophy, and foundational concepts. With this background we aimed to clarify the enduring value of the factors of soil formation as an overarching framework in SOM modeling. We also emphasized recent advances in SOM measurement methods and data-model integration, two areas with the potential to revolutionize the field in coming years.

We then highlighted recent advances in five areas of SOM model development: SOM stabilization mechanisms, saturation kinetics, temperature sensitivity, dynamics in deep soils, and

incorporation in earth system models. Across these five areas some challenges are universal, including in particular the integrating of understanding across varying scales. We support the view, suggested by others (Manzoni and Porporato, 2009; Stockmann et al., 2013), that SOM model use for hypothesis testing must be done with careful consideration of both the scale at which the model was developed, as well as the scope of the underlying questions being addressed. However, within these five areas some challenges are specific. SOM saturation kinetics, for example, are supported by both laboratory and field experimental data and additionally are relatively well understood from a SOM modeling standpoint. Yet it has seen extremely limited implementation in SOM models, due to its potential to drastically increase SOM model complexity while yielding SOM model improvements only in a narrow set of ecological conditions. Modeling SOM temperature sensitivity and deep soil dynamics, on the other hand, are more sharply limited by data availability, as temperature effects are difficult to isolate, while deep soils are difficult to access. In earth system modeling, however, the strongest limitations come from both the availability of sufficient computational technology to accommodate SOM models with higher complexity, as well as from varying data availability to drive SOM models for globally-scaled analyses. Understanding both the universal and the specific challenges in these areas of SOM model development can help target research to most effectively advance the field.

We see SOM research entering an exciting time. New measurement methods reveal new insights for the relationship between SOM's chemical nature, spatial distribution, and dynamics in the soil environment. Advances in computational capacity and development of collaborative networks for data sharing, management, and data-model integration increasingly relieve the bottlenecks in advancing the conceptual understanding of SOM. These efforts provide better environments to apply SOM models and test hypotheses for SOM dynamics across scales. As Pitman (2003) described more than a decade past (in regards to land surface modeling) the breadth of knowledge required to tackle comprehensive analysis using models of such complexity is vast, and the best approach is through multidisciplinary

collaborations. We recognize, alongside others (Stockmann et al., 2013), the continual importance of collaborative, interdisciplinary effort in SOM modeling.

Within this context there is also room for more openly sourced involvement in model-data integration. Data management is an increasingly sophisticated branch of research and development. Particularly if NSF funded, projects must generally make data management and the open provision of data a component of proposals and final products from research efforts. Libraries are developing capacity to house citable datasets, with standardized approaches to metadata and organizational structure. We believe data-model integration has only touched the surface of what is feasible, given more openly source collaborative networks of data sharing and model-data integration.

We urge the importance of considering scale in all of these developments, using general categories of microsite, site, regional, and global to frame the applicability of the SOM concepts being developed and tested. Due to overlapping levels of complexity, and the many ways in which data and models can interact, keeping consistency and clarity in underlying assumptions is particularly difficult. More standardized frameworks for data sharing and model-data integration will resolve this challenge. Given the continual and expanding importance of SOM models in understanding and managing soils as a natural resource we see this as a challenging, but fruitful, path forward.

Table 2-1 A sele	ection of	SOM	reviews
------------------	-----------	-----	---------

Publication	Description			
Oades, (1988)	"The retention of organic matter in soils"- clarifies terminology for soil C cycling and the linkage between the term 'humus' and 'biomass'. Emphasis on physical stabilization and soil matrix interaction mechanisms that impact rates at which SOM is mineralized.			
Sollins <i>et al.,</i> (1996)	"Stabilization and destabilization of soil organic matter: mechanisms and controls" - A conceptual framework for SOM dynamics aiming to bring together the state of the arring terms of mechanisms underlying SOM stability.			
Falloon & Smith, (2000)	"Modeling refractory soil organic matter"- review of soil organic matter models- including static, dynamic, organismal- or process-based- approaches to modeling SON with the longest turnover times.			
Kuzyakov <i>et al.,</i> (2000)	"Review of mechanisms and quantification of priming effects"- consideration of SOM priming, and its potential to increase or slow SOM turnover rates. Additional: Fontaine <i>et al.</i> , (2007) – "Stability of organic carbon in deep soil layers controlled by fresh carbon supply"			
Six et al., (2002)	"Stabilization mechanisms of soil organic matter: Implications for C-saturation of soils"- evaluates mechanisms for SOM protection that have potential for saturation, proposing a fractionation method to evaluate soils for saturation kinetics.			
Lützow <i>et al.,</i> (2006)	"Stabilization of organic matter in temperate soils: mechanisms and their relevance under different soil conditions – a review"- Methodical discussion of all SOM stabilization mechanisms occurring in temperature soils, identifying uncertainties and inconsistencies. Questions inherent chemical recalcitrance as a stabilization mechanism.			
Trumbore, (2009)	"Radiocarbon and soil carbon dynamics"- A review of radiocarbon dating as an integrated measure of SOM cycling processes and turnover time over long-term timescales, centered on a state-factor approach to test hypotheses for SOM dynamics and showing differences in cycling dynamics across ecosystem components and timescales. Additional: Paul <i>et al.</i> , (1997)- "Radiocarbon dating for determination of soil organic matter pool sizes and dynamics".			
(Kleber, 2010)	"What is recalcitrant soil organic matter"- Complete examination of the term 'recalcitrant' in terms of general, mechanistic, and operational definitions. Examines logic for recalcitrance as a concept, arguing that it is largely semantic rather than providing meaningful connection to SOM dynamics.			

Model	Timestep, Simulation Timeframe	Depth	Drivers	SOM	DOM	Roots	Biotur- bation
SOLVEG-II (Ota et al., 2013)	0.25hr timestep, immediate to long-term simulations	5.5m, 27 varying layer depths	Temp., moisture, soil texture	1 <sup>st</sup> order, CENTURY-type structure, 3 pools each layer (active, slow, passive)	<i>Explicit</i> , f(diffusion, advection, uptake, water flow, decomp)	Exponential function across depth	NA
<b>C-TOOL</b> (Taghizad eh-Toosi et al., 2014)	Monthly, medium to long-term simulations	100cm, 0 – 25cm & 25- 100cm	Temp, clay content, soil C/N, OM inputs	1 <sup>st</sup> -order, 3 pool	Implicit	Exponential function across depth	NA
SOMPROF (Braakhek ke et al., 2013)	Monthly, Medium to long-term simulations	0.7 – 2m, variable layer depths	Temp, moisture	1 <sup>st</sup> order, root & fragmented litter, leachable & non-leachable slow OM in mineral layer	<i>Implicit,</i> effective advection with liquid tranport	1 <sup>st</sup> order decay	Single rate, with diffus. trans- port
RothPC-1 (Jenkinson and Coleman, 2008)	Monthly, Medium to long-term simulations	92cm: in 0-23, 23-46, 46-69, and 69- 92cm depth increme nts	Temp, moisture, clay content	1 <sup>st</sup> order, including microbial biomass, humus, and inert OM pools	Implicit	1 <sup>st</sup> order decay, implicit in C flows through 'decompos- able' versus 'resistant' plant OM pools	NA

Table 2-2 Comparison of a selection of SOM models simulating deep soil dynamics.

Name	Description	Status and Resources	Select Contributions	Publications
International Land Model Benchmarking Project (ILAMB)	Model-data integration and intercomparison; develop and promote benchmarks; software system for benchmarking	Ongoing, www.ilamb.org	Identify ecosystem benchmarks for measured/modeled comparison, to reduce equifinality problem	(Luo et al., 2012)
Project for Intercomparison of Land surface Parameterization Schemes (PILPS)	Landsurface process modeling intercomparison of parameters; community- based documentation, comparison, and validation of parameters	Designed to be on- going, listed as a former model intercomparison project by World Climate Research Programme, http://www.wcrp- climate.org/	Comparison of 1 <sup>st</sup> vs 2 <sup>nd</sup> generation models showed improvement when plant-soil interactions with atmosphere are dynamic, instead of passive; developing use of isotopes to improve parameterization	(Henderson- Sellers, 2006; Pitman, 2003)
Coupled Carbon Cycle Climate Model Intercomparison Project (C4MIP), now incorporated in the Coupled Model Intercomparison Project Phase 5 (CMIP5)	Isolate feedback between the carbon cycle and the climate in the presence of external forcing	Ongoing, http://c4mip.lsce.i psl.fr/, focus area within 5 <sup>th</sup> Coupled Modeling Intercomparison Project (CMIP5): e.g. http://journals.am etsoc.org/page/C4 MIP	10 model comparison suggests high northern latitudes (poleward of 60degrees N) will be C sink to 2100, under warming and increased CO <sub>2</sub>	(Friedlingstein et al., 2006; Qian et al., 2010)

Table 2-3 Selection of model intercomparison and model-data integration projects focused on carbon cycling and including a soils component.



Figure 2-1 First order single and multi-compartment models (a & b, respectively) versus non-first order multi-compartment models showing compartment feedback on the rate of loss and compartment feedback on the rate of transfer between compartments (dotted arrow, c & d, respectively). Boxes indicate model states.



Figure 2-2 Theoretical relationship between C input level and soil organic C (SOC) contents at steadystate, with and without C saturation. Steady-state SOC accumulation dynamics expressed over time (a) produces a linear relationship when expressed over C input level (b) Under the conditions of C saturation, SOC stabilization with increasing input rates (at steady-state) is not proportional (c) resulting in an asymptotic relationship when expressed over C input level (d). Reprinted with permission from (Stewart et al., 2007).



Figure 2-3 Hypothetical decomposition response curves with increasing temperature, showing greater temperature sensitivity of  $R_2$  versus  $R_1$ , as exemplified by  $\Delta R_2$  being larger in magnitude than  $\Delta R_1$  over the same temperature interval.



Figure 2-4 Conceptual diagram showing factors that can interact with atmospheric CO<sub>2</sub> emissions and soil respiration under climate warming, with hypothesized positive and negative feedbacks as originally described by Luo et al. (2001) (A), versus new approaches that separate mechanisms that affect microbial processes from mechanisms that affect SOM availability (B, vis à vis (Conant et al., 2011)). Colored boxes and lines can be considered a plausible scenario under increasing temperatures. Many other interactions are possible.



Figure 2-5 Conceptual grouping of carbon models across scales, in terms of process versus application, where GCMs refers to global circulation models (e.g. earth system models), and DGVMs refer to dynamic global vegetation models. Reprinted with permission from Ostle et al. (2009).
# 3 TRACKING THE FATE OF LITTER C INTO CO<sub>2</sub>, MICROBIAL PRODUCTS AND DISSOLVED ORGANIC CARBON BY THE LITTER DECOMPOSITION AND LEACHING (LIDEL) MODEL USING A HIERARCHICHAL BAYESIAN APPROACH TO ESTIMATE PARAMETERS AND VARIABILITY<sup>2</sup>

## **3.1 Introduction**

Litter decomposition creates a dynamic link between atmosphere and terrestrial carbon C (Houghton, 2007), tying into concerns with the interaction of these two main global C pools in the context of global climate change (Friedlingstein et al., 2006, 2001). Decomposers interact dynamically with the physical characteristics and chemistry of litter, as well as with other biotic and abiotic factors (e.g. climate) through time. These interactions result in different rates of litter decomposition that have the potential to generate different types and quantities of decomposition products (e.g. Don and Kalbitz, 2005). Litter decomposition is therefore a fundamental mechanism determining the rate and partitioning of litter C—i.e. atmospheric CO<sub>2</sub> fixed via photosynthesis in terrestrial primary productivity –between returning to the atmosphere quickly as respired CO<sub>2</sub> and remaining in soils as partially decomposed litter or microbially-processed OM. Accurately representing litter decomposition in ecosystem and global C cycle models is therefore necessary to project climate change impacts, both to anticipate specific ecosystem changes as well as predict the magnitude and direction of feedbacks that affect global C dynamics.

Litter decomposition occurs through three general processes: fragmentation, catabolism to  $CO_2$ , and leaching (Swift et al., 1979). Current litter decomposition models are generally based on litter mass loss and  $CO_2$  flux (e.g. Adair et al., 2008; Manzoni et al., 2012), an approach which does not consider leaching and the generation of dissolved organic matter (DOM). Dissolved organic matter can make up 6-39% of litter decomposition products (Don and Kalbitz, 2005; Magill and Aber, 2000; Qualls et al.,

<sup>&</sup>lt;sup>2</sup> In preparation for *Soil Biology and Biochemistry* with W.B. Parton, J.L. Soong, N.T. Hobbs, M.F. Cotrufo, and K. Paustian.

1991). It is also recognized as an important near-term mechanism for vertical OM movement through soils, with the potential to contribute towards long-term SOM persistence by adsorption to soil minerals (Kaiser and Kalbitz, 2012). While some models simulate DOM movement in soils (Braakhekke et al., 2013; Jenkinson and Coleman, 2008; Ota et al., 2013; Tipping et al., 2012), there has been limited attention to the dynamics of DOM generation during litter decomposition. Studies show DOM generation can vary in quantity and chemistry between litter types and through time, in patterns that are often unrelated to cumulative litter mass loss dynamics (Don and Kalbitz, 2005; Soong et al., 2015). A dynamic model for DOM generation from litter decomposition is needed to examine the implications of these patterns on OM cycling, either at a fine scale—e.g. within a specific soil profile—or more generally in global biogeocycling processes.

Microbial carbon use efficiency (CUE)—the quantity of microbial biomass generated per unit of substrate use (Lekkerkerk et al., 1990)—is another important mechanism in litter decomposition, specifically within the process of catabolism to CO<sub>2</sub>. Microbial CUE determines the partitioning of litter C between respired CO<sub>2</sub> versus the formation of microbial biomass and products. Microbial CUE is related to DOM generation and movement, as DOM shows increasing evidence of microbial processing across the vertical soil profile (Kaiser and Kalbitz, 2012; Tipping et al., 2012). However, to our knowledge this interaction has not been examined in a litter decomposition model.

Evidence suggests microbial CUE is often variable during litter decomposition, depending on the decomposition environment, chemical characteristics of substrates being decomposed, and the physiological characteristics of microbes and microbial communities (Frey et al., 2013; Lekkerkerk et al., 1990; Manzoni et al., 2008). However many widely used models simulate CUE as a static variable (e.g. DAYCENT (Parton, 1988)). This is an important area for litter decomposition model development, as evidence supports microbial products as an important contributor to SOM formation (Miltner et al., 2011; Schmidt et al., 2011). Theoretical implications of variable CUE in SOM formation are summarized

in the Microbial Efficiency-Matrix Stabilization (MEMS) framework, proposing potential for greater persistence of litter OM that is more efficiently decomposed due to greater generation of microbial material that can then be stabilized in the soil matrix (Cotrufo et al., 2013). The MEMS hypothesis is suggested as an alternative to past hypotheses linking SOM stability to inherent litter chemical recalcitrance, that have since been strongly criticized (Dungait et al., 2012; Kleber, 2010; Kleber et al., 2011). Litter decomposition models that include variable CUE would more accurately reflect partitioning of decomposed litter C to CO<sub>2</sub> *versus* microbial materials, and thus be able to test new hypotheses linking microbial processes to SOM persistence (Wieder et al., 2014).

In this paper we propose the Litter Decomposition and Leaching (LIDEL) model as a new approach to modeling litter decomposition that 1) includes explicit modeling of DOM as a litter decomposition product, and 2) dynamically links substrate chemistry with variable CUE and the generation of DOM and other litter decomposition products. In the LIDEL model, we focus on the control of CUE by litter N and lignin content. Nitrogen, whether internally (e.g. as a component of plant litter) or externally (e.g. as present in the surrounding environment) available, is an important driver of litter decomposition dynamics (Adair et al., 2008; Aerts, 1997; Manzoni et al., 2008; Sinsabaugh et al., 2013). Lignin is a litter chemistry metric that also shows strong explanatory power when incorporated into litter decomposition models (Adair et al., 2008), and has been linked to mechanistic controls on microbial processes during decomposition (Moorhead et al., 2013). The LIDEL model specifically integrates Moorhead et al. (2013) and Sinsabaugh et al. (2013) hypotheses for lignin and N impacts on microbial CUE, with experimental data from Soong et al. (2015) for the impact of litter chemistry on the generation of CO<sub>2</sub> *versus* DOC from litter decomposition.

Linking changes in measured C pools with conceptual understanding of decomposition dynamics, particularly as C moves from the surface into the soil matrix, is a fundamental challenge modeling litter decomposition and soil C dynamics (Elliott et al., 1996). There are a variety of established

methods to optimize model parameters and evaluate uncertainty in model performance in state-space models characteristic of modeling C dynamics (e.g. Del Grosso et al., 2010; Ogle et al., 2010, 2007). However, these analyses are often not satisfactory in accounting for multiple types of measurement data, accommodating different types of measurement and model uncertainty, or integrating these two components in time-series models and observations. Hierarchical Bayesian methods offer an approach to factor highly complex problems with high dimensionality into problems with lower dimensions, specifically by separating the problem into the process model (e.g. the LIDEL model), data models, and associated parameter models, and having uncertainty associated with each component (Berliner, 1996; Clark, 2007; Geremia et al., 2014). Bayesian methods for model-data integration provides a powerful approach to estimate unknown model parameters and compare C cycling models (Tuomi et al., 2008), clarifying how measured data connect to ecological hypotheses (Hararuk and Luo, 2014). Bayesian methods can also help identify areas where data are needed or where additional data would help clarify the understanding of underlying dynamics (Braakhekke et al., 2013). Hierarchical Bayesian methods have further been used to successfully partition variability in C cycling dynamics (Ogle and Pendall, 2015). In the hierarchical Bayesian framework it is also possible to separate model error from observation error if we use informative priors on error terms, for example if error in a specific measurement technique is well understood (Hobbs and Hooten, in press).

In this paper we present the LIDEL model structure and assumptions, as well as parameter estimates using hierarchical Bayesian methods (Berliner, 1996; Clark, 2007; Geremia et al., 2014) to integrate LIDEL model simulations with data from an experiment measuring the generation of decomposition products from five litters that varied in initial lignin and N content (Soong et al., 2015). We explore separate hypotheses for N control on litter decomposition and the efficiency of DOC generation from litter fractions. Specifically we examine 1) a logistic *versus* a linear relationship between N content, N controls on litter decomposition rates, and microbial CUE, to test whether N limitation is

better modeled as occurring suddenly at a threshold, or as occurring more gradually as N content decreases. We also examine 2) equal DOC/CO<sub>2</sub> generation efficiency from both the soluble and cellulose-dominated litter fractions *versus* higher DOC generation from the soluble litter fraction. The combination of these alternate hypotheses for 1) N limitation and 2) DOC/CO<sub>2</sub> generation efficiency yielded four separate LIDEL model formulations. Parameter estimates and overall model predictive accuracy were compared between these four LIDEL model formulations, to identify the model structure that best predicted measured litter decomposition dynamics.

#### 3.2 Material and methods

### 3.2.1 LIDEL model description

In the LIDEL model (Figure 3-1) we propose separating litter into three C fractions including: water soluble, lignin, and non-lignin structural (NLS). A similar fractionation approach separating plant litter into lignin, labile, and cellulose pools has been successfully applied in other models to include the impacts of microbial growth efficiency on C dynamics, yielding improved global simulations of soil C (Wieder et al., 2013). However we use water soluble litter instead of labile litter to better link to DOC and the ecological importance of water in generating and moving this OC fraction. We model lignin, similarly to Weider et al. (2013), in order to explicitly implement the Moorhead et al. (2013) hypotheses for lignin control on CUE, described below. The remaining litter material is estimated by difference and thus termed non-lignin structural. However we assume that the NLS fraction is dominated by cellulose and hemicellulose—termed holocellulose—and treat it as such when implementing Moorhead et al. (2013) hypotheses for lignin controls on decomposition. We therefore refer to the NLS pool using the more familiar term holocellulose, even though technically by our definition this pool will contain a small quantity of non-holocellulose material.

In order to incorporate dynamic microbial CUE impacts on the generation of microbial biomass and microbial products we model litter material movement into microbial biomass and then microbial

products pools, with losses to DOC and CO<sub>2</sub>. It should be noted that lignin does not directly contribute to microbial biomass, but only decomposes to CO<sub>2</sub> and DOC. This is drawn from the Moorhead et al. (2013) hypothesis that lignin is not energetically favorable for microbes to decompose, and is therefore decomposed to access holocellulose without yielding microbial biomass directly (Moorhead et al., 2013). It should also be noted that microbial products only contains structural microbial material, as soluble microbial products from microbial biomass will be lost as DOC (Figure 3-1).

The mathematical representation of Figure 3-1 is expressed as a system of seven ordinary differential equations, where:

$$\frac{DC_3}{dt} = -k_3C_3 , \qquad \qquad \text{Eqn 3}$$

$$\frac{DC_4}{dt} = -k_4 C_4 + \mu_\beta \beta_2 (1 - \lambda_1) \mu_k k_2 C_2 + \mu_\beta \beta_1 (1 - \lambda_4) \mu_k k_1 C_1 , \qquad \text{Eqn 4}$$

$$\frac{DC_5}{dt} = -k_5C_5 + \beta_3(1 - \lambda_2)k_4C_4,$$
 Eqn 5

$$\frac{DC_6}{dt} = \lambda_1 \mu_k k_2 C_2 + \lambda_3 k_3 C_3 + \lambda_2 k_4 C_4 + \lambda_3 k_5 C_5 + \lambda_4 \mu_k k_1 C_1 \text{ , and}$$
Eqn 6

$$\frac{DC_7}{dt} = \left( (1 - \mu_\beta \beta_1) (1 - \lambda_4) \right) \mu_k k_1 C_1 + \left( (1 - \mu_\beta \beta_2) (1 - \lambda_1) \right) \mu_k k_2 C_2 + (1 - \lambda_3) k_3 C_3 + \left( (1 - \beta_3) (1 - \lambda_2) \right) k_4 C_4 + (1 - \lambda_3) k_5 C_5 .$$
Eqn 7

The LIDEL model runs on a daily time step. Parameters  $k_1$  and  $k_2$  are the maximum rates (day<sup>-1</sup>) of water soluble and holocellulose decay, and are parameters fitted in our analysis. The rate (day<sup>-1</sup>) of microbe biomass decay ( $k_4$ ), the generation of microbial products from microbe biomass ( $\beta_3$ ), and the generation of DOC from the decay of microbe biomass ( $\lambda_2$ ) are also fitted parameters in our analysis. Parameters  $\beta_1$  and  $\beta_2$  are the maximum growth efficiencies (g microbial biomass/g decayed material) for microbial use of water soluble and holocellulose litter, set at values of 0.6 and 0.5, assuming the maximum CUE stipulated by Sinsabaugh et al. (2013) for soluble C and assuming a lower maximum for

decomposition of holocellulose (Figure 3-2b). The remaining parameters are based on hypotheses described in the following sections.

### 3.2.1.1 Linking hypotheses for microbial CUE and litter decay

We differentiate individual N and lignin effects in the LIDEL model by setting a maximum CUE limit and a threshold for N effects on CUE (Moorhead et al., 2013; Sinsabaugh et al., 2013). Lignin affects litter decomposition, as well as microbial CUE, by limiting microbial access to usable plant material (i.e. holocellulose) (Herman et al., 2008; Manzoni et al., 2008; Moorhead et al., 2013; Sinsabaugh et al., 2013). We adapted these hypotheses to set lignin and N limitations on decomposition and CUE in the LIDEL model, as based on the initial litter N content and the dynamic lignocellulose index (L<sub>c</sub>), calculated as;

$$L_{c} = \frac{lignin(C_{3})}{lignin(C_{3}) + NLS(C_{2})}.$$
 Eqn 8

Based on prior publications (Moorhead et al., 2013; Sinsabaugh et al., 2013) and DOC generation dynamics in the Soong et al (2015) experiment we hypothesized that the water soluble and holocelluose pools are affected by both L<sub>c</sub> and N. We assumed the litter chemistry (i.e. either L<sub>c</sub> or N) exerting the greatest limitation at any given point in time during decomposition is, at that time point, controlling decomposition dynamics and microbial uptake of these pools. Therefore;

$$\mu_k = minimum(\gamma, \varepsilon_k)$$
, and Eqn 9

$$\mu_{\beta} = minimum(\gamma, \varepsilon_{\beta}),$$
 Eqn 10

where  $\mu_k$  is the litter chemistry limitation on the maximum rate of decay for the water soluble and holocellulose fractions,  $\mu_\beta$  is the litter chemistry limitation on the maximum rate of microbial uptake of these pools. The value for  $\mu_k$  is determined by the minimum of  $\gamma$  — the rate modifier based on N limitation (Figure 3-3)—*versus*  $\varepsilon_k$ , a rate modifier for L<sub>c</sub>-dependent limitation on decay. The value for  $\mu_\beta$ is determined by the minimum of  $\gamma$  *versus*  $\varepsilon_\beta$ , a rate modifier for L<sub>c</sub>-dependent limitation on CUE (Figure 3-2). Moorhead et al. (2013) propose microbes do not decompose lignin when L<sub>c</sub> <0.4, when holocellulose is highly available for microbial use. They further propose a linear increase in rates of lignin decay accompanied by a linear decrease in rates of holocellulose decay when L<sub>c</sub> is between 0.4 – 0.7. Finally, they propose an equal decay rate of both pools when L<sub>c</sub> > 0.7. We approximated these equations with a continuous equivalent (Figure 3-2). We expressed  $\varepsilon_k$  as;

$$\varepsilon_k = e^{-3.0*L_c}.$$
 Eqn 11

We expressed the control of L<sub>c</sub> on water soluble and holocellulose CUE ( $\varepsilon_{\beta}$ ) when L<sub>c</sub> is less than 0.7 as:

$$\varepsilon_{\beta} = 1 - e^{-0.7(|L_c - 0.7| + 10)}$$
. Eqn 12

We expressed the control of  $L_c$  on the rate of lignin decay ( $k_3$ , in units of day<sup>-1</sup>), based on the holocellulose maximum rate of decay ( $k_2$ , in units of day<sup>-1</sup>), as:

$$k_3 = k_2 \frac{0.2}{(1 + \frac{200}{e^{8.15L_c}})}.$$
 Eqn 13

Microbial products are hypothesized to be an important contributor to SOM formation (Cotrufo et al., 2013), but are difficult to measure (Preston et al., 2009), and their dynamics remain poorly understood. We assume microbial products are decomposed slowly, and do not directly contribute to the generation of microbial biomass (Figure 3-1). We expressed the rate of microbial products decay ( $k_5$ , in units of day<sup>-1</sup>), for simplicity (i.e. without adding additional model parameters), as similar to the maximum possible rate of lignin decay, where:

$$k_5 = k_2 e^{-2.1}$$
. Eqn 14

Sinsabaugh et al. (2013) link substrate N content to microbial CUE, providing evidence for a threshold for N availability above which N does not limit CUE. Based on this hypothesis, we propose that above 3% N, either in litter or as available to microbes from external sources (e.g. soil, which we did not consider in this analysis), CUE and litter decomposition are unaffected by N availability. However, below that threshold we explore two mathematical expressions of N limitation:

$$\gamma_1 = \frac{1}{1 + e^{-N_{Max}(\tau - N_{mid})}} \phi, \text{ and}$$
Eqn 15

$$\gamma_2 = N_{min} + \frac{1 - N_{min}}{N_{Max}} \phi, \qquad \qquad \text{Eqn 16}$$

where  $\gamma$  modifies rates of decay from the water soluble and holocellulose C pools, as well as the rate of C uptake by microbes. In both Eq.15 and 16,  $N_{Max}$  is the maximum percent litter N for a N effect while  $\phi$  is the internally available initial litter percent N. In Eq. 15,  $N_{mid}$  determines the midpoint of the N limitation curve. In Eq. 16,  $N_{min}$  is the intercept determining minimum N limitation.  $N_{min}$  can be considered a measure of external N availability, and could be an estimated parameter with application of the LIDEL model in systems that vary by both litter N and externally available N (e.g. connecting the LIDEL model to a model of soil organic matter dynamics) (Figure 3-3).

#### 3.2.1.2 Hypotheses from experimental data: DOC versus mass loss

Data from the litter decomposition experiment by Soong et al. (2015) were used to develop the remaining structure of the LIDEL model. In this experiment five litter types were selected that exhibited a range of initial % N and initial L<sub>c</sub> (Table 3-1), in order to examine their relative impacts on the character and quantity of DOC and CO<sub>2</sub> generation across a 365 day incubation. Each litter type was exposed to a microbial inoculum and allowed to decompose in absence of soil and without any additional N, with periodic leaching events that were more frequent in early stages of decomposition (approximately every 1-3 days, then approximately weekly, then approximately every 1-2 months). Measurements of DOC and CO<sub>2</sub> were taken at each leaching event, and measures of litter mass loss were taken by destructive harvesting half of the samples at day 95 and the remaining half at the conclusion of the experiment on day 365 (for more details about the experiment see Soong et al.,2015). This experiment provides a unique dataset explicitly evaluating the fractionation of DOC generation *versus* CO<sub>2</sub> through time, across litters that vary in their N and L<sub>c</sub> chemistry. We use these data to develop empirically-based hypotheses for the generation of DOC from decomposition from the litter pools included in the LIDEL model.

We hypothesized linear declining relationships between the ratio of DOC versus litter mass loss (i.e.  $DOC/(DOC+CO_2)$ ) as a function of initial litter % N and L<sub>c</sub>, with highest proportion of DOC generation at the lowest initial % N and L<sub>c</sub> values. Based on these hypotheses, we examined the predictive power of initial litter % N and L<sub>c</sub> in estimating DOC generation, across the range of DOC versus litter mass loss observed in the Soong et al. (2015) experiment (0.15 - 0.48 g DOC per g litter mass loss from day 10 - 20, 0.004 - 0.27 g DOC per g litter mass loss from day 20 - 365). We examined the predictive power of these relationships individually as well as by taking the minimum of the DOC versus mass loss predicted by each factor—similar to the approach applied to determine  $\mu_k$  and  $\mu_\beta$  (Eqn's 8 and 9). Predictions using the minimum of the two factors had a higher  $r^2$  than predictions based on either of these factors individually (minimum predicted:  $r^2$ =0.319, slope=0.616, intercept=0.113; L<sub>c</sub> predicted:  $r^2$ =0.149, slope=0.477, intercept=0.111; N predicted: r<sup>2</sup>=0.105, slope=0.269, intercept=0.128), suggesting litter N and L<sub>c</sub> influence different mechanisms for DOC generation during decomposition. Predictions had a substantially higher  $r^2$  as well as a slope closer to 1 and intercept closer to 0 when accounting for the higher proportion of DOC generation between early versus later decomposition (i.e. day 10 - 20 versus day 20 – 365), again using the minimum of predictions from initial litter %N and L<sub>c</sub> (Figure 3-4) (minimum predicted:  $r^2$ =0.862, slope=0.991, intercept=0.018; L<sub>c</sub> predicted:  $r^2$ =0.704, slope=0.896, intercept=0.01; N predicted: r<sup>2</sup>=0.613, slope=0.702, intercept=0.012). The generation of DOC has been linked to the water soluble fraction of litter (Magill and Aber, 2000; Smolander and Kitunen, 2002). This is further supported by Soong et al. (2015) data that show early DOC generation is strongly plantderived, and strongly predicted by the hot water extractable fraction. We therefore assume that early DOC generation is predominantly derived from the litter water soluble fraction, and hypothesize that DOC generated from both the water soluble and holocellulose litter fractions are similarly affected by litter N and L<sub>c</sub>. We express the rate of DOC generation from decomposition of water soluble and holocellulose C ( $\lambda_4$  and  $\lambda_1$ ) as;

$$\lambda_4 = minimum \left( (E_{SMax} - \frac{(E_{SMax} - E_{Smin})}{L_{cMax}} * L_c \right), (E_{SMax} - \frac{(E_{SMax} - E_{Smin})}{N_{Max}} * \phi \right) \right), \text{ and } Eqn 17$$

$$\lambda_1 = minimum \left( (E_{HMax} - \frac{(E_{HMax} - E_{Hmin})}{L_{cMax}} * L_c \right), (E_{HMax} - \frac{(E_{HMax} - E_{Hminm1})}{N_{Max}} * \phi) \right).$$
Eqn 18

where  $E_{Smin}$  and  $E_{SMax}$  are the minimum and maximum leached fractions from the soluble pool,  $E_{Hmin}$ and  $E_{HMax}$  are the minimum and maximum leached fraction from the holocellulose pool, and  $L_{cMax}$  is the maximum L<sub>c</sub> index. We use this component of the LIDEL model to evaluate our hypotheses for DOC generation efficiency from litter decomposition, developing two formulations of the LIDEL model with respect to the maximum and minimum leached fractions from the soluble versus the holocellulose pools. In one formulation these values were set equal to each other, at a minimum and maximum of 0.005 day<sup>-1</sup> and 0.15 day<sup>-1</sup>. In the other formulation the soluble pool values were set higher than the holocellulose pool values, at 0.1 and 0.5 day<sup>-1</sup> *versus* 0.001 and 0.1 day<sup>-1</sup>. The latter approach assumes greater efficiency of plant-derived DOC release from decomposition of the holocellulose pool *versus* the soluble pool.

Finally, in the LIDEL model DOC is also generated from lignin, microbes, and microbial products. The generation of soluble microbial products from microbial biomass turnover is not well understood and is therefore a fitted parameter in this analysis. We assume for the lignin and microbial products pools that only a very small, constant portion of DOC is generated from decomposition. We used measurements of DOC generation from late in alfalfa decomposition (that decomposed very quickly and had very little mass at the end of the incubation) to set this rate, where:

$$\lambda_3 = 0.038 \ g \ DOC/g \ decayed \ material$$
 Eqn 19

### 3.2.1.3 LIDEL initial conditions

The initial conditions of the LIDEL model pools (referred to by  $\tau$  subscripts) were assumed to be determined by the fraction of initial litter material in the soluble, lignin, and DOC pools, with the remaining material in the holocellulose pool. These initial conditions were calculated as;

$C_{1\tau i} = \left(C_{Ti} f_{s_i}\right) - \left(C_{Ti} f_{s_i} f_{DOC}\right),$	Eqn 20
$C_{2\tau i} = C_{Ti} - (C_{Ti} (f_{s_i} + f_{LIG_i})),$	Eqn 21
$C_{3\tau i} = C_{Ti} f_{LIG_{i'}}$	Eqn 22
$C_{4\tau i}=0,$	Eqn 23
$C_{5\tau i}=0,$	Eqn 24
$C_{6\tau i} = C_{Ti} f_{s_i} f_{DOC}$ , and	Eqn 25
$C_{7\tau i}=0$ ,	Eqn 26

where  $C_{Ti}$  is the total initial litter C mass of litter *i*,  $f_{S_i}$  is the soluble fraction for a given litter *i*,  $f_{DOC}$  is the fraction of soluble material that is not microbially processed, and  $f_{LIG_i}$  is the lignin fraction for a given litter *i*.

### 3.2.2 Statistical Approach

#### 3.2.2.1 Data

Data used in this analysis include experimental time series data from the laboratory litter decomposition experiment described in detail in Soong et al. (2015) and literature data (Adair et al., 2008; Parton et al., 2007) for the initial litter chemistry of the five litters; alfalfa (*Medicago sativa*), ash (*Fraxinus excelsior*), big bluestem (*Andropogon gerardii*), oak (*Quercus macrocarpa*) and pine (*Pinus ponderosa*). To initialize the LIDEL model C pools we used Soong et al. (2015) experimental data for measurements of initial litter chemical fractions. The acid unhydrolyzable residue (AUR) measured from the acid detergent fiber (ADF) digestion method (Van Soest and Wine, 1968) used in Soong et al. (2015) is a reasonable approximation of lignin at the initial phase of the experiment when the litters have not been exposed to microbial processing. It therefore was the only source of data used to inform the initial size of the lignin C pool in the LIDEL model. However the connection is less clear between the soluble C pool used in the LIDEL model and either hot water extraction (noted as HWE) or measurement of this fraction as the mass loss during NDF digestion (Soong et al., 2015). Therefore we used both of these measurements from Soong et al. (2015), supplemented with literature values across the five litters to more accurately reflect their variance. Conceptually, the initial value of the DOC pool in the LIDEL model is representative of soluble litter C that is released immediately with exposure to water, without any interaction with microbial processes. We made the assumption that the DOC released in the first 10 days of the experiment is an approximation of this initial DOC value. The initial holocellulose pool was assumed to consist of the remaining initial litter C after removal of the soluble, initial DOC, and lignin litter C (Eq. 21). Full details on these data and their use to initialize the LIDEL model are described in Appendix 1.

We additionally used the series of  $CO_2$ , DOC, and total mass loss measurements taken by Soong et al. (2015) through the 365 day incubation to evaluate LIDEL temporal dynamics. Each time series measurement had three replicates for each litter. Full details on these data and their use to evaluate the LIDEL model are described in Appendix 2.

#### 3.2.2.2 Hierarchical Bayesian Analyses

Using the data described above, we applied hierarchical Bayesian models to use the experimental data to estimate initial LIDEL model pools as well as five parameters that drive LIDEL temporal dynamics. The five parameters include the decay of the soluble  $(k_1)$ , non-lignin structural  $(k_2)$ , and microbial biomass pools  $(k_4)$ , the generation of microbial products from microbe biomass  $(\beta_3)$ , and the generation of DOC from the decay of microbe biomass  $(\lambda_2)$ . These parameters are important for the dynamics of litter decomposition through time as well as the generation of products from microbial processes. However, they are poorly understood in part due to the difficulty in measuring them experimentally.

The LIDEL model includes several equations where the minimum of two values drive model dynamics at a given point in time (Eq.'s 9, 10, 17, and 18). The LIDEL model is therefore non-linear and cannot be solved analytically, requiring the use of an ordinary differential equations solver to resolve model dynamics into daily time steps. We split the data and processes into two hierarchical Bayesian analyses. The first analysis estimated the initial conditions (IC) of the LIDEL model. The second analysis used the results of the first as priors in the estimate of LIDEL temporal dynamics (TD), using the four LIDEL formulations under consideration (linear *versus* logistic N control, Eqns 15 vs 16, and equal *versus* unequal DOC generation efficiency from the soluble and holocellulose pools, Eqns 17 & 18). This approach was meant to reduce the number of estimated parameters in the TD analysis, where the ordinary differential equations solver is required. The TD analysis used the posterior distributions of the IC analysis as priors to initialize LIDEL model pools on day 0. To further simplify the TD analysis a subset of measured time points for DOC and CO<sub>2</sub> were used to estimate a subset of latent states, measurement error and model error.

Since the five main parameters of interest for the TD analysis each have a single value across all litter types and all time points, subsetting time points was considered an appropriate method to reduce computational complexity by reducing the number of estimated values (by reducing the estimated latent states) while still informing the parameters of interest with time series measurements across the decomposition experiment. The subset of time points were selective to be representative of the 'early', 'middle' and 'late' stage decomposition dynamics identified by Soong et al. (2015). The subset of time points included all measurements for all litter types on days 7, 15, 28, 64, 95, and 365. The remaining measurements were used for a non-exhaustive cross validation at each iteration of the MCMC, calculating the root mean square error (RMSE) for the model prediction of all measured time points excluded from the subset. Litter mass remaining measurements were only taken at two time points, on day 95 and day 365. Therefore all mass measurements were used in the TD analysis to estimate model

parameters. RMSE results are therefore an indication of model fit for predicting out-of-sample DOC and CO<sub>2</sub> measurements.

In both the IC and the TD hierarchical Bayesian analyses, latent states and parameter marginal posterior distributions were estimated using a random-walk Metropolis Hastings Markov Chain Monte Carlo (MCMC) algorithm. The analyses were written in R (R Core Team, 2014). We used the Isoda function in the R package deSolve as the ODE solver (Soetaert et al., 2010). Three MCMC chains were run out 100,000 iterations for the IC analysis, discarding the first 50,000 iterations as the burn-in period. In the four LIDEL model formulations considered in the TD analysis, 80,000 iterations were simulated total with the first 40,000 iterations discarded as the burn-in period. Individual chains were assessed visually for convergence using trace and marginal density plots, as well as using Geweke convergence diagnosis Z-scores (Geweke, 1992). The 3 chains were then assessed for convergence using the Gelman-Rubin MCMC convergence diagnostic (Gelman and Rubin, 1992). Once the 3 chains passed convergence tests, summary statistics of model parameters were evaluated (mean and 95% Bayesian credible intervals). All of these convergence diagnoses and summary analyses were completed using the *coda* package in R (Plummer et al., 2006). Details for both the IC and the TD analyses, including the full Bayesian models as well as MCMC algorithms are described in Appendix 1 and 2, respectively.

#### 3.3 Results

Summary statistics for estimates of litter fractions—including the soluble and lignin fractions by litter type as well as the initial DOC fraction—are presented in Table 3-2. Parameter estimates for the variance of the hot water extraction and mass-difference soluble fraction measurements, as well as the final estimated soluble fraction parameters are presented in Figure 3-5. Across litter types, the initial lignin fraction in the LIDEL model—informed by AUR data in the Soong et al. (2015) experiment—shows a narrower width than the other fractions estimated in this analysis (Table 3-2).

The hierarchical Bayesian approach allows litter fractionation parameter variance and uncertainty, estimated in the IC analysis, to be carried into the estimates of LIDEL temporal dynamic parameters. Overall there is high variability in LIDEL estimates (Figure 3-6). This is particularly apparent in LIDEL DOC and CO<sub>2</sub> predictions, as indicated by the width of 95% BCI's (Figure 3-6). These wide ranges are likely related to the variable initial litter fraction estimates yielded from the IC analysis, used to initialize the LIDEL model litter pools. Both the initial soluble and DOC pools are less constrained by data than the initial lignin pool. However in the TD analysis the data suggest the soluble LIDEL pool decomposes much more quickly than the holocellulose pool and is therefore responsible for providing the input that drive DOC and CO<sub>2</sub> generation, particularly early in decomposition (Table 3-3). Variability in the initial soluble pool size for each litter will therefore contribute to variability in model estimates of DOC and CO<sub>2</sub> generation.

Models 1, 2 and 3 had comparable  $k_1$  estimates, although the last was less well informed by the data (as shown by its broader 95% BCI width, Table 3-3). Estimates of  $k_2$  were more tightly constrained, only ranging by 0.0045 day<sup>-1</sup> between all LIDEL model formulations. Microbial biomass turnover was weakly informed by the data but suggests fast turnover, with all estimated means exceeding 0.5 and with values that were closer to the upper range of the 95% BCI than the lower range (indicating a predominance of higher parameter values in the MCMC). Microbial product generation from microbial biomass turnover was very weakly informed by the data in models 1, 2, and 4, and not informed by the data in model 3. DOC from microbial biomass was more strongly informed by the data, showing similar characteristics in models 1 & 2 (higher) versus models 3 & 4 (lower). According to RMSE values the lowest predictive accuracy for out-of-sample DOC and CO<sub>2</sub> measurements was model 4, versus the highest predictive accuracy in models 1 and 3 (Table 3-3). The measured/modeled comparison of estimated time points, however, shows improved mass remaining estimation in models 1 & 2 versus 3 & 4 (Figure 3-6, A & D versus G & J), and improved high DOC predictions in models 2 & 4 versus models 1 & 8

3 (Figure 3-6, E & K versus B & H). It is important to note that all mass measurements from Soong et al. (2015) were used in the TD Bayesian analysis. We only used out-of-sample measurements to calculate RMSE values, which therefore only assesses out-of-sample measured vs. model prediction for DOC and CO<sub>2</sub>. The RMSE values therefore do not factor in the bias in LIDEL mass remaining prediction apparent in Model 3 and 4 results (Figure 3-6, G & J). It is also notable that RMSE is lower with models 1 and 3 versus 2 and 4, whereas in the latter there is a clear improvement in model predictions of high DOC measurements (Figure 3-6, E & K versus B & H). However DOC predictions also become more variable in models 2 and 4, as indicated by the wider 95% BCI's (Figure 3-6).

### 3.4 Discussion

An ongoing debate in modeling terrestrial OM dynamics is the degree to which explicit modeling of microbial biomass and processes is needed to accurately simulate C cycling and ecosystem dynamics (Stockmann et al., 2013; Treseder et al., 2012). This debate is in part centered on accurately capturing ecosystem responses to direct and indirect human impacts, such as global warming and land use change (Wieder et al., 2014). New modeling efforts are aimed to better reflect microbial processes in litter decomposition and SOM dynamics (Davidson et al., 2014; Moorhead and Sinsabaugh, 2006; Wieder et al., 2013; W. R. Wieder et al., 2014). However, there is the potential for increases in model complexity and unrealistic oscillatory behavior with the added feedback between explicit microbial biomass and decomposition dynamics (Wang et al., 2014). Data to drive and evaluate models with explicit microbial biomass and mechanisms are also more limited (Treseder et al., 2012). With the LIDEL model, we look to strike a balance between model complexity and simulating dynamic microbial processes. The LIDEL model includes an explicit microbial biomass pool, is driven with broadly available data, and includes dynamic microbial CUE, but maintains a relatively simple first-order structure by not linking microbial biomass directly to decomposition rates (Figure 3-1).

The LIDEL model incorporates variable microbial CUE and litter chemistry to explicitly simulate DOC versus structural microbial products generation during litter decomposition, an important new direction for the simulation of litter decomposition dynamics in SOM models. DOC and persistent microbial products are recognized both as important contributors to SOM dynamics across soil profiles, and as shortfalls in many SOM models (Rumpel and Kögel-Knabner, 2010; Schmidt et al., 2011). From this analysis there remains uncertainty in the 'internal' dynamics of the LIDEL model, particularly pertaining to microbial biomass turnover and the fractionation between microbially-derived soluble versus structural material. However, the LIDEL model provides a robust theoretical structure to connect litter chemistry and microbial processes to the generation of these separate products (Moorhead et al., 2013; Sinsabaugh et al., 2013), as well as suggest testable hypotheses. For example, we only used data on total DOC and CO<sub>2</sub> to inform our parameter estimations. We did not inform our statistical models with data indicating the relative contribution of microbially-derived versus plant-derived DOC. Therefore while the DOC/CO<sub>2</sub> fractionation difference explored in models 1 and 3 versus 2 and 4 supported the hypothesis of higher efficiency in the generation of DOC/CO<sub>2</sub> from soluble litter pools versus holocellulose litter pools during decomposition, the parameters controlling these are an important area for further examination and refinement. Measurements of microbial products or using changes in DOC C/N ratios to determining plant litter versus microbially-derived DOC through time would address these remaining uncertainties regarding LIDEL simulations of microbial processes.

Another area for further development in the LIDEL model is the relationship between N availability and microbial biomass characteristics (e.g. fractionation between soluble *versus* structural material with microbial biomass turnover). For example, is the partitioning of microbial biomass between soluble and non-soluble products controlled by N availability? N control of microbial generation of DOC versus microbial products can be simulated by the LIDEL model but was not a

hypothesis we could explore given the data used in this analysis. However, this is potentially an important consideration for stabilization of litter inputs in soils due to N effects on microbial processes.

The hierarchical Bayesian approach used to estimate LIDEL parameters in this analysis suggests the importance of examining the relationship between measures of soluble plant litter fractions and variability in the true plant litter soluble fractions both within and across litter types. As discussed above, the soluble plant litter pool is conceptually well defined but more difficult to measure in a way that links to its ecological function and variability. Measurement of the soluble pool using HWE is clearly more variable than measurement by mass-difference (Figure 3-5, A & B). However, both types of measurements are variable relative to the estimated true state of the soluble fraction (Figure 3-5, C), making it difficult to determine whether one or the other more accurately reflects the true litter soluble fraction, or if another measure of the litter soluble fraction would be more consistent. The results of this analysis support the importance of the soluble litter fraction in both generating DOC and driving microbial dynamics during litter decomposition. Comparative measures of a variety of litters from a variety of sources, using both analysis techniques, is needed to clarify the connection between soluble litter C, measurements of soluble litter fractions, and the generation of DOC *versus* persistent microbial products during litter decomposition.

In the IC analysis we estimated an entirely unknown model parameter:  $f_{DOC}$ , or the fraction of soluble plant litter that avoids microbial processes and is immediately released from plant litter when it is exposed to water. In our IC analysis we informed the estimate of initial DOC using the cumulative DOC measured from day 0 – day 10 of the Soong et al. (2015) experiment. As shown in Table 3-2, this fraction estimate has the broadest 95% BCI width, likely in part due to a lack of any direct measurements as well as the estimation of one single initial DOC fraction across all litter types. We observed that the use of this parameter distribution to initialize the pine DOC pool from the Soong et al. (2015) experiment in the TD analysis resulted in initial DOC estimates that were higher than DOC ever cumulatively reached in the

experiment. We therefore added a 'structural protection' coefficient (0.4) for pine initial DOC, based on the assumption that needle litters have a molecular structure that allows for a greater protection of plant soluble C. This suggests the need to examine factors that physically protect soluble C from leaching, in addition to the microbial processes reflected in the generation of DOC in the LIDEL model. Litter-specific initial  $f_{DOC}$  parameters may need to be estimated to better reflect litter-specific variability in this OM fraction.

DOC showed its highest measured values early in the decomposition experiment, with the majority of leachate accumulating within the first few weeks. Simulating these high early fluxes is important from the standpoint of accurately modeling the largest litter inputs into soils, further supporting closer examination of the  $f_{DOC}$  parameter. It is also important to accurately simulate overall litter decomposition dynamics (i.e. mass remaining), as undecomposed litter can be an increasingly important contributor to SOM formation in late stages of decomposition (Cotrufo et al., in review, Nature Geosciences). The results of this analysis statistically support Model 1 as the best fit given the data (RMSE values, Table 3-3). However, litter contributions to soils would likely be more accurately simulated by a LIDEL model formulation more in the direction of Model 2, in terms of simulating the generation of the products of litter decomposition with the potential to make the greatest contributions to SOM dynamics. Therefore, further development of the LIDEL model should focus on simulating these key periods in litter decomposition, in order to more accurately reflect the connection between litter decomposition, the generation of products from litter decomposition, and SOM dynamics.

### **3.5 Conclusions**

In this analysis we present the LIDEL model, a new litter decomposition model that includes dynamic lignin and N controls on microbial CUE and simulates the partitioning of litter decomposition products to DOM, CO<sub>2</sub>, microbes, and microbial products. By simulating the production of these dissolved and particulate litter C products, the LIDEL model can more accurately connect litter

decomposition with important SOM C input pathways identified in litter decomposition experiments (e.g. Cotrufo et al., in review, Nature Geosciences), but poorly represented in many SOM models (e.g. Schmidt et al., 2011). We use hierarchical Bayesian methods to inform LIDEL model parameters with experimental and literature-derived data, both for initial LIDEL conditions as well as parameters that drive LIDEL temporal dynamics. Some parameters were informed by the data more strongly than others, due to variability in the measured data as well as the strength of the linkage between measurements and conceptually defined LIDEL model pools. The LIDEL model provides a robust theoretical structure to connect litter chemistry to microbial processes and the generation of products from litter decomposition, suggesting testable hypotheses for controls on microbial processes and their connection to the generation of products from litter decomposition.

Litter	Average initial %N	Initial $L_c$ range
Alfalfa (Medicago sativa)	4.09	0.15-0.42
Ash (Fraxinus excelsior)	0.88	0.28-0.64
Bluestem (Andropogon gerardii)	0.48	0.22-0.64
Oak (Quercus macrocarpa)	1.32	0.48-0.86
Pine ( <i>Pinus ponderosa</i> )	0.41	0.51-0.80

Table 3-1 Litter chemistry characteristics from Soong et al. (2015).

Fraction	Litter	Mean	95% BCI	95% BCl Width
DOC	All	0.15	0.09-0.21	0.12
Soluble C	Alfalfa	0.70	0.65-0.75	0.11
	Ash	0.52	0.46-0.58	0.11
	Bluestem	0.27	0.22-0.33	0.11
	Oak	0.40	0.35-0.45	0.11
	Pine	0.35	0.29-0.40	0.11
Lignin C	Alfalfa	0.08	0.05-0.12	0.07
	Ash	0.15	0.11-0.18	0.08
	Bluestem	0.13	0.09-0.16	0.07
	Oak	0.27	0.23-0.31	0.08
	Pine	0.32	0.28-0.38	0.08

Table 3-2 Estimated parameters for fractions used to initialize LIDEL model pools, reporting mean and 95% Bayesian credible intervals as well as credible interval widths.

		Model 1		Model 2		Model 3		Model 4	
N Limitation Curve:		Logistic		Logistic		Linear		Linear	
DOC/CO <sub>2</sub> Frac.: Sol vs Holo:		Equal		Soluble Higher		Equal		Soluble Higher	
Paramete r	Definition	Mean	95% BCI	Mean	95% BCI	Mean	95% BCI	Mean	95% BCI
<i>k</i> <sub>1</sub>	Soluble decay	0.24	0.13- 0.50	0.37	0.16- 0.70	0.097	0.038- 0.23	0.28	0.071- 0.93
<i>k</i> <sub>2</sub>	Holocell. decay	0.0079	0.0016- 0.017	0.0090	0.0011- 0.020	0.0034	0.0015- 0.0067	0.0049	0.0018- 0.012
$k_4$	Mic. biomass decay	0.60	0.14- 0.97	0.57	0.11- 0.97	0.55	0.071- 0.97	0.51	0.053- 0.97
$eta_3$	Mic. prod. from mic. biomass	0.27	0.022- 0.76	0.33	0.028- 0.77	0.50	0.048- 0.92	0.43	0.042- 0.88
$\lambda_2$	DOC from mic. biomass	0.16	0.018- 0.41	0.19	0.022- 0.42	0.063	0.0092- 0.23	0.099	0.011- 0.32
	RMSE	15280		16:	150	15	350	17	380

Table 3-3 Mean and 95% Bayesian credible intervals for the five estimated LIDEL parameters, as well as mean root mean square error for LIDEL model predictions versus out-of-sample measurements for each iteration of the TD analysis after the burn-in period.

LITTER



Figure 3-1 Schematic showing the soluble ( $C_1$ ), non-lignin structural ( $C_2$ ), lignin ( $C_3$ ), microbial biomass ( $C_4$ ), microbe products ( $C_5$ ), DOC ( $C_6$ ), and CO<sub>2</sub> ( $C_7$ ) pools within the LIDEL model, in addition to the fluxes that connect model pools and the parameters that control the rates of these fluxes. The three litter pools ( $C_1$ ,  $C_2$ , and  $C_3$ ) generate microbial products ( $C_5$ ) and DOC ( $C_6$ ) as separate physical pools that, alongside remaining litter, can then enter soils and soil organic matter formation pathways.



Figure 3-2 Limitation on water soluble and holocellulose decay rate (A), and microbial CUE with use of these litter fractions (B), by lignocellulose index. Water soluble litter is assumed to have a maximum CUE closer to the theoretical maximum of 0.6 proposed by Moorhead et al. (2013), while holocellulose (NLS) is proposed to have a lower maximum of 0.5. Equations modified from those presented in Moorhead et al. (2013) to be expressed as a single continuous non-linear equation rather than the combination of linear equations (Moorhead et al., 2013).



Figure 3-3 Limitation on CUE and decay of water soluble and holocellulose pools by initial litter percent N, as based on a 3% threshold element ratio drawn from (Sinsabaugh et al., 2013).



Figure 3-4 Measured versus predicted DOC generation per mass lost during decomposition, taking the minimum of the predicted fractionation based on % initial N and initial  $L_c$ , for early and late-stage decomposition.



Figure 3-5 Initial Condition Bayesian analysis results, showing trace density plots of estimated variance parameters for hot water extraction (A) and mass-difference (B) measurements of the litter soluble fractions, as well as the final estimate of the soluble fraction parameters (showing means and 95% Bayesian credible intervals) by litter type as compared to measured values (C).



Figure 3-6 Measured (x-axes), versus modeled (y-axes) LIDEL prediction of litter mass remaining (A, D, G, & J) as well as the generation of DOC and  $CO_2$  (B & C, E & F, H & I, K & L) for the four model formulations evaluated in the TD hierarchical Bayesian analysis, presenting posterior means (dark circles) and 95% BCIs for each LIDEL prediction of measured data. Diagonal lines indicate 1:1 relationships.

# 4 ASSESSING THE SOIL CARBON, BIOMASS PRODUCTION, AND NITROUS OXIDE EMISSION IMPACT OF CORN STOVER MANAGEMENT FOR BIONERGY FEEDSTOCK PRODUCTION USING DAYCENT<sup>3</sup>

### 4.1 Introduction

Corn (*Zea mays* L.) stover, the plant residue remaining after harvest, is an attractive source of biomass for bioenergy in the United States (US). The US Department of Energy Billion Ton Study Update (2011) considers agricultural residues such as corn stover to be among the best sources of raw material to support bioenergy industry, as they are already produced on a large scale at low costs in regions with established crop production infrastructure. Corn stover currently accounts for ~70% of all crop residue production in the US and is therefore of high interest as a feedstock to support the expansion of industrially-scaled bioenergy production (U.S. Department of Energy, 2011).

The impetus to expand bioenergy production in the US, exemplified at the federal level with the Energy Information and Security Act of 2007 and its revised Renewable Fuels Standard (RFSII), includes replacing the use of fossil fuels to strengthen US fuel security, reduce the climate change impact of fossil fuel combustion, and concomitantly enhance rural development (Lynch and von Lampe, 2011). However, the removal of crop residues to expand bioenergy production must consider potential impacts to the agroecosystem services provided by agricultural byproducts like corn stover. While corn stover is widely available in the corn production regions in the US, care must be taken with residue removal in order to maintain agroecosystem health and productivity (Halvorson and Jantalia, 2011; Karlen et al., 2011b). Crop residues influence multiple agroecosystem functions which impact crop productivity, including providing the building blocks for soil organic matter that in turn contribute to water and nutrient-holding capacity as well as nutrient availability (Burgess et al., 1996; Linden et al., 2000; Power

<sup>&</sup>lt;sup>3</sup> Campbell, E.E., Johnson, J.M.F., Jin, V.L., Lehman, R.M., Osborne, S.L., Varvel, G.E., Paustian, K., 2014. Assessing the Soil Carbon, Biomass Production, and Nitrous Oxide Emission Impact of Corn Stover Management for Bioenergy Feedstock Production Using DAYCENT. BioEnergy Research 7, 491–502. doi:10.1007/s12155-014-9414-z

et al., 1986). Crop residues can reduce soil erosion, help maintain soil fertility, and impact greenhouse gas emissions (GHGs), including nitrous oxide (N<sub>2</sub>O) and methane (CH<sub>4</sub>) fluxes from soil and changes in soil organic carbon (SOC) storage and net carbon dioxide (CO<sub>2</sub>) emissions (Crutzen et al., 2008; Wilhelm et al., 2004). Corn stover removal can reduce or increase grain yields depending on other land characteristics and management practices (Burgess et al., 1996; Varvel et al., 2008). Furthermore, replacing nutrients removed by stover removal can increase fertilizer costs (Karlen et al., 2011a). Sustainable agricultural management practices involving residue use must support a robust bioenergy industry, meet climate impact reduction (e.g. RFSII) standards for renewable fuels, and maintain cropland health to support growing crop production demands.

An economically viable and environmentally sustainable bioenergy industry based on corn stover feedstock must incorporate both crop management practice recommendations for corn stover production as well as accurate predictions of production potential. Past studies evaluated impacts of corn stover removal on subsequent grain yield (Burgess et al., 1996; Maskina et al., 1993; Power et al., 1986; Wilhelm et al., 1986), and used long-term sites to address other ecosystem effects such as SOC changes (Hooker et al., 2005; Moebius-Clune et al., 2008; Wilts et al., 2004). A research goal of the Sun Grant Initiative's Corn Stover Regional Partnership team- which is also supported by the USDA-Agricultural Research Service's Resilient Economic Agricultural Practices (ARS-REAP) project- was to supplement these studies by establishing an extensive network of field trials across a range of climatic and soil types to assess varying levels of residue removal on soil C, water, nutrient content, as well as biomass characteristics. More information on the Regional Partnership studies are included within this and previous publications (Karlen, 2010; Karlen et al., 2011a, 2011b).

The Regional Partnership corn stover trials greatly expand of the amount of site-specific primary data on corn stover residue production and the agroecosystem impacts of its removal. Empirical data are needed to calibrate, validate, and refine process-based models so they can be used to help establish

valid sustainable harvest rate guidelines (Johnson et al. (this issue) and (Wilhelm et al., 2004)). The need for such data was recognized by the Regional Partnership corn stover team; therefore the project was designed to support predictive modeling, linking field trial data to expanded regional projections (Karlen, 2010).

A modeling tool suited to accommodate region-specific factors and provide regional predictions for a broad array of agroecosystem impacts is DAYCENT. The DAYCENT model, a process-based ecosystem model developed at Colorado State University, simulates GHG fluxes as well as plant/soil C dynamics and many other ecosystem processes (Parton, 1988, 1987). DAYCENT and its predecessor CENTURY have been applied and tested in many agricultural systems both in the US and globally (Abdalla et al., 2010; Del Grosso et al., 2008; M. V. Galdos et al., 2009), including several types of bioenergy production systems (Adler et al., 2007; Davis et al., 2012, 2010). While DAYCENT modeling of corn production has been widely tested for model applications such as estimating agricultural land use emissions for the US Environmental Protection Agency annual GHG inventory report (U.S. Environmental Protection Agency, 2013), evaluations of its performance simulating the impacts of corn stover harvest are limited. One example is a recent study by Gao et al. (2013), where DAYCENT was applied as part of a Michigan-specific life-cycle assessment (LCA) of corn stover management (Gao et al., 2013).

There is a critical need for the application of process-based models such as DAYCENT in designing scientifically sound decision support tools for the development of bioenergy feedstock production (Huggins et al., 2011). Therefore, members of the Regional Partnership team completed a review to identify economic and sustainability metrics that impact the potential for corn stover residue harvest in the Midwest (Wilhelm et al., 2010). Based on this review (Wilhelm et al., 2010), an integrated stover removal tool was designed that linked several existing models (i.e. the Revised Universal Soil Loss Equation Version 2 (RUSLE2), the Wind Erosion Prediction System (WEPS) and the Soil Conditioning Index (SCI)) to predict maximum residue removal rates that would meet multiple sustainability criteria

(Muth and Bryden, 2013). Recently, the DAYCENT model was integrated into this framework to provide dynamic estimates of biomass yields, SOC changes and GHG fluxes, thereby refining the evaluation and recommendation of 'sustainable' residue removal rates on regional and site-specific levels. Evaluating DAYCENT simulations of corn stover residue removal effects against empirical data from the Regional Partnership field sites contributes to the development of the sustainability assessment tool that can provide region-specific recommendations to support agroecosystem services and the bioenergy industry.

Balancing agroecosystem services and developing a viable large-scale bioenergy industry will require changing management practices (e.g., growing cover crops, modifying fertilizer form or rate) to increase or supplement C and nutrients in the system. Cover crops and manure application (Fronning et al., 2008; Wiggans et al., 2012), increased synthetic N fertilizer application (Sindelar et al., 2013, 2012), and reduction in tillage intensity (Hooker et al., 2005; Linden et al., 2000) have been identified as amelioration approaches to develop 'sustainable' practices for corn residue harvest (Gollany et al., 2010; Wilhelm et al., 2010) and subsequently are recurring focal treatments in field studies assessing corn stover management. In this study, we evaluated DAYCENT performance in simulating SOC change, corn grain and stover yields, and direct  $N_2O$  emissions- a powerful greenhouse gas (Forster, 2007)- from soils against measured data from three of the Regional Partnership corn stover sites and two long term published corn stover removal experiments (Clapp et al., 2000; Follett et al., 2012; Hammerbeck et al., 2012; Linden et al., 2000; Reicosky et al., 2002; Stetson et al., 2012; Varvel et al., 2008; Wilts et al., 2004). For the current study, simulated treatments were based on site-specific variation in residue removal rates, tillage, N fertilizer and cover crops. We focus our analyses on overall DAYCENT performance in these systems, as well as on the measured versus modeled impacts of residue removal rates combined with variation in tillage treatments.

#### 4.2 Materials and Methods

#### 4.2.1 Experimental Data

To test the performance of the DAYCENT model, a series of data were assembled from published literature evaluating two sites in Rosemount, MN and Morris, MN for the soil and crop production impacts of long-term corn stover removal, as well as different levels of nitrogen (N) fertilizer application and types of tillage (Clapp et al., 2000; Linden et al., 2000; Reicosky et al., 2002; Wilts et al., 2004). Data were also assembled from three Corn Stover Regional Partnership sites in Ithaca, NE, Brookings, SD and a different site in Morris, MN established as a subset of ARS-REAP to evaluate the sustainability of corn stover harvest (Karlen, 2010). The Regional Partnership sites test multiple levels of residue removal combined with differences in tillage, N fertilizer application, and cover crop management practices and are described in greater detail in this issue as well as in prior publications (Hammerbeck et al., 2012; Stetson et al., 2012; Varvel et al., 2008). Measurements at the Regional Partnership sites included grain and stover yields, SOC change from 0 – 20 cm, and direct soil N<sub>2</sub>O emissions (Table 4-1). The Regional Partnership data and published literature values allowed us to test DAYCENT's performance in simulating biomass (i.e. grain and stover) production, SOC change, and N<sub>2</sub>O

### 4.2.2 DAYCENT Model Overview

The DAYCENT model runs on a daily time-step and simulates various ecosystem processes to a soil depth of 20 cm. The model includes routines for simulating the movement of soil nutrients, the movement of water through soil layers, plant growth, and many other ecosystem components that are described in greater detail elsewhere (Del Grosso et al., 2008). The key drivers of DAYCENT include maximum and minimum daily temperature, daily precipitation, soil texture, and land management (including specific plant types grown and soil management such as tillage and nutrient additions).

The DDcentEVI version of DAYCENT (a version of DAYCENT with the option to use Enhanced Vegetation Index- i.e. EVI- data) was used for this analysis. DDcentEVI was developed and tested to estimate total agricultural land use emissions for the US Environmental Protection Agency's (EPA) GHG emission inventory annual report (U.S. Environmental Protection Agency, 2013). This repeated annual set of simulations involve model parameterization for common agricultural crops- including corn and soybean- as well as millions of model runs and established protocols for estimating model uncertainty (Del Grosso et al., 2010; Ogle et al., 2010, 2007), documented in detail in Annex 3.12 of the most recent inventory report (U.S. Environmental Protection Agency, 2013). Given the extensive parameterization process with DDcentEVI, we chose to use this version in order to focus our efforts on validating model performance using all data for corn stover residue removal experiments available for this analysis.

An important component of any DAYCENT model simulation is initializing the model based on the native ecosystem type expected for the specific site and using the best available information about land management after the native ecosystem is converted for agricultural use. Given temperature and precipitation as key drivers of biogeochemical processes, the longest possible continuous daily climate datasets are needed to run model initializations as well as drive simulations for the experimental periods of interest. The climate data used to drive model simulations for this study were derived using the latitude and longitude of site locations to determine the nearest North American Regional Reanalysis (NARR) grid cell and the associated daily maximum and minimum temperatures and total precipitation from 1979 – 2009. NARR data, given at a 32 km scale, were generated as an extension of the National Centers for Atmospheric Research Global Reanalysis project and are freely available online (http://www.emc.ncep.noaa.gov/mmb/rreanl/). NARR data are generated using algorithms to interpolate weather for areas between weather stations. The NARR dataset is the standard used for simulations in the US GHG inventory, and therefore was the source of climate data for all sites and years in this analysis where site-specific data were not available. For the three Regional Partnership sites, site-
specific daily weather data were available for the experimental time periods, and these data were used instead of NARR weather data in the years available, extending the climate data time period at these sites to 2010.

Soil texture information was gathered either from direct field measurements reported by the Regional Partnership sites or from soil texture data reported in publications for the two non-Regional Partnership sites. Prior land use history used for model initialization were drawn from 1) county-level native vegetation assumptions used in simulations for the EPA GHG emission annual report and 2) information gained from literature and personal communication for agricultural management from when native vegetation was converted into cropland up to the treatment period. Experimental management practices, such as planting and harvest dates, dates and quantities of fertilizer application, and corn stover harvest rates, were drawn from reported literature and Regional Partnership field data. These land use data were used to schedule events within DAYCENT model simulations. A total of 53 different corn stover management scenarios, matching experimental management practices across the five experimental sites, were simulated to generate model results to compare against measured data.

### 4.2.3 Statistical Analyses

Statistical analyses were completed using R-2.15.1 software as well as the Hmisc and car packages (Fox and Weisberg, 2011; Harrell Jr., 2012; R Core Team, 2012). Regression analyses were applied to compare measured versus modeled grain yield (Mg C ha<sup>-1</sup>), stover yield (Mg C ha<sup>-1</sup>), SOC change over the measurement period (Mg C ha<sup>-1</sup>), and annual N<sub>2</sub>O flux (kg N<sub>2</sub>O-N ha<sup>-1</sup> yr<sup>-1</sup>) across all sites and all years. Two treatment effects were also selected for measured versus modeled estimate evaluation: (1) residue removal level, and (2) residue removal level + tillage. For measurements taken across multiple years, averages by treatment were used to compare measured versus modeled results across all sites. For evaluation of residue removal level alone, measured versus modeled estimates of grain C and SOC change were compared for three stover removal levels: full removal (100 %), moderate

removal (29 – 50%), and no removal (0%). Soil N<sub>2</sub>O emissions were evaluated using two treatment levels: stover removal (>0%) and no removal (0%). For evaluation of residue removal level + tillage, measured versus modeled estimates of grain C and SOC change were compared using four levels: conventional tillage + 0% removal, conventional tillage + >0% removal, no tillage + 0% removal, no tillage + >0% removal. Measured versus modeled estimates were assessed for normality using a Shapiro-Wilk test, as well as assessed for equal variance. These measured versus modeled estimates were then assessed across these treatment effects using type III sums of squares two-way analyses of variance (ANOVAs) to account for unequal sample sizes, where the fixed effects tested were residue removal level and tillage.

## 4.3 Results

## 4.3.1 Biomass

DAYCENT simulated annual grain yields with significant correlation with measured yields but with high dispersion and significant bias – there was a significant positive intercept and a slope less than 1 (Figure 4-1a). DAYCENT simulated annual stover harvest with a tighter significant correlation and less bias, with a slope closer to 1 and an insignificant intercept (Figure 4-1b).

Measured and modeled average grain yields across sites did not differ between residue removal levels (Figure 4-2a, p>0.05). Across all treatment levels, measured values were significantly higher than modeled estimates (Figure 4-2a, p=0.012). Measured and model estimates of average grain yields did not show a significant overall effect of tillage + residue removal levels, measured versus modeled estimates, or a significant interaction between these factors (Figure 4-2b, p>0.05).

## 4.3.2 Soil C Change

Modeled and measured SOC change for all sites exhibited a significant correlation but deviation from the 1:1 linear relationship, with a significant positive intercept and slope less than 1 (Figure 4-3a). Excluding sites that had ten or fewer years between initial and final SOC measurements (Figure 4-3b),

and thus greater uncertainty in the magnitude and direction of change, gave a tighter correlation between measured and modeled estimates. However, removing short-term sites did not change the model overestimation of SOC loss rates for sites where measured SOC stocks were declining, with a less significant positive intercept and a lower slope (Figure 4-3b).

For treatments where the time between initial and final SOC measurements exceeded 10 years, SOC change did not vary significantly by residue removal rate or by the interaction between residue removal rate and the measured versus modeled estimates (Figure 4-4a, p>0.05). However measured versus modeled estimates did differ significantly, with the model consistently overestimating SOC change in the same direction as the measured data (Figure 4-4a, p<0.001). For these same treatments, SOC did vary significantly by tillage + residue removal level and measured versus modeled estimates alone, as well as by the interaction between these factors (Figure 4-4b, p=0.003, p=0.003, and p=0.006, respectively). Both the measured and modeled data suggest conventional tillage leads to a loss of SOC, while no tillage leads to a gain or little change in SOC that may depend on whether residue is removed. However the model shows a clear bias of overestimating SOC loss with conventional tillage, with or without residue removal.

### 4.3.3 N<sub>2</sub>O emission

Modeled versus measured annual N<sub>2</sub>O emissions showed a significant relationship and an intercept that did not differ significantly from 0, but the slope > 1 indicates a model bias of underestimating annual N<sub>2</sub>O emissions. The increasing divergence from the 1:1 line between modeled and measured values with higher measured emission rates suggests better DAYCENT performance in low-emission systems (<2 kg N<sub>2</sub>O-N ha<sup>-1</sup> yr<sup>-1</sup>) but underestimation for sites and years with high emission rates (i.e., > 3 kg N<sub>2</sub>O-N ha<sup>-1</sup> yr<sup>-1</sup>) (Figure 4-5a). There were no significant differences in N<sub>2</sub>O as a function of residue removal in either the measured or modeled estimates (Figure 4-5b, p>0.05 for all factors).

A qualitative comparison of measured versus modeled estimates of daily N<sub>2</sub>O emission shows no obvious pattern of divergence between daily modeled and measured values when N<sub>2</sub>O emissions are low (Figure 4-6 & 7). However there is substantial divergence when N<sub>2</sub>O emissions are in peak periods. DAYCENT sometimes simulated peak periods with similar timing to measured peak periods but underestimated their magnitude (Figure 4-6b, Figure 4-7c). DAYCENT simulation of peaks sometimes did not match the timing of measured peaks (Figure 4-6d, Figure 4-7b). DAYCENT also failed to simulate some peak periods reflected in the measured data (Figure 4-6b, Figure 4-7a & 7b).

## 4.4 Discussion

## 4.4.1 Biomass

Modeled results suggest that DAYCENT can reasonably simulate corn stover yields across the management practices considered in this study, a key concern for the expansion of bioenergy industry based on corn residue as a feedstock material. However the slope did exceed 1- the value that would reflect perfect model simulation of measured values- indicating some bias towards DAYCENT underestimating stover yields that should be considered in the use of DAYCENT model results within the context of a bioenergy decision support tool.

DAYCENT performed more poorly in simulating annual grain yields. The low coefficient of determination ( $r^2$ ) reflected high dispersion, and the model tended to overestimate grain yields in years where measured grain yields were low and underestimate grain yields in years where measured grain yields were high (Figure 4-1a). DAYCENT simulates the growth of aboveground biomass based on interactions between moisture and temperature, and then simulates grain harvest based on the harvest index specified by the user. Fine-scaled timing-specific interactions between temperature, moisture, and grain yields (e.g. high or low precipitation or temperature events that impact flowering or grain filling) are not yet included in DAYCENT, and may cause its variable performance simulating annual grain yields. When the data were aggregated across years, DAYCENT did successfully simulate the overall non-

significant impact of corn stover residue removal levels across these treatments; however, the significant difference between modeled versus measured results indicates an overall model bias of underestimating grain yields (Figure 4-2a). This bias was not apparent in the comparison between measured versus modeled grain yields across combined tillage and residue removal levels; in this analysis, the model successfully captured the overall insignificant impact of residue removal levels and tillage on grain yields, with no significant difference between measured and modeled results (Figure 4-2b).

These results suggest DAYCENT can be used to successfully model the relative impacts of residue removal on grain yields, but should be used carefully if simulating quantities of grain yield on an annual basis. The DDcentEVI version of DAYCENT has shown a tendency to overestimate water stress effects on grain production in the northern Midwest region (Steve Williams, personal communication). This region was the location of several sites used in this study that showed substantial model underestimation of annual grain yields relative to measured values (open circles, Figure 4-1a). This model bias should be addressed for evaluating sustainability factors for corn stover residue harvest in the northern Midwest region, with a better representation of water stress effects on grain yields- particularly better representation of the effects of precipitation timing, which can be critical for grain production- to alleviate this bias in modeled results.

## 4.4.2 Soil C change

The DAYCENT model had variable performance modeling SOC change. DAYCENT simulations matched the direction of SOC change for most sites (Figure 4-3a), particularly sites with greater periods of time between initial and final SOC measurements used to calculate SOC change (Figure 4-3b). DAYCENT results, however, showed increasing divergence from measured values showing SOC losses (Figure 4-3). Soil C is highly heterogeneous and variable, and often longer periods of time are required to effectively measure and observe SOC changes due to changes in land management. The sites in this

study with shorter time sets had greater measured variation than was reflected in the model estimates, which tended to model treatments at these sites as just above or below 0 change (Figure 4-3a). It would be useful to repeat measurement of SOC at these sites in the future to assess the accuracy of model simulations over a longer time period. The sites might experience more SOC change following near-term management changes than the model is predicting, but the model might account for some of this SOC change if it simulates the effects of the practice (such as increased decomposition) for longer than the site experiences. Alternatively the model may be underestimating total changes occurring with the management practices at these sites. The modeled versus measured comparison of SOC changes over a longer time period (e.g., 10 years or greater, as were available for the other sites in this analysis) would address this question, with either increasing correlation between measured versus modeled results supporting the former behavior of underestimating near-term changes, or continued disconnect between modeled and measured SOC change indicating poor model performance in these locations. In the latter case, further model assessment of SOC simulations would be required to identify the cause for these site-specific discrepancies in modeled simulations as compared to measured data.

Measured SOC changes by residue removal show no significant SOC change with different levels of residue removal, but suggest greater SOC loss with full residue removal and the potential for SOC gain with partial removal (Figure 4-4a). This supports a number of studies suggesting that SOC change can be minimized by partial rather than full residue removal (Blanco-Canqui and Lal, 2007; Johnson et al., 2010; Varvel et al., 2008; Wilhelm et al., 1986). In contrast, modeled results show the unexpected behavior of greatest SOC gain with moderate residue removal, and potential for loss with both no and full residue removal (Figure 4-4a). However these results could be due to interacting effects with the overestimated modeled SOC loss with conventional tillage, which can be observed in Figure 4-4b.

DAYCENT captures the measured trend of potential for SOC loss with conventional tillage across residue removal treatments versus some SOC gains with residue treatments combined with no tillage

(Figure 4-4b). However the model is clearly overestimating the SOC losses in the conventional tillage treatments (Figure 4-4b). Simulated soil decomposition processes in the DDcentEVI version of DAYCENT may be too sensitive to tillage. Comparison of modeled estimates to total aboveground and belowground biomass, as well as an analysis of modeled soil decomposition sensitivity to tillage, would clarify the model processes causing this overestimation of SOC loss. Despite oversensitivity to tillage, our results support other modeled analyses that identified the potential for no tillage management practices to maintain SOC with residue removal (Dalzell et al., 2013). Specifically, our measured and modeled SOC changes indicated a minimal or positive SOC change when no tillage is combined with residue removal, versus a more strongly negative SOC change when conventional tillage is combined with residue removal (Figure 4-4b). Numerous studies have focused on combining management practices such as increased N fertilizer and reduced tillage with corn stover residue harvest, in order to maintain yields and soil fertility (Burgess et al., 1996; Hooker et al., 2005; Sindelar et al., 2013, 2012). In the set of sites analyzed in this study, we were only able to analyze tillage in addition to residue removal for modeled versus measured analyses. While N fertilizer levels were included in some of the experiments considered in this analysis, aggregating data for cross-site comparison between N fertilizer levels was not possible due to differences in experimental designs. The importance of combined tillage plus residue removal level in this analysis suggests that other combined management practices such as N fertilizer across residue removal levels should be the next focal area of cross-region model validation and assessment, in order to better support the simulation of these management practices within a bioenergy decision support tool framework.

## 4.4.3 N<sub>2</sub>O emission

The DDcentEVI version of DAYCENT performed more accurately modeling lower annual measured estimates of N<sub>2</sub>O flux, as indicated by an insignificant intercept in the regression analysis. However the slope indicates increasing underestimation of simulated annual flux as measured values

increased, which is particularly apparent for sites and years >3 kg N<sub>2</sub>O-N ha<sup>-1</sup> yr<sup>-1</sup> (Figure 4-5a). Due to how data were reported, error measurements were only available for some of the treatments and were included to give some visualization of the variability of measured annual N<sub>2</sub>O estimates (Figure 4-5a). Including error measurements on all measured annual N<sub>2</sub>O emission estimates would help clarify the comparison between modeled and measured values. The model showed an overall bias of underestimating annual N<sub>2</sub>O flux, when results were averaged and compared across residue removal levels (Figure 4-5b).

N<sub>2</sub>O emissions are highly variable and transient, with high flux occurring often in short timeframes following certain events, such as high precipitation, spring-thaw, or N fertilizer application. A comparison of daily measured versus modeled N<sub>2</sub>O flux shows that the divergence between measured and modeled annual flux may be due in part to DAYCENT's failure to capture the presence, timing, or magnitude of transient peak periods (Figures 4-6 & 4-7). For example, in the representative treatment shown in Figure 4-6, the model seems to do well in years when soybean is grown and N<sub>2</sub>O emissions are consistently low (Figure 4-6, a & c), perhaps missing a small peak around the harvest period. In comparison, when measured values reflect larger and more frequent peaks during corn growth years the model performs more variably. In 2009 (Figure 4-6b) the model simulations seems to match the timing but not the magnitude of the highest N<sub>2</sub>O peak and then missed a peak period at the end of the season, while in 2011 (Figure 4-6d) the model simulations seemed to match the magnitude but not the timing of the highest N<sub>2</sub>O peak.

Despite evidence for variable DAYCENT performance simulating transient peak periods of  $N_2O$ flux, it is challenging to validate DAYCENT performance either cumulatively or on a daily basis using the discontinuous and sometimes sparse time series of  $N_2O$  emission measurements made available in this study. These types of  $N_2O$  emission datasets are common, as more frequent measurements taken by hand or by automatic chambers are resource-intensive. Where continuous  $N_2O$  emission measurements are not possible, discontinuous samples taken at time points aimed to capture transient periods of high flux as well as background flux across the season is a common methodological approach. However sampling frequency has been recognized to affect cumulative estimates of N<sub>2</sub>O emissions, with increasing divergence between true and estimated N<sub>2</sub>O emissions as sampling intervals increase in length (Parkin, 2008).

In the comparison between DAYCENT modeled estimated and measured values, when N<sub>2</sub>O flux changes occur at a time resolution finer than the measured data, the accuracy of DAYCENT simulation between measured data points will remain unclear. This can be observed in the results of this study, with specific examples include the peaks and lows simulated for N<sub>2</sub>O emissions between consecutive high measured data points in Figures 4-6d (measured data on either side of day 200), 4-7a (the first two measured data points), and 4-7b (measured data points on either side of the highest simulated peak). As another example, while the highest measured peak in Figure 4-6d is comparable to the modeled peak in magnitude, it is unknown whether the model is simulating the peak too early, or whether the measurement was taken as the flux was coming down from a higher peak (Figure 4-6).

There is additional potential for sparse, discontinuous N<sub>2</sub>O measurements to overestimate annual flux, depending on the resolution and timing of the measured data and the method of integration used to generate an annual estimate. For example, measured time points might miss a peak or a period of low emissions between peaks, or might miss the timing with which a transient peak returns to a baseline flux. The first two data points in Figure 4-7a demonstrate this potential; if the period of low emissions simulated by DAYCENT did occur at that site, but is not considered in the integration of the two measured data points to estimate total flux for that period, their integration will result in an overestimation of N<sub>2</sub>O flux. It is possible that DAYCENT's underestimation of high N<sub>2</sub>O flux may be due to how the measurement data were integrated to determine cumulative emissions. However it should be noted that a comparison of continuous versus discontinuous N<sub>2</sub>O emission

measurements demonstrated a pattern of *underestimation* of cumulative emissions using discontinuous data, due to failure to capture transient peaks in the time interval between measurements (Fassbinder et al., 2013). In this latter case DAYCENT simulations of N<sub>2</sub>O emissions would diverge even further from measured estimates in years with high flux. Without data at a higher temporal resolution, it is not possible to determine either the accuracy of the annual measured estimates or the extent to which DAYCENT simulations diverge from true N<sub>2</sub>O emissions.

Due the transience and the magnitude of N<sub>2</sub>O flux changes across the growing season, there would be great benefits in comparing continuously measured N<sub>2</sub>O data against DAYCENT model results, in order to inform the magnitude and timing of peak flux events in model simulations as well as more accurately compare annual flux to measured values. For the purposes of using DAYCENT to evaluate the N<sub>2</sub>O emissions of different production practices as a sustainability metric, care should be taken not to underestimate N<sub>2</sub>O emissions in systems of potentially high flux.

## 4.5 Conclusion

Overall DAYCENT had variable performance simulating the impacts of treatments for corn stover harvest included in the 5 sites used in this analysis, with the greatest accuracy simulating corn stover yields and consistency in capturing management practice impacts on the relationship and direction of change with SOC and corn grain biomass. DAYCENT had variable performance simulating N<sub>2</sub>O emissions, with more accurate performance where annual emissions are low. Cumulatively the model concurred with measured results suggesting little overall grain yield impacts and suggested the potential for negative SOC impacts with corn stover residue removal and conventional tillage. The model has a tendency to underestimate grain yields- particularly in some regions where the model might be overestimating the impacts of water stress- as well as overestimate SOC loss with conventional tillage, and underestimate treatments with high N<sub>2</sub>O emissions compared to modeled data. These behaviors are important to consider when integrating DAYCENT results into a larger sustainability estimate, where

theses tendencies could respectively lead to underestimation of corn grain production potential, overestimation of the negative soil C impacts of residue removal and tillage, and underestimation of the emission of  $N_2O$ .

Our residue removal and tillage results support the concept of pairing changes in management for corn stover harvest with treatments such as conversion to no-till to maintain productivity and soil health. Modeled results suggested the potential for interactive effects between residue removal and tillage. We also suggest that other combined management practices such as fertilizer application and cover crops be included in subsequent analyses of measured and modeled data comparisons, as these are key practices being considered and recommended as large-scale corn stover harvest for bioenergy moves forward.

This study reflects one of the original purposes of the Regional Partnership corn stove project; integrating field data with predictive modeling of corn stover removal management practices on a regional basis, in order to support the recommendation of sustainable practices to advance a robust bioenergy industry based on corn stover as a feedstock material.

Source	Location	Lat/ Long	Management Description	Grain yield	Stover yield	Soil C	$N_2O$
(Reicosky et al., 2002; Wilts et al., 2004)	Morris, MN	45.6/ -95.9	29 years continuous corn; moldboard tillage; low fert. (83 kg N/ha), high fert. (166 kg N/ha), & control (0 kg N/ha); 0 & 100% stover removal	x	Х	Х	-
(Clapp et al., 2000; Linden et al., 2000)	Rosemount, MN	44.7/ -93.1	13 years continuous corn; chisel, moldboard & no till tillage; 0 & 200 kg N/ha; 0 & 100% stover removal	x	х	x	-
Reg. Partnership: Swan Lake Exper. Site	Morris, MN	45.7/ -95.8	7 years corn/soy rotation; chisel & no till tillage; 130 kg N/ha; 0, 50, 100% stover removed	х	х	Х	x
Reg. Partnership: U of N Ag. Res & Develop. Center(Follett et al., 2012; Varvel et al., 2008)	Ithaca, NE	41.2/ -96.4	13 years continuous corn; no till tillage; 60, 120, & 180 kg N/ha; 0 & 100% stover removed	x	Х	x	x
Reg. Partnership: N. Cent. Ag. Res Lab.(Hammerbeck et al., 2012; Stetson et al., 2012)	Brookings, SD	44.3/ -96.8	7 years corn/soy rotation; no till tillage; avg. 135 kg N/ha; 0, ~29% & ~97% residue removal, with and without cover crop	х	х	х	х

Table 4-1 Summary of study locations and treatments.



Figure 4-1 Measured versus modeled grain yields (*A*) and corn stover harvest (*B*) for all treatments and all years for which daily weather data was available. The grain comparison (A) shows a significant relationship and intercept (adjusted  $r^2$ =0.13, slope=0.47, intercept=1.43, p<<0.001; RMSE=0.95). The stover comparison (B) shows a significant relationship, but insignificant intercept (adjusted  $r^2$ =0.53, slope = 1.18, p<<0.001; intercept=0.36, p=0.11; RMSE=0.90).



Figure 4-2 Modeled versus measured grain yields by three levels of residue removal rates (100% = full, 100% > moderate > 0%, and 0% = none) (*A*), as well as four levels of combined tillage (conventional versus no tillage) and residue removal (0% = none versus > 0% = rem) (*B*). Error bars show standard error, with the number of replicates reported above each set of measured versus modeled comparison.



Figure 4-3 Measured versus modeled SOC change from 0 – 20cm for all sites and all treatments (*A*) and for sites and treatments with greater than 10 years between initial and final SOC measurements (*B*). All sites (A) show a significant relationship and intercept (adjusted  $r^2$ =0.54, slope=0.43, p<<0.001; intercept=1.97, p=0.002; RMSE=4.03). Sites with longer SOC measurements (B) show a significant relationship and slope, but a weakly significant intercept (adjusted  $r^2$ =0.67, slope=0.48, p<<0.001; intercept=2.7, p=0.03; RMSE=3.89).



Figure 4-4 Modeled versus measured SOC change at sites with > 10 years between initial and final SOC measurements from 0 - 20cm by three levels of residue removal rates (100% = full, 100%>moderate>0\%, and 0%=none) (A), as well as four levels of combined tillage (conventional versus no tillage) and residue removal (0%=none versus >0%=rem) (B), showing standard error.



Figure 4-5 Measured versus modeled annual N<sub>2</sub>O flux for two sites (A) showing measured standard error where reporting made these data available, a significant relationship, and an insignificant intercept (adjusted  $r^2$ =0.22, slope=1.67, p=0.019; intercept=0.078; RMSE=1.2). Bar graph (B) shows measured versus modeled mean annual N<sub>2</sub>O flux by residue removal level (none=0% stover harvest (n=3), removal= >0% stover harvest (n=4)), with standard error.



Figure 4-6 Daily  $N_2O$  flux measurements (hatched diamonds) compared to daily model simulations (line) of  $N_2O$  flux for one representative site under four years of corn - soybean rotation, with no residue removal in soybean years (A and C) partial residue removal during corn years (B and D).



Figure 4-7 Daily measured (hatched diamonds) versus modeled (line)  $N_2O$  flux for a selection of sites and years where annual emissions calculated from measured values exceeded simulated annual emissions by at least 1.5 kg  $N_2O$ -N ha<sup>-1</sup> year<sup>-1</sup> (see Figure 4-5a), showing examples of model failure to simulate  $N_2O$  peaks (A & B), failure to simulate the timing of  $N_2O$  peaks (B), and underestimation of  $N_2O$  peak magnitude (C).

# 5 EVALUATING DAYCENT MODEL PERFORMANCE ESTIMATING SOIL CARBON AND NITROUS OXIDE EMISSIONS IN SOUTH CENTRAL BRAZILIAN SUGARCANE PRODUCTION SYSTEMS<sup>4</sup>

## 5.1 Introduction

In the developing global bioenergy industry, Brazilian ethanol production from sugarcane (Saccharum officinarum L.) holds an important position due to its history as a dominant nationally-scaled biofuel, combined with its recognized potential to serve growing global demands for bioethanol (Goldemberg et al., 2014). Brazil has been a lead producer of fuel ethanol from sugarcane since the 1970's, with policies that supported large-scale national development and use (Solomon et al., 2007). In the most recent decade international export of Brazilian sugarcane has increased. This increase is driven in part by the high productivity of sugarcane combined with the relative ease with which it can be converted into ethanol, as well as in part by Brazilian efforts to export national expertise in ethanol production (Solomon et al., 2007; Wells and Faro, 2011). Growing international demand has consequently supported the expansion of land area devoted to sugarcane production in Brazil. This expansion is reflected in the construction of increasingly large sugarcane processing facilities. As of 2010, for example, 30 existing facilities could process over 3 Mt sugarcane per year while developing facilities were anticipated to process 3 - 4 Mt sugarcane per year, each of the latter requiring ~56,000 ha of land (Seabra et al., 2010). Between the years 2000 – 2012 land area for sugarcane production increased by nearly 5 Mha in Brazil, with half devoted to bioenergy production (Goldemberg et al., 2014; Mello et al., 2014).

The rapid, large-scale, and potentially continued expansion of sugarcane production in Brazil elicits several concerns for the sustainability of sugarcane-based bioenergy. Generally, sustainable cropbased biofuel development is expected to meet criteria for positive or minimal impact on soil fertility

<sup>&</sup>lt;sup>4</sup> In preparation for submission to BioEnergy Research with A. Silva, F.F.C Mello, C.E.P. Cerri, M.J. Easter, C. Davies, C.C. Cerri and K. Paustian.

and function, food production, soil C storage, soil greenhouse gas (GHG) flux, and ecosystem services (Reijnders, 2006). In developing global legislation, however, there is an emphasis on supporting biofuels with minimal greenhouse gas impacts ((Congressional Research Service, 2007), see also (Goldemberg et al., 2014)). Greenhouse gases accumulating in the atmosphere—largely from human sources—are warming the globe, causing impacts on human societies and natural systems which will continue to expand and be sustained for centuries (Collins et al., 2013; Forster, 2007). Fossil fuel combustion is a dominant source of these emissions, releasing carbon into the atmosphere that otherwise would have remained stored for geologically long periods of time. Thus bioenergy has the potential to reduce the GHG emissions by substituting for fossil fuels, but this requires understanding and effectively managing the fluxes from which greenhouse gases can be derived in the process of growing, processing, and using crop-based biofuels for energy (Cherubini, 2010). While greenhouse life cycle assessment methods address many aspects of biofuel production chains with increasing sophistication (Cherubini, 2010; Lee, 2004), the heterogeneity of ecosystem processes involved in determining the soil C and direct GHG fluxes impacts of bioenergy crop production remain a challenge.

In Brazilian sugarcane production there has been a recent focus on evaluating direct soil C impacts of land use conversion into sugarcane production (Mello et al., 2014). Land use change has been long recognized as an area for agriculture to act as a sink versus source of CO<sub>2</sub> (Paustian et al., 1998) and in bioenergy production is additionally recognized as holding long-term implications for the life cycle GHG emissions associated with crop-based bioenergy production (Searchinger et al., 2008). In Brazil there is particular concern with expanded sugarcane production pushing soybean production and cattle grazing, accompanied with deforestation, closer to the Amazon region (Wells and Faro, 2011), although legislation has been established to prevent the conversion of native lands into crop production (Goldemberg et al., 2014). Rather, there is greater focus on changing how existing agricultural and pasture lands are used, to accommodate the expansion of sustainable sugarcane production

(Goldemberg et al., 2014; Mello et al., 2014). These developments combine with a widespread shift from the tradition of burned sugarcane harvest to unburned and mechanized harvest—driven by legislation in the late-1990's and early 2000's—to make soil C a focal component of both experimental and modeling work in Brazilian sugarcane production systems (Galdos et al., 2009; Galdos et al., 2009a; Hassuani et al., 2005).

A missing component coming now under close scrutiny is the impact of sugarcane management practices on GHG emissions, with specific focus on changes in nitrous oxide  $(N_2O)$  emissions from burned versus mechanical harvest (de Oliveira Bordonal et al., 2012; Galdos et al., 2009), application of organic ethanol production byproducts such as vinasse and filtercake (Carmo et al., 2013; de Oliveira et al., 2013; Paredes et al., 2014), different levels of residue return following harvest (Carmo et al., 2013), and application of different forms of synthetic fertilizer (Signor et al., 2013). Other studies have used lower tier Intergovernmental Panel on Climate Change (IPCC) methods to determine the N<sub>2</sub>O emissions from different Brazilian sugarcane production practices (De Figueiredo and La Scala Jr., 2011). However, direct measurements indicate a high level of variability in the emission of N<sub>2</sub>O from nitrogen inputs during sugarcane production, relative to IPCC estimates (Carmo et al., 2013; de Oliveira et al., 2013; Signor et al., 2013). Process-based ecosystem models can integrate multiple land use changes and crop management practices, across varying soil and climate conditions, in order to estimate the soil C and GHG emission impacts of different types of bioenergy production systems (Campbell et al., 2014; Galdos et al., 2009; Muth and Bryden, 2013; Zhang et al., 2010). In order to evaluate the climate impact of sugarcane production practices at larger scales, there is a need to use such process-based models to integrate the interaction between land use change, sugarcane production practices, and spatially and temporally varying biophysical factors that drive the emission of GHGs from soils.

The DAYCENT ecosystem model has been used to simulate soil C and GHG emission impacts of bioenergy cropping systems (Campbell et al., 2014; Duval et al., 2013; Gao et al., 2013), while its

predecessor CENTURY has been used to successfully evaluate some of the different Brazilian sugarcane management practices listed above (Galdos et al., 2009). The DAYCENT model was used to evaluate the GHG impacts of land use conversion from pasture into energy cane in Florida, USA (Duval et al., 2013). In this analysis we use data from a collection of studies in South Central Brazil to parameterize and validate DAYCENT model performance simulating sugarcane production and soil C. One of the challenges in Brazil is the relative paucity of high quality weather stations and meteorological data over large areas of the country, since weather variables (temperature and precipitation) are important drivers for soil N<sub>2</sub>O fluxes. Thus we evaluated DAYCENT estimation of N<sub>2</sub>O emissions and compare modeled versus measured estimates of N<sub>2</sub>O emission factors using two different sources of climate data: 1) nearest meteorological station data and 2) interpolated weather data from the Climate Forecast System Reanalysis (CFSR) global meteorological dataset (Fuka et al., 2014). DAYCENT simulations are highly limited by the quality, availability, and continuity of available weather data. A standardized resource like CFSR can be invaluable to drive DAYCENT simulations over larger areas. The comparison of weather data estimations in this analysis aims to contribute towards the use of DAYCENT for larger-scale regional and national analyses of GHG emissions with sugarcane production practices in Brazil.

### 5.2 Materials and Methods

#### 5.2.1 Experimental Data

To parameterize the DAYCENT model, data were collected from a selection of published studies grouped in the South Central region of Brazil (Figure 5-1). Most studies measured a single type of data, specifically soil C, N<sub>2</sub>O emissions, or aboveground production. Therefore the parameterization studies were selected to represent data gathered from experiments that included the main management practices of interest from the perspective of N<sub>2</sub>O emissions, including mineral fertilizer application, vinasse application, residue return, and burned versus unburned harvest (Table 5-1). Parameterizing DAYCENT to accommodate this range of management practices, across yields, soil C, and N<sub>2</sub>O emissions

may therefore result in a more general model performance for estimating GHG impacts with management practices over a broader area.

Measured data were converted into units comparable to DAYCENT model results, expressed in units of g C m<sup>-2</sup> for sugarcane production, g N<sub>2</sub>O-N ha<sup>-1</sup> yr<sup>-1</sup> (or day<sup>-1</sup>, or cumulative over the 15- and 50-day experiments) for nitrous oxide flux, and g C m<sup>-2</sup> from 0 – 20cm for soil organic matter. Aboveground production and stalk yields were reported as either weight at harvest, dry weight, or C mass per unit area (Table 5-1). Sugarcane stalks have a percent moisture that can exceed 80% during peak growth, with an ideal range for sugarcane harvest between 70-74% (Bakker, 2012). No moisture contents were directly reported in studies with data on stalk harvest weights. If stalk measurements were not reported in terms of C mass per unite area, the measurements were assumed to contain 70% water and 45% C content, using these values to convert harvest or dry weight into g C m<sup>-2</sup>. This approach is similar to that used by Galdos et al., (2009) in a CENTURY model study of sugarcane production, but may overestimate stalk C if 70% water is lower than the true water content at harvest.

Soil C data from Mello et al. (2014) consist of a large set of paired field comparison sites (under different land uses), which included all sites where sufficient data were available for DAYCENT model simulations. We were able to include 63 of the original 135 study sites, as well as 28 of the original 75 conversion pairs, across eight regions (Table 5-1). This subset of sites predominantly represents areas of pasture conversion into sugarcane, although one region included conversion of two native vegetation sites into sugarcane while another included three sites with conversion of agricultural lands into sugarcane (Table 5-1). The eight regions are geographically spaced across the longitudinal gradient of the main South Central sugarcane producing region in Brazil (Figure 5-1). This study also provides the only dataset where sufficient data were available to split their use between DAYCENT parameterization versus out-of-sample validation of DAYCENT performance. We therefore divided these sites into two groups, using one group of sites across four regions (Itirapina, Ipaussu, Igarapava, and Gioiatuba) for soil

C parameterization while reserving those located in the other four regions (Anhembi, Arapora, Maracaju, and Iacanga) for soil C validation. We split these sites randomly across the longitudinal extent of the South Central region, such that parameterization and validation sites were located across comparable longitudinal reach (Figure 5-1). Insufficient sites were available to extend parameterization versus validation sites across comparable latitudinal area. The sites in the regions reserved for DAYCENT validation included >70% of all of the Mello et al. (2014) sites simulated in this analysis, or 46 of the 63 total.

To evaluate N<sub>2</sub>O emissions we used three studies with direct measurement of N<sub>2</sub>O emissions in four locations, representing 27 different experimental treatments that included burned versus unburned harvest, application of vinasse, varying levels and forms of synthetic fertilizer addition, and different amounts of residue return following harvest (Table 5-2). Exact experimental start dates were variably reported; Carmo et al. (2013) reported the experimental start day for one location (Piracicaba), while only reporting the starting month for the other location (Jaú). Signor et al. (2013) reported the exact start date, while Oliveira et al. (2013) indicated only that the experiment started in July and continued into August. When not reported, we used estimated values to schedule the days of experimental events in DAYCENT simulations. In all studies, N<sub>2</sub>O emissions measured using static polyvinyl chloride chambers had between 3 and 5 replicate chambers per treatment.

### 5.2.2 DAYCENT model overview

The DAYCENT model runs on a daily time-step and simulates soil processes by horizon across the soil profile, including soil nutrient movement, water movement, plant growth, and many other ecosystem components (Del Grosso et al., 2008). The key drivers of DAYCENT include maximum and minimum daily temperature, daily precipitation, soil texture, and land management (including specific plant types grown and soil management such as tillage and nutrient additions). Nitrous oxide emissions are derived from nitrification and denitrification, based on soil water content and soil texture, NH<sub>4</sub><sup>+</sup> and

NO<sub>3</sub><sup>-</sup> concentration, and availability of labile C as estimated using heterotrophic respiration as a proxy (Parton et al., 2001). Nitrate is mobile and can be either leached or taken up by plants. It is therefore controlled by interactions with either plants or water movement across the soil profile. Ammonium is immobile and only distributed in the top 15cm of the soil profile. On a daily time step, denitrification is determined by soil texture, water content, temperature, NO<sub>3</sub><sup>-</sup> availability, and heterotrophic respiration, while nitrification is determined by the same soil factors (texture, water content, temperature) alongside NH<sub>4</sub><sup>+</sup> concentration only in the 0 – 15cm layer (Del Grosso et al., 2008). The DDcentEVI version of DAYCENT (a version of DAYCENT with the option to use Enhanced Vegetation Index- i.e. EVI- data) was used for this analysis. DDcentEVI was developed and tested to estimate total agricultural land use emissions for the US Environmental Protection Agency's (EPA) GHG emission inventory annual report (U.S. Environmental Protection Agency, 2013).

Temperature and precipitation are key drivers of biogeochemical processes, and are particularly important in the simulation of N<sub>2</sub>O emissions. Long-term, continuous daily climate datasets are therefore needed to both represent regional climate patterns in spin-up simulations—i.e. that bring DAYCENT to equilibrium conditions prior to simulating experimental treatments—as well as to drive simulations for the experimental periods of interest. As mentioned above, in this study we considered two climate datasets for simulating N<sub>2</sub>O flux from experimental treatments, in addition to directly reported climate data. The first climate dataset consisted of climate data drawn from the nearest available meteorological station to each study location, provided by the *Agrometeorological Monitoring System* (AGRITEMPO). These data ranged in their temporal span, starting as early as 1917 and as late as 2007, and extending mainly to 2010 although in the Maracju region data were available until 2011. Our second climate data source was the Climate Forecast System Reanalysis (CFSR) global meteorological data are given at a 38-km scale from 1979 – 2013

using algorithms to interpolate weather for areas between weather stations. The CFSR dataset is under active evaluation and is considered, in the tropics, comparable to a quality weather station at a 30 – 40km distance from the experimental location (D. Fuka, personal communication). These climate datasets vary considerably from each other and from climate data reported in the studies considered in this analysis, in terms of temperature biases, the timing, magnitude, and cumulative amount of precipitation, as well as in the prevalence of missing data. A summary of reported climate versus the two climate datasets used to drive DAYCENT simulations in this analysis are reported in Table 5-2.

Soil texture and prior land use history used for model initialization were drawn, where available, from data reported in the publications included in this study, as well as from additional publications presenting different datasets from the same experimental sites. For the Mello et al. (2014) land conversion sites, additional land use history and sugarcane production practices data were acquired directly from managers of nearby sugarcane processing facilities. In addition to the 63 land use conversion sites from Mello et al. (2014), 42 different sugarcane management practices were simulated matching experimental management practices in the 6 studies that provided the sugarcane production and N<sub>2</sub>O emission measurements used in this study (Table 5-1).

### 5.2.3 Climate data modifications for simulating N<sub>2</sub>O emissions

In order to evaluate the accuracy with which the two climate datasets described above reflect environmental conditions driving N<sub>2</sub>O emissions, in this analysis we chose to focus on reported versus climate dataset estimations of precipitation. Specifically, in our analysis we compared measured versus DAYCENT modeled N<sub>2</sub>O emissions for all treatments as estimated using unaltered version of each climate dataset, and as estimated using as many directly reported precipitation data as were available for each experimental site. Only mean annual climate data were reported in Carmo et al. (2013) (Table 5-2). Therefore we simulated these treatments using only each of the two climate datasets, without alteration. In Signor et al. (2013), daily precipitation data were reported for the 50-day experimental

period. Therefore these treatments were simulated using each of the two climate datasets, without alteration, as well as simulated using each of the climate datasets with the reported precipitation data integrated in place of the original values. In the 15-day Oliveira et al (2013) study only monthly summaries of climate data were reported for the year in which the experiment took place (Figure 5-7). However, water-filled pore space was reported for the 15-day experimental period. The water-filled pore space data indicated a likely precipitation event early in the experimental period not reflected in either of the two climate datasets, while the reported monthly summaries of precipitation data show that each of the climate datasets were substantially underestimating precipitation during the experimental period (Figure 5-7). Therefore the treatments in Oliveira et al. (2013) were simulated using each of the two climate datasets, without alteration, as well as using each of the two climate datasets were substantially underestimating precipitation during the experimental period (Figure 5-7). Therefore the treatments in Oliveira et al. (2013) were simulated using each of the two climate datasets, without alteration, as well as using each of the two climate datasets were precipitation was added, evenly divided on a daily basis for the month preceding the experimental period.

### 5.2.4 Statistical Analyses

Statistical analyses were written in R-3.1.1 software (R Core Team, 2014), using tidyr (Wickham, 2014), dplyr (Wickham and Francois, 2015), hydroGOF (Zambrano-Bigiarini, 2014), and ggplot2 packages (Wickham, 2009). Regression analyses were used to compare measured versus modeled total C in aboveground sugarcane production (g C m<sup>-2</sup>), C in sugarcane stalk production (g C m<sup>-2</sup>), total soil C (g C m<sup>-2</sup>), and the difference in soil C between paired Mello et al. (2014) land use conversion sites (g C m<sup>-2</sup>). Regression analyses were additionally used to compare measured versus modeled cumulative and daily N<sub>2</sub>O flux (g N<sub>2</sub>O-N ha<sup>-1</sup> yr<sup>-1</sup>/day<sup>-1</sup>/15-day<sup>-1</sup>/50-day<sup>-1</sup>) across all sites where these data were available. Root mean square errors were calculated for all comparisons, to evaluate relative model performance across different types of measured observations. A two-tailed test was used to compare measured versus modeled validation soil C results by region, calculating 95% confidence intervals (mean +/- the

standard error \* 1.96), where overlap between confidence intervals indicates no statistical difference between means.

## 5.3 Results

Primary productivity and soil C are a primary focus for evaluating baseline DAYCENT performance; in either case in this analysis, DAYCENT was able to capture trends in treatment effects on sugarcane yields and soil, but with high variability. For example, after parameterization, DAYCENT simulated total aboveground sugarcane production with minimal bias but high dispersion, with a slope close to 1 and an intercept that did not differ significantly from 0, but a low adjusted r<sup>2</sup> of 0.3 (Figure 5-2, A). Stalk production had higher dispersion (a lower adjusted r<sup>2</sup>) and more bias, with a slope less than 1 and an intercept that significantly differed from 0 (Figure 5-2, B). However stalk yield data were more uncertain due to unreported percent water content at harvest. Therefore, we have greater confidence in the quality of the total aboveground measured data in evaluating DAYCENT performance (Figure 5-2, A). Predicted C:N ratios of the sugarcane trash were consistent with reported values, ranging from 79-100 (data not shown), which are very similar to the 80 – 100 C:N ratio reported in published literature (Gava et al., 2005). DAYCENT may therefore be variable in how accurately it simulates the quantity of litter input that it simulates for sugarcane, but DAYCENT is relatively consistent in simulating the litter's C:N ratio.

DAYCENT did reflect overall trends in soil C in most regions simulated in this analysis, but given the high variability of the measured soil C data it is difficult to be conclusive in the valuation of overall DAYCENT performance for soil C dynamics. After parameterization, model predictions of total soil C showed a significant relationship, with a higher r<sup>2</sup> than production predictions (Figure 5-2), and an intercept that did not differ significantly from 0. However, model predictions showed a tendency to overestimate low soil C values while underestimating high soil C values, as indicated by having a slope <1 (Figure 5-3). The set of regions from Mello et al. (2014) used to validate the DAYCENT model showed

variable model performance by region. In simulating total soil C and the difference between paired LUC sites included in Mello et al (2015), model performance was more comparable to measured values in Arapora and Anhembi, performed more poorly estimating total soil C in lacanga, and did not accurately estimate either total soil C or the difference in soil C between paired sites in Maracaju (Figure 5-4).

DAYCENT simulations of  $N_2O$  emissions were highly sensitive to precipitation data inputs. Precipitation data from the two climate datasets considered in this study were, in turn, variable relative to reported precipitation information. In estimating annual N<sub>2</sub>O emissions for the two sites in the Carmo et al. (2013) study, DAYCENT performance was improved using the nearest weather station versus the interpolated dataset (Figure 5-5, A), but with both analyses showing a tendency to overestimate annual  $N_2O$  emissions (intercept significantly greater than 0) that was more pronounced with the interpolated climate dataset (Figure 5-5, A). In Carmo et al. (2013) the only reported climate data were annual averages (Table 5-2). It is therefore difficult to assess this site's model performance and sensitivity related to precipitation data on a finer timescale. The model estimation of annual emissions in Carmo et al. (2013) using the nearest weather station climate dataset provided the best estimate of treatment effects on relative  $N_2O$  emissions, with a slope closest to 1 of all cumulative modeled/measured comparison (Figure 5-5). The overestimation of annual emissions using interpolated climate data was likely impacted by a bias towards high annual precipitation in the dataset for this location (Table 5-2). Cumulative measured versus modeled comparisons for the shorter-term experiments (50 days for Signor et al. (2013) (Figure 5-5 & 5-6), 15 days for Oliveira et al. (2013)) showed poorer DAYCENT simulation of treatment effects, with the latter having no significant relationships between measured and modeled values using either climate dataset.

In Signor et al. (2013), precipitation measurements for the experimental period were directly reported; however, these precipitation data did not improve cumulative model performance when incorporated into the climate datasets used to drive model simulations (Figure 5-6, A & B). These results

suggest the precipitation *preceding* the experimental period is important for accurately capturing N<sub>2</sub>O emissions *during* the experimental period; model performance was improved using reported precipitation in the interpolated climate dataset, which has higher overall precipitation (Figure 5-5 & 5-6, B).

However, on a daily basis direct precipitation measurements did support more accurate N<sub>2</sub>O emission estimates in some treatments from Signor et al. (2013). Using the interpolated dataset and reported precipitation, DAYCENT performed well with both the timing and magnitude of daily N<sub>2</sub>O emissions in Signor et al., when N additions and measured N<sub>2</sub>O emissions were low (Figure 5-6, C). However these model simulations increasingly diverge as N additions and measured N<sub>2</sub>O emissions (180 kg N ha<sup>-1</sup> ammonium nitrate) was excluded from these measured/modeled comparisons (Figure 5-6, A & B), to better evaluate the range of treatments where DAYCENT currently shows greater success. DAYCENT performance for the specific 180 kg N ha<sup>-1</sup> ammonium nitrate treatment was universally poor, indicating factors that result in exceptionally high emissions in sugarcane systems are not represented well in DAYCENT.

Oliveira et al. (2013) cumulative simulated emissions were 1 – 2 orders of magnitude less than reported measurements, with no significant relationship between cumulative measured versus modeled values using either of the two climate datasets. This indicates very poor model performance at this site. However, while summary climate data would suggest these climate datasets are estimating excessively high precipitation on an annual basis for Oliveira et al. (2013) (Table 5-2), a comparison of monthly reported versus estimated precipitation reveals a bias towards very low precipitation estimates in both climate datasets during the experimental period (7.2 cm in July reported, versus 1.8cm from the nearest weather station and 2.4cm from the interpolated climate dataset, Figure 5-7). When the difference in precipitation was added on an even daily basis in the month preceding the experimental period, model

performance for daily emissions in the burned treatments improved substantially, although showing more sensitivity to water additions than to N addition with vinasse (Figure 5-8). Model simulations of mechanical harvest treatments were also improved in these simulations but still an order of magnitude too low relative to measured values (data not shown). Therefore there are structural improvements needed for DAYCENT to better simulate the impact on N<sub>2</sub>O emissions of high litter inputs added back to the soil with mechanical harvest. The exact experimental start date was not reported in Oliveira et al. (2013). Model variability based on available climate data versus the need for structural model improvements would be better clarified by lining up the precipitation data and the experimental period more exactly, particularly with such a short-term experiment.

Modeled estimates of N<sub>2</sub>O emission factors—the percentage of N<sub>2</sub>O-N released per unit of N applied to the soil—showed high variability, due to both variability in climate data accuracy as well as the need for improved model simulation of very high N<sub>2</sub>O emission periods and locations. Only the original climate datasets were used in this analysis of EFs. Model EFs ranged from greater than 5% to less than 0.1%. This range therefore extends an order of magnitude lower and at least 5x higher than the IPCC emission factor (Figure 5-9). This variability was comparable to measured variability that exhibited a similar high range, but with a lowest value of only 0.31%. Regardless of climate data source, the highest measured EF was not successfully predicted by model simulations. The tendency of N<sub>2</sub>O overprediction in the Piracicaba site in Carmo et al. (2013), in turn, resulted in extremely high predicted emission factors.

## 5.4 Discussion

In this analysis, the DAYCENT model was parameterized using measured data for sugarcane aboveground production, soil C and N<sub>2</sub>O emissions drawn from publications that represented a range of management practices and land use conversions of key concern in determining the sustainability of Brazilian sugarcane production. The results from the validation component of this study, evaluating

DAYCENT performance simulating soil C, demonstrate that the DAYCENT model is capable of simulating soil C trends under different land conversion scenarios, but with variable regional performance. The poorest performance in the validation component of this study was in the simulation of sites in the Maracju region. However Maracaju differs from the all other sites used in parameterization in two respects. First, Maracaju is situated furthest west and is geographically distanced from all other sites (Figure 5-1). Second, sites in the Maracaju region include only land use conversion from active agricultural production into sugarcane. The sites used for parameterization in this study did not include any conversion of agricultural production into sugarcane. Poor DAYCENT performance simulating Maracaju sites indicates a need to expand DAYCENT parameterization into both this region as well as into the evaluation of agricultural land conversion into sugarcane production. This would allow for DAYCENT use simulating land conversion of key concern in the development of Brazilian sugarcane (Carvalho et al., 2009; de Souza Braz et al., 2013; Mello et al., 2014).

With the exception of Maracaju, model results generally matched the direction and variability of mean regional measured values of differences in soil C between paired land use conversion sites (Figure 5-4, B). As discussed above, the sites in these remaining regions predominantly represent pasture conversion into sugarcane. Conversion from pasture into sugarcane was shown in Mello et al. (2014) to result in a cumulative soil C loss, a trend reflected in measured means in these regions (Figure 5-4, B). Our modeled results support that trend, but with high variability. By necessity the number of paired sites in this DAYCENT validation study was substantially reduced from those included in the original analysis, due to limited data available to drive DAYCENT simulations. Studies show that it is possible for productive soils to store more C than native Cerrado (Carvalho et al., 2009), particularly with high C storage under pasture (Figueiredo et al., 2013) and grasslands due to root contributions (Fidelis et al., 2013). Results shown in Figure 5-4 do reflect the trend of higher C storage in pasture versus sugarcane production. However a greater diversity of soil C measured data are needed to explore other aspects of

soil C change with land management practices, and further evaluate this aspect of DAYCENT model performance.

Model parameterization and validation focused on aboveground sugarcane production and soil C were mainly intended to ensure a sufficient level of basic DAYCENT performance simulating key ecosystem properties. These values are considered benchmarks in the evaluation of land-surface models (Luo et al., 2012), and are similarly a primary concern in assessing baseline DAYCENT performance. In the simulation of N<sub>2</sub>O emissions from sugarcane production management practices, the selection of studies included in this analysis represent the main management practices of concern for impacts on N<sub>2</sub>O emissions—i.e. residue management, mineral fertilizer application, organic fertilizer application, and harvest methods. However, we had insufficient replications of treatments across sites to consider general rather than site-specific estimations of N<sub>2</sub>O emissions with specific management practices. Therefore, we considered model performance in aggregate across management practices, which further demonstrate the challenge of modeling the extreme temporal and spatial heterogeneity of N<sub>2</sub>O emissions in sugarcane production systems.

In this analysis, evaluating DAYCENT simulation of N<sub>2</sub>O emissions was limited by the short 15and 50-day length of the Signor et al. (2013) and Oliveira et al. (2013) experiments, in addition to the variable supplemental data on experimental start dates and precipitation that these studies provided. While these types of short-term measurement periods are valuable in evaluating model performance simulating GHG emissions for specific types of management practice events, model performance will likely diverge from measured values on a daily basis if the timing of experimental treatments is estimated rather than informed directly. For example, model simulations are less likely to estimate the exact timing or the extremes of N<sub>2</sub>O emissions, if estimated days for experimental treatments do not line up with the environmental conditions when the measurements were taken—e.g. if simulated fertilizer application occurs in a dry series of days, when in reality it occurred in close proximity to a

large rain event. Additional short-term studies would be useful for further evaluating N<sub>2</sub>O emissions from sugarcane management practices of interest, and support DAYCENT estimations for the GHG impacts of specific management practices on a more regional basis. However, longer-term N<sub>2</sub>O emission measurement time series are also needed, to better determine the time scale at which DAYCENT can most accurately estimate N<sub>2</sub>O flux from sugarcane production.

Despite their limitations, the data used in this analysis do give widespread support to the importance of using high quality, long-term climate data to drive the evaluation of N<sub>2</sub>O emissions from Brazilian sugarcane production practices using DAYCENT. In this analysis we experimented with different forms and combinations of weather data, using the two climate datasets in addition to reported climate information. The results suggest that climate data need to accurately reflect conditions before and during experimental treatments being evaluated, in order to more closely approximate the N<sub>2</sub>O emission dynamics that occur from specific types of sugarcane management practices. This again addresses the need, mentioned above, for longer-term N<sub>2</sub>O measured datasets. In other studies DAYCENT performance tends to improve when evaluating N<sub>2</sub>O emissions over longer temporal periods: the model may not accurately capture specific daily emission events, particularly if the driving climate data do not reflect true climate conditions. However, DAYCENT tends to compensate over longer time-spans, again as long as the driving climate data reflect true conditions over the long term. Poor climate data—for example simulating inaccurately dry conditions or excessive precipitation when N sources are applied, as seen in this analysis—can substantially over- or under-estimate N<sub>2</sub>O flux, sometimes even when a period of directly measured climate data are incorporated (Figure 5-6, A & B).

While climate data variability was responsible for some poor DAYCENT performance simulating N<sub>2</sub>O flux in this study, DAYCENT also showed an inability to simulate the sometimes extremely high N<sub>2</sub>O emissions than can occur in Brazilian sugarcane systems. In another study (Campbell et al., 2014), DAYCENT tended to perform more poorly simulating high levels of N<sub>2</sub>O emissions (Figure 5-5 B),

particularly at a daily level (Figure 5-8). In this study the measured data (i.e. from Signor et al. (2013)) show the potential for conditions to occur in sugarcane that can release very high emission, that were subsequently missed by sometimes several orders of magnitude in DAYCENT. Similar conditions have been observed in Austrialian sugarcane systems, and are currently undergoing evaluation using DAYCENT (W. Parton, *personal communication*). Although these high emission events may be rare and short-term, they can be a substantial contributor to total N<sub>2</sub>O emissions. One aspect of DAYCENT performance that may be contributing to these patterns of underestimation is DAYCENT's lack of explicit dissolved organic carbon simulation, which could transport some of the very large amounts of surface litter inputs following mechanical harvest in a form and a direction that would support higher denitrification in the soil. However, DAYCENT does not yet simulate the movement of DOC across the soil profile, although this feature is currently in development (Campbell et al., in preparation for Soil Biology and Biochemistry). The lack of sensitivity of the mechanically harvested treatments in Oliveira et al. (2013) to additional precipitation, as compared to the burned harvested treatments, supports that a limiting factor related to litter inputs is likely contributing to poor DAYCENT N<sub>2</sub>O emission simulation of those treatments.

Our results further demonstrate the challenge of simulating the extreme variability in emissions factors for N<sub>2</sub>O. Certainly the standard Tier 1 IPCC value may not reflect true N<sub>2</sub>O emissions from a given location or sugarcane management practice. However, while DAYCENT modeling would support a Tier 2 IPCC approach to GHG evaluation, the large discrepancy between modeled versus measured evaluation of high N<sub>2</sub>O emissions must first be addressed, and a robust climate dataset to drive regional DAYCENT simulations either identified or developed.
Table	5-1	Site	descri	ptions.
-------	-----	------	--------	---------

Study	Collection Period	Data description	Lat/ Long	Texture	Description	
N <sub>2</sub> O Emissions						
(Carmo et al., 2013)- Piracicaba	335 days, on event basis	Est. daily & annual flux, from replicated static polyvinyl chloride chambers	-22.68/ -47.55	Sandy Clay Loam	Sugarcane production from 1990, unburned harvest, 0, 7, 14, 21 Mg/ha trash w vinasse	
(Carmo et al., 2013)- Jaú	314 days, on event basis	Est. daily & annual flux, from replicated static polyvinyl chloride chambers	-22.25/ -48.57	Sandy Loam	Sugarcane production since 1990, unburned harvest, mineral fertilizer & vinasse	
(Signor et al., 2013)- Piracicaba	50 days, every 1-2 days	Est. daily & cumulative flux, from replicated static polyvinyl chloride chambers	-22.72/ -47.63	Clay	Urea and ammonium nitrate fertilizer applications	
(de Oliveira Bordonal et al., 2012)- Boa Vista Farm Piracicaba	15 days, every 1-2 days	Est. daily & cumulative flux, from replicated static polyvinyl chloride chambers	-18.72/ -46.97	Clay	Burned vs unburned, vinasse vs water	
Sugarcane Product	ion					
(Franco et al., 2010)- Pirassunuga	Plant cane harvest (1 yr)	Dry matter (kg ha <sup>-1</sup> ): total, stalk	-21.92/ -47.17	Clay	Different rates of urea applied to crops at planting	
(Franco et al., 2010; Otto et al., 2009)- Jaboticabal	Plant cane harvest (1 yr)	Dry matter (kg ha <sup>-1</sup> ): total, stalk	-21.32/ -48.32	Sandy Clay Loam	Different rates of urea applied to crops at planting	
(Hassuani et al., 2005)- Usina São Martinho	Plant cane harvest + 5 ratoons (6 yrs)	Harvest weight (t ha <sup>-1</sup> ): aboveground total, trash	-22.36/ -47.38	Sandy Clay Loam	unburned harvest with 0, 33%, 66%, and 100% trash	
(Hassuani et al., 2005)- Usina Santa Luisa	Plant cane harvest + 1 ratoon (2 vrs)	Harvest weight (kg plot <sup>-1</sup> ): aboveground total, stalk, trash	-22.58/ -47.52	Clay	unburned harvest of sugarcane	
(M V Galdos et al., 2009)- Usina São Martinho	Plant cane harvest + 7 ratoons (8 yrs)	Stalk C (g m <sup>-2</sup> ), assuming 70% water content & 44% C content from fresh weight	-21.37/ -48.05	Clay	burned vs unburned sugarcane	
Soil Carbon						
(Mello et al., 2014)- Igarapava	Sampled with replicates, 2010	Soil C (%) by 10cm layer from 0 – 100cm	-20.16/ -47.77	Clay	7 sites: sugarcane/pasture conversion, 2 conversion pairs (parameterization)	
(Mello et al., 2014)- Ipaussu	Sampled with replicates, 2010	Soil C (%) by 10cm layer from 0 – 100cm	-23.11/ -49.52	Clay	5 sites: sugarcane/pasture conversion, 3 conversion pairs (parameterization)	
(Mello et al., 2014)- Itirapina	Sampled with replicates, 2010	Soil C (%) by 10cm layer from 0 – 100cm	-22.37/ -47.91	Loamy Sand	4 sites: sugarcane/pasture, 3 conversion pairs (parameterization)	
(Mello et al., 2014)- Gioiatuba	Sampled with replicates, 2010	Soil C (%) by 10cm layer from 0 – 100cm	-18.09/ -49.62	clay loam	1 sugarcane/pasture conversion site (parameterization)	
(Mello et al., 2014)- Anhembi	Sampled with replicates, 2010	Soil C (%) by 10cm layer from 0 – 100cm	-22.75/ -48.02	Loamy Sand	11 sites: sugarcane/pasture conversion, 4 conversion pairs (validation)	
(Mello et al., 2014)- Arapora	Sampled with replicates, 2010	Soil C (%) by 10cm layer from 0 – 100cm	-18.47/ -49.12	Clay	25 sites: sugarcane/pasture/ native conversion, 13 conversion pairs (validation)	
(Mello et al., 2014)- lacanga	Sampled with replicates, 2010	Soil C (%) by 10cm layer from 0 – 100cm	-21.97/ -49.04	Loamy Sand	4 sites: sugarcane/pasture conversion, 3 conversion pairs (validation)	
(Mello et al., 2014)- Maracaju	Sampled with replicates, 2010	Soil C (%) by 10cm layer from 0 – 100cm	-21.43/ -55.35	Clay	6 sites: agriculture/sugarcane conversion, 3 conversion pairs (validation)	

Table 5-2 Summary climate data for the four sites with  $N_2O$  emission data, as reported and as calculated from weather data given by the nearest weather station versus the interpolated dataset from the Climate Forecast System Reanalysis global meteorological dataset .

Site	<b>Reported Climate</b>		Nearest Weather Station			Interpolated	
	Mean	Mean	Mean	Mean	Missing	Mean	Mean
	Annual	Temp	Annual	Temp	data	Annual	Temp
	Precip. (cm)	(C)	Precip. (cm)	(C)	(days/total)	Precip. (cm)	(C)
Carmo- Piracicaba	139	21	132.7	22.7	91/2922	178	22.1
Carmo- Jaú	139	21	129.9	24.2	312/2557	133.9	23
Signor- Piracicaba	127	21.5	130.9	22.5	101/4383	170.8	22.2
Oliveira- Piracicaba	141.9	NR	122.2	23.3	13/3287	166.9	21.9



Figure 5-1 Location of sites used to parameterize and validate the DAYCENT model. Validation sites were selected to represent similar longitudinal gradients as sites used to parameterize the model.



Figure 5-2 Within-sample measured (x-axes) versus modeled (y-axes) total aboveground production (A) and stalk production (B), both showing significant relationships (adjusted  $r^2$ =0.3, slope=1.05, p=0.001, RMSE=497 and adjusted  $r^2$ =0.25, slope=0.65, p<0.001, RMSE=396, respectively). However, while the total production had an insignificant intercept (intercept=25), the stalk production intercept was weakly significant (intercept=487, p=0.04). The 1:1 relationships are shown with the solid line, while linear regression relationships are shown with the dotted lines and shading.



Figure 5-3 Within-sample measured (x-axes) versus modeled (y-axes) total soil C, showing a significant relationship and insignificant intercept (adjusted  $r^2$ =0.4, slope=0.67, p=0.008, RMSE=791; intercept=1660). The 1:1 relationship is shown with the solid line, while the linear regression relationship is shown with the dotted lines and shading.



Figure 5-4 Out-of-sample measured versus modeled prediction of total soil carbon (A) and the difference in soil carbon between paired land use change sites from Mello et al. (2014) (B), for DAYCENT validation, showing 95% confidence intervals as well as the number of sites included in these analyses in each region. In A & B, Maracaju modeled and measured confidence intervals did not overlap. Iacanga total soil C modeled results were substantially higher than measured values (A), but soil C difference confidence intervals for the 3 sites with LUC pairs did overlap (B). Modeled and measured confidence intervals for Anhembi and Arapora overlapped in total (A) as well as by difference in paired sites (B).



Figure 5-5 Measured (x-axes) versus modeled (y-axes) predictions of cumulative N<sub>2</sub>O emissions for both sites in the long-term study by Carmo et al. (2013) (A) versus the Piracicaba sites in the 50-day study by Signor et al. (2013) (B), using nearest weather station versus interpolated climate data. Both types of climate data resulted in significant relationships between measured and modeled values in the Carmo et al. (2013) study (adjusted  $r^2$ =0.38, slope=0.98, p=0.011, RMSE=4700; and adjusted  $r^2$ =0.32, slope=2.1, p=0.021, RMSE= 14000, A), but intercepts that were significantly different than 0 (intercept=4799, p=0.0036; intercept=8900, p=0.015, A). In the Signor et al. (2013) study the nearest weather station data resulted in a significant relationship (adjusted  $r^2$ =0.66, slope=0.26, p=0.0047, RMSE=6771) and an intercept that was not significantly difference from 0 (intercept = 762, B). However, the interpolated climate data did not result in a significant relationship between measured values and modeled prediction. The 1:1 relationships are shown with the solid line, while linear regression relationships are shown with the dotted lines and shading.



Figure 5-6 Measured (x-axis) versus modeled (y-axes) predictions, excluding the 180 kg N ha<sup>-1</sup> ammonium nitrate treatment, of cumulative N<sub>2</sub>O emissions over 50 days in Signor et al. (2013) treatments—as estimated using the nearest weather station (A) and interpolated (B) climate datasets either with (light points) or without (dark points) the use of reported precipitation data—alongside daily measured (red line) versus modeled (blue and green lines) N<sub>2</sub>O emissions in the 60 and 80 kg N ha<sup>-1</sup> ammonium nitrate treatments (C & D, respectively), considering only climate datasets where reported precipitation data were included. Cumulative emissions all showed significant relationships between measured and modeled values, with intercepts that did not differ significantly from 0 (adjusted  $r^2$ =0.62, p=0.012, RMSE=3295, intercept = -38 and adjusted  $r^2$ =0.65, p=0.0094, RMSE=5200, intercept=197, original versus with reported climate data, A; adjusted  $r^2$ =0.70, p=0.0057, RMSE=2200, intercept = 1394 and adjusted  $r^2$ =0.61, p=0.013, RMSE= 4000, intercept=750, B). However, slopes were substantially different, at values of 0.45, 0.05, 0.48 and 0.19, A & B, respectively. DAYCENT successfully simulates the timing of N<sub>2</sub>O flux as N application increases (D).



Figure 5-7 Monthly cumulative precipitation for 2009, as reported by Oliveira et al. (2013) (black bars) versus from the nearest weather station and interpolated climate datasets (medium and light bars, respectively). Biases vary by time of year, type of climate data, and climate dataset. Notably, both climate datasets tend to overestimate precipitation in wet months and underestimate precipitation in dry months relative to reported values.



Figure 5-8 Precipitation data used to drive model simulations for the Oliveira et al. (2013) experimental period—as originally reported in the nearest weather station (A) and interpolated (B) climate datasets as well as with the addition of precipitation in the month preceding the experimental period in order to match, by day 210, the reported precipitation for July, 2009 (C & D)—alongside measured versus modeled predictions of daily N<sub>2</sub>O emissions from application of water versus vinasse in a burned harvest site (E & F, respectively). Climate drivers both during and preceding the experimental periods can strongly affect DAYCENT simulation of either the magnitude or the timing of high N<sub>2</sub>O emissions. Simulated N<sub>2</sub>O emissions are more affected by the addition of water than N in these simulations; modeled N<sub>2</sub>O emissions are comparable in both treatments in timing and magnitude, while measured values are much lower when only water is applied.



Figure 5-9 Measured versus modeled emission factors (EF) for treatments within each  $N_2O$  study, as compared to IPCC emission factor for N addition in agriculture (dotted black line), using nearest weather station climate data (A) versus interpolated climate data (B). On the x-axis, the first treatment in each study is labeled, and all treatments to the right of the label until the subsequent label fall within that study. OL refers to 2 treatments with vinasse additions in Oliveira et al. (2013), SIG to the 8 urea and ammonium nitrate fertilizer addition treatments in Signor et al. (2013), CAR1 to the vinasse + residue return treatments in Piracicaba and CAR2 to the fertilizer treatments in Jaú, respectively, from Carmo et al. (2013). Emission factors are highly variable, with model results under predicting the highest emission factor using both types of climate data. However, interpolated climate data resulted in model predictions that substantially overestimated more treatments (predicting EFs of >5% with 5 treatments (B), versus predicting an EF >5% at only 1 treatment (A)).

## 6 CONCLUSION

The research presented in this dissertation examines the theory and development of SOM modeling, as well as its application in evaluating crop-based bioenergy production systems when integrated into the DAYCENT ecosystem model. There remains a great deal of uncertainty in how SOM heterogeneity and variability integrates across temporal and spatial scales, e.g. whether mechanisms and empirical relationships known to be of importance at small scales continue to effect cumulative SOM dynamics at larger scales. As a result, SOM modeling structure and approaches must carefully consider the scale at which they are being applied, both in terms of the data used to drive SOM models as well as in terms of model structure and formulation.

The review chapter supported the importance of scale in across advancing areas in SOM modeling development, particularly as SOM models of increasing complexity are integrated into earth system models. Clarification and consistency in the underlying assumptions for either SOM formulation or data used to drive SOM simulations are of high importance for crossing scales from microsite to regional and global levels. Beyond the issue of scale, the review chapter also suggests increasing sophistication in data-model integration in SOM modeling is needed, to capitalize on collaborations across multiple disciplines as well as advances in computational capacity. These efforts can help identify and target SOM measurement and SOM model parameterization accommodated high variability and uncertainty in measurements as well as in the LIDEL model formulation. The data used for LIDEL model analysis were able to strongly inform three of five unknown parameters, and suggested targeted data evaluating microbial processes during litter decomposition would resolve the high uncertainty for the remaining two parameters. The LIDEL model itself provides an innovative new approach to litter

decomposition simulations that includes both variable microbial CUE and the simulation of microbial products and DOC generation. This is a first step in meeting some of the shortfalls in SOM models discussed in the preceding review chapter.

The final two chapters return to the issue of scale, consistency, and clarity in the application of SOM models to evaluated crop-based bioenergy production systems. In the analysis of corn stover residue removal, DAYCENT model results support the importance of compensating for the removal of biomass with management practices that contribute towards soil C accumulation and storage—e.g. reduced tillage, cover cropping, or additional fertilizer application to increase crop productivity. In the analysis of sugarcane production in Brazil, model results support measured trends in soil C loss with conversion of pasture land into sugarcane production. However, both analyses highlight poor DAYCENT performance simulating N<sub>2</sub>O emissions, which are of high concern in estimating the cumulative GHG impact of crop-based bioenergy feedstock production. N<sub>2</sub>O emissions are inherently highly temporally and spatially variable. In both analyses DAYCENT model simulations improved at longer temporal scales, although model performance remained poor in treatments where measured N<sub>2</sub>O emissions were very high. A comparison of nearest weather station versus interpolated, gridded climate data in the analysis of Brazilian sugarcane highlighted the importance of climate data accuracy in driving DAYCENT model simulations, particularly in the estimation of GHG emissions associated with different management practices.

In summation, this body of work suggests that SOM model theory and development must be carefully connected to their applications in evaluating the GHG impacts of crop-based bioenergy production systems. The application of SOM models in these systems must consider both the scale and assumptions used for SOM model formulation, as well as the scale and assumptions behind data used to drive SOM model simulations. As crop-based biofuels expand globally, this further supports the

147

importance of collaborative, multi-disciplinary efforts to support SOM model use in these systems, to ensure the strength of their basis on theory for SOM dynamics.

## REFERENCES

- Abdalla, M., Jones, M., Yeluripati, J., Smith, P., Burke, J., Williams, M., 2010. Testing DayCent and DNDC model simulations of N2O fluxes and assessing the impacts of climate change on the gas flux and biomass production from a humid pasture. Atmospheric Environment 44, 2961–2970. doi:10.1016/j.atmosenv.2010.05.018
- Adair, E.C., Parton, W.J., Del Grosso, S.J., Silver, W.L., Harmon, M.E., Hall, S.A., Burke, I.C., Hart, S.C., 2008. Simple three-pool model accurately describes patterns of long-term litter decomposition in diverse climates. Global Change Biology 14, 2636–2660. doi:10.1111/j.1365-2486.2008.01674.x
- Adler, P.R., Del Grosso, S.J., Parton, W.J., 2007. Life-cycle assessment of net greenhouse-gas flux for bioenergy cropping systems. Ecological Applications 17, 675–691. doi:10.1890/05-2018
- Aerts, R., 1997. Climate, Leaf Litter Chemistry and Leaf Litter Decomposition in Terrestrial Ecosystems: A Triangular Relationship. Oikos 79, 439–449. doi:10.2307/3546886

Ågren, G.I., 2000. Temperature dependence of old soil organic matter. AMBIO 29, 55.

- Ågren, G.I., Bosatta, E., 1996. Quality: a bridge between theory and experiment in soil organic matter studies. Oikos 76, 522–528.
- Allison, S.D., Wallenstein, M.D., Bradford, M.A., 2010. Soil-carbon response to warming dependent on microbial physiology. Nature Geoscience 3, 336–340. doi:10.1038/ngeo846
- Arneth, A., Sitch, S., Bondeau, A., Butterbach-Bahl, K., Foster, P., Gedney, N., de Noblet-Ducoudré, N., Prentice, I.C., Sanderson, M., Thonicke, K., Wania, R., Zaehle, S., 2009. From biota to chemistry and climate: towards a comprehensive description of trace gas exchange between the biosphere and atmosphere. Biogeosciences Discuss. 6, 7717–7788. doi:10.5194/bgd-6-7717-2009

- Averill, C., Turner, B.L., Finzi, A.C., 2014. Mycorrhiza-mediated competition between plants and decomposers drives soil carbon storage. Nature 505, 543–545. doi:10.1038/nature12901
- Baker, J.M., Ochsner, T.E., Venterea, R.T., Griffis, T.J., 2007. Tillage and soil carbon sequestration- what do we really know? Agriculture, Ecosystems & Environment 118, 1–5.

Bakker, H., 2012. Sugar Cane Cultivation and Management. Springer Science & Business Media.

- Baldock, J.., Skjemstad, J.., 2000. Role of the soil matrix and minerals in protecting natural organic materials against biological attack. Organic Geochemistry 31, 697–710. doi:10.1016/S0146-6380(00)00049-8
- Balesdent, J., 1987. The turnover of soil organic fractions estimated by radiocarbon dating. Science of The Total Environment 62, 405–408. doi:10.1016/0048-9697(87)90528-6
- Balesdent, J., Chenu, C., Balabane, M., 2000. Relationship of soil organic matter dynamics to physical protection and tillage. Soil and Tillage Research 53, 215–230. doi:10.1016/S0167-1987(99)00107-5
- Bardgett, R.D., Freeman, C., Ostle, N.J., 2008. Microbial contributions to climate change through carbon cycle feedbacks. The ISME Journal 2, 805–814. doi:10.1038/ismej.2008.58
- Batjes, N. h., 1996. Total carbon and nitrogen in the soils of the world. European Journal of Soil Science 47, 151–163. doi:10.1111/j.1365-2389.1996.tb01386.x
- Bellamy, P.H., Loveland, P.J., Bradley, R.I., Lark, R.M., Kirk, G.J.D., 2005. Carbon losses from all soils across England and Wales 1978–2003. Nature 437, 245–248. doi:10.1038/nature04038
- Benbi, D.K., Boparai, A.K., Brar, K., 2014. Decomposition of particulate organic matter is more sensitive to temperature than the mineral associated organic matter. Soil Biology and Biochemistry 70, 183–192. doi:10.1016/j.soilbio.2013.12.032

- Berliner, L., 1996. Heirarchical Bayesian time series models., in: Hanson, K.M., Silver, R.N. (Eds.), Maximum Entropy and Bayesian Methods. Fundamental Theories of Physics. Springer, New York, NY, pp. 15–22.
- Bidwell, O.W., Hole, F.D., 1965. Man as a factor of soil formation. Soil Science 99, 65–72. doi:10.1097/00010694-196501000-00011
- Blanco-Canqui, H., Lal, R., 2007. Soil and crop response to harvesting corn residues for biofuel production. Geoderma 141, 355–362. doi:10.1016/j.geoderma.2007.06.012
- Bonan, G.B., Levis, S., Kergoat, L., Oleson, K.W., 2002. Landscapes as patches of plant functional types:
  An integrating concept for climate and ecosystem models. Global Biogeochemical Cycles 16, 5–
  1. doi:10.1029/2000GB001360
- Bonan, G.B., Oleson, K.W., Fisher, R.A., Lasslop, G., Reichstein, M., 2012. Reconciling leaf physiological traits and canopy flux data: Use of the TRY and FLUXNET databases in the Community Land Model version 4. Journal of Geophysical Research: Biogeosciences 117, G02026. doi:10.1029/2011JG001913
- Bond-Lamberty, B., Thomson, A., 2010. Temperature-associated increases in the global soil respiration record. Nature 464, 579–582. doi:10.1038/nature08930
- Bosatta, E., Ågren, G.I., 1999. Soil organic matter quality interpreted thermodynamically. Soil Biology and Biochemistry 31, 1889–1891. doi:10.1016/S0038-0717(99)00105-4
- Bosatta, E., Ågren, G.I., 1991. Dynamics of carbon and nitrogen in the organic matter of the soil: a generic theory. The American Naturalist 138, 227–245.
- Braakhekke, M.C., Wutzler, T., Beer, C., Kattge, J., Schrumpf, M., Ahrens, B., Schöning, I., Hoosbeek, M.R., Kruijt, B., 2013. Modeling the vertical soil organic matter profile using Bayesian parameter estimation. Biogeosciences 10, 399–420.

- Brady, N.C., Weil, R.R., 2007. The Nature and Properties of Soils, 14th Edition, 14 edition. ed. Prentice Hall, Upper Saddle River, N.J.
- Burgess, M.S., Mehuys, G.R., Madramootoo, C.A., 1996. Tillage and crop residue effects on corn production in Quebec. Agronomy Journal 88, 792–797.
- Campbell, C.A., Paul, E.A., Rennie, D.A., McCallum, K.J., 1967. Applicability of the carbon-dating method of analysis to soil humus studies. Soil Science 104, 217–224.
- Campbell, E.E., Johnson, J.M.F., Jin, V.L., Lehman, R.M., Osborne, S.L., Varvel, G.E., Paustian, K., 2014. Assessing the Soil Carbon, Biomass Production, and Nitrous Oxide Emission Impact of Corn Stover Management for Bioenergy Feedstock Production Using DAYCENT. BioEnergy Research 7, 491–502. doi:10.1007/s12155-014-9414-z
- Campbell, E.E., Parton, W.B., Soong, J.L., Hobbs, N.T., Cotrufo, M.F., Paustian, K., In Preparation. Tracking the fate of litter C into CO2, microbial products and DOC by the Litter Decomposition and Leaching (LIDEL) model using a Hierarchical Bayesian approach to estimate parameters and variability. For Soil Biology & Biochemistry.
- Carmo, J.B. do, Filoso, S., Zotelli, L.C., de Sousa Neto, E.R., Pitombo, L.M., Duarte-Neto, P.J., Vargas, V.P., Andrade, C.A., Gava, G.J.C., Rossetto, R., Cantarella, H., Neto, A.E., Martinelli, L.A., 2013. Infield greenhouse gas emissions from sugarcane soils in Brazil: effects from synthetic and organic fertilizer application and crop trash accumulation. GCB Bioenergy 5, 267–280. doi:10.1111/j.1757-1707.2012.01199.x
- Carpenter, S.R., 1981. Decay of heterogenous detritus: A general model. Journal of Theoretical Biology 89, 539–547. doi:10.1016/0022-5193(81)90026-6
- Carvalho, J.L.N., Cerri, C.E.P., Feigl, B.J., Píccolo, M.C., Godinho, V.P., Cerri, C.C., 2009. Carbon sequestration in agricultural soils in the Cerrado region of the Brazilian Amazon. Soil and Tillage Research, Contains papers from HighLand 2006: Land Degradation and Soil and Water

Conservation in Tropical Highlands, Mekelle, Ethiopia, 21-25 September 2006 103, 342–349. doi:10.1016/j.still.2008.10.022

- Castellano, M.J., Kaye, J.P., Lin, H., Schmidt, J.P., 2012. Linking Carbon Saturation Concepts to Nitrogen Saturation and Retention. Ecosystems 15, 175–187. doi:10.1007/s10021-011-9501-3
- Chaopricha, N.T., Marín-Spiotta, E., 2014. Soil burial contributes to deep soil organic carbon storage. Soil Biology and Biochemistry 69, 251–264. doi:10.1016/j.soilbio.2013.11.011
- Chapman, S.K., Langley, J.A., Hart, S.C., Koch, G.W., 2006. Plants actively control nitrogen cycling: uncorking the microbial bottleneck. New Phytologist 169, 27–34. doi:10.1111/j.1469-8137.2005.01571.x
- Cherubini, F., 2010. GHG balances of bioenergy systems Overview of key steps in the production chain and methodological concerns. Renewable Energy 35, 1565–1573. doi:10.1016/j.renene.2009.11.035
- Clapp, C.E., Allmaras, R.R., Layese, M.F., Linden, D.R., Dowdy, R.H., 2000. Soil organic carbon and 13C abundance as related to tillage, crop residue, and nitrogen fertilization under continuous corn management in Minnesota. Soil & Tillage Research 55, 127–142. doi:10.1016/S0167-1987(00)00110-0
- Clark, J.S., 2007. 9.6 Bayesian State Space Models, in: Models for Ecological Data, an Introduction. Princeton University Press, Princeton, NJ, pp. 272–282.
- Clemmensen, K.E., Bahr, A., Ovaskainen, O., Dahlberg, A., Ekblad, A., Wallander, H., Stenlid, J., Finlay, R.D., Wardle, D.A., Lindahl, B.D., 2013. Roots and Associated Fungi Drive Long-Term Carbon Sequestration in Boreal Forest. Science 339, 1615–1618. doi:10.1126/science.1231923
- Collins, M., Knutti, R., Arblaster, J., Dufresne, J.-L., Fichefet, T., Friedlingstein, P., Gao, X., Gutowski, W.J., Johns, T., Krinner, G., Shongwe, M., Tebaldi, C., Weaver, A.J., Wehner, M., 2013. Long-term Climate Change: Projections, Commitments and Irreversibility, in: Stocker, T.F., Qin, D., Plattner,

G.-K., Tignor, M., Allen, S.K., Boschung, J., Nauels, A., Xia, Y., Midgley, P.M. (Eds.), Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press, Cambridge, UK and NY, USA, pp. 1029–1136.

- Conant, R.T., Drijber, R.A., Haddix, M.L., Parton, W.J., Paul, E.A., Plante, A.F., Six, J., Steinweg, J.M., 2008. Sensitivity of organic matter decomposition to warming varies with its quality. Global Change Biology 14, 868–877. doi:10.1111/j.1365-2486.2008.01541.x
- Conant, R.T., Ryan, M.G., Ågren, G.I., Birge, H.E., Davidson, E.A., Eliasson, P.E., Evans, S.E., Frey, S.D., Giardina, C.P., Hopkins, F.M., Hyvönen, R., Kirschbaum, M.U.F., Lavallee, J.M., Leifeld, J., Parton, W.J., Megan Steinweg, J., Wallenstein, M.D., Martin Wetterstedt, J.Å., Bradford, M.A., 2011. Temperature and soil organic matter decomposition rates – synthesis of current knowledge and a way forward. Global Change Biology 17, 3392–3404. doi:10.1111/j.1365-2486.2011.02496.x
- Congressional Research Service, 2007. Energy Independence and Security Act of 2007: a summary of major provisions (Congressional Research Service Report for Congress). Congressional Research Service, Washington, D.C.
- Cotrufo, M.F., Soong, J.L., Horton, A.J., Campbell, E.E., Haddix, M.L., Wall, D.H., Parton, W.J., In Review. All is not lost! Pathways of soil organic matter formation from litter decomposition. Nature Geoscience.
- Cotrufo, M.F., Wallenstein, M.D., Boot, C.M., Denef, K., Paul, E., 2013. The Microbial Efficiency-Matrix Stabilization (MEMS) framework integrates plant litter decomposition with soil organic matter stabilization: do labile plant inputs form stable soil organic matter? Global Change Biology 19, 988–995. doi:10.1111/gcb.12113

- Cox, P.M., Betts, R.A., Jones, C.D., Spall, S.A., Totterdell, I.J., 2000. Acceleration of global warming due to carbon-cycle feedbacks in a coupled climate model. Nature 408, 184–187. doi:10.1038/35041539
- Cramer, W., Bondeau, A., Woodward, F.I., Prentice, I.C., Betts, R.A., Brovkin, V., Cox, P.M., Fisher, V.,
  Foley, J.A., Friend, A.D., Kucharik, C., Lomas, M.R., Ramankutty, N., Sitch, S., Smith, B., White, A.,
  Young-Molling, C., 2001. Global response of terrestrial ecosystem structure and function to CO2
  and climate change: results from six dynamic global vegetation models. Global Change Biology 7,
  357–373. doi:10.1046/j.1365-2486.2001.00383.x
- Crutzen, P.J., Mosier, A.R., Smith, K.A., Winiwarter, W., 2008. N2O release from agro-biofuel production negates global warming reduction by replacing fossil fuels. Atmospheric Chemistry and Physics 8, 289–395.
- Dalzell, B.J., Johnson, J.M.F., Tallaksen, J., Allan, D.L., Barbour, N.W., 2013. Simulated impacts of crop residue removal and tillage on soil organic matter maintenance. Soil Sci Soc Am J 77, 1349 – 1356. doi:10.2136/sssaj2012.0221
- Davidson, E.A., Janssens, I.A., 2006. Temperature sensitivity of soil carbon decomposition and feedbacks to climate change. Nature 440, 165–173. doi:10.1038/nature04514
- Davidson, E.A., Janssens, I.A., Luo, Y., 2006. On the variability of respiration in terrestrial ecosystems: moving beyond Q10. Global Change Biology 12, 154–164. doi:10.1111/j.1365-2486.2005.01065.x
- Davidson, E.A., Samanta, S., Caramori, S.S., Savage, K., 2012. The Dual Arrhenius and Michaelis–Menten kinetics model for decomposition of soil organic matter at hourly to seasonal time scales. Global Change Biology 18, 371–384. doi:10.1111/j.1365-2486.2011.02546.x
- Davidson, E.A., Savage, K.E., Finzi, A.C., 2014. A big-microsite framework for soil carbon modeling. Global Change Biology 20, 3610–3620. doi:10.1111/gcb.12718

- Davidson, E.A., Trumbore, S.E., Amundson, R., 2000. Biogeochemistry: Soil warming and organic carbon content. Nature 408, 789–790. doi:10.1038/35048672
- Davis, S.C., Parton, W.J., Dohleman, F.G., Smith, C.M., Del Grosso, S., Kent, A.D., DeLucia, E.H., 2010.
   Comparative Biogeochemical Cycles of Bioenergy Crops Reveal Nitrogen-Fixation and Low
   Greenhouse Gas Emissions in a Miscanthus × giganteus Agro-Ecosystem. Ecosystems 13, 144–
   156. doi:10.1007/s10021-009-9306-9
- Davis, S.C., Parton, W.J., Grosso, S.J.D., Keough, C., Marx, E., Adler, P.R., DeLucia, E.H., 2012. Impact of second-generation biofuel agriculture on greenhouse-gas emissions in the corn-growing regions of the US. Frontiers in Ecology and the Environment 10, 69–74. doi:10.1890/110003
- De Figueiredo, E.B., La Scala Jr., N., 2011. Greenhouse gas balance due to the conversion of sugarcane areas from burned to green harvest in Brazil. Agriculture, Ecosystems & Environment 141, 77– 85. doi:16/j.agee.2011.02.014
- Del Grosso, S.J., Halvorson, A.D., Parton, W.J., 2008. Testing DAYCENT model simulations of corn yields and nitrous oxide emissions in irrigated tillage systems in Colorado. Journal of Environment Quality 37, 1383–1389. doi:10.2134/jeq2007.0292
- Del Grosso, S.J., Ogle, S.M., Parton, W.J., Breidt, F.J., 2010. Estimating uncertainty in N2O emissions from U.S. cropland soils. Global Biogeochemical Cycles 24, n/a–n/a. doi:10.1029/2009GB003544
- Del Grosso, S.J., Parton, W.J., Mosier, A.R., Holland, E.A., Pendall, E., Schimel, D.S., Ojima, D.S., 2005. Modeling soil CO2 emissions from ecosystems. Biogeochemistry 73, 71–91. doi:10.1007/s10533-004-0898-z
- De Oliveira, B.G., Carvalho, J.L.N., Cerri, C.E.P., Cerri, C.C., Feigl, B.J., 2013. Soil greenhouse gas fluxes from vinasse application in Brazilian sugarcane areas. Geoderma 200–201, 77–84. doi:10.1016/j.geoderma.2013.02.005

- De Oliveira Bordonal, R., de Figueiredo, E.B., La Scala, N., 2012. Greenhouse gas balance due to the conversion of sugarcane areas from burned to green harvest, considering other conservationist management practices. GCB Bioenergy 4, 846–858. doi:10.1111/j.1757-1707.2012.01193.x
- De Souza Braz, A.M., Fernandes, A.R., Alleoni, L.R.F., 2013. Soil Attributes After the Conversion from Forest to Pasture in Amazon. Land Degradation & Development 24, 33–38. doi:10.1002/ldr.1100
- Dokuchaev, V.V., 1967. Russian chernozem, in: V.V. Dokuchaev, His Selected Works. Israel Program from scientific Translations, Jerusalem.
- Don, A., Kalbitz, K., 2005. Amounts and degradability of dissolved organic carbon from foliar litter at different decomposition stages. Soil Biology and Biochemistry 37, 2171–2179. doi:10.1016/j.soilbio.2005.03.019
- Drake, J.E., Gallet-Budynek, A., Hofmockel, K.S., Bernhardt, E.S., Billings, S.A., Jackson, R.B., Johnsen,
  K.S., Lichter, J., McCarthy, H.R., McCormack, M.L., Moore, D.J.P., Oren, R., Palmroth, S., Phillips,
  R.P., Pippen, J.S., Pritchard, S.G., Treseder, K.K., Schlesinger, W.H., DeLucia, E.H., Finzi, A.C.,
  2011. Increases in the flux of carbon belowground stimulate nitrogen uptake and sustain the
  long-term enhancement of forest productivity under elevated CO. Ecology Letters 14, 349–357.
  doi:10.1111/j.1461-0248.2011.01593.x
- Dungait, J.A.J., Hopkins, D.W., Gregory, A.S., Whitmore, A.P., 2012. Soil organic matter turnover is governed by accessibility not recalcitrance. Global Change Biology 18, 1781–1796. doi:10.1111/j.1365-2486.2012.02665.x
- Duval, B.D., Anderson-Teixeira, K.J., Davis, S.C., Keogh, C., Long, S.P., Parton, W.J., DeLucia, E.H., 2013. Predicting Greenhouse Gas Emissions and Soil Carbon from Changing Pasture to an Energy Crop. PLoS ONE 8, e72019. doi:10.1371/journal.pone.0072019

- Eilers, K.G., Debenport, S., Anderson, S., Fierer, N., 2012. Digging deeper to find unique microbial communities: The strong effect of depth on the structure of bacterial and archaeal communities in soil. Soil Biology and Biochemistry 50, 58–65. doi:10.1016/j.soilbio.2012.03.011
- Elliott, E.T., Paustian, K., Frey, S.D., 1996. Modeling the Measurable or Measuring the Modelable: A Hierarchical Approach to Isolating Meaningful Soil Organic Matter Fractionations, in: Powlson, D.S., Smith, P., Smith, J.U. (Eds.), Evaluation of Soil Organic Matter Models, NATO ASI Series. Springer Berlin Heidelberg, pp. 161–179.
- Erich, M.S., Plante, A.F., Fernández, J.M., Mallory, E.B., Ohno, T., 2012. Effects of profile depth and management on the composition of labile and total soil organic matter. Soil Science Society of America Journal 76, 408. doi:10.2136/sssaj2011.0273
- Esser, G., Kattge, J., Sakalli, A., 2011. Feedback of carbon and nitrogen cycles enhances carbon sequestration in the terrestrial biosphere. Global Change Biology 17, 819–842. doi:10.1111/j.1365-2486.2010.02261.x
- Falloon, P.D., Smith, P., 2000. Modelling refractory soil organic matter. Biology and Fertility of Soils 30, 388–398. doi:10.1007/s003740050019
- Fang, C., Smith, P., Moncrieff, J.B., Smith, J.U., 2005. Similar response of labile and resistant soil organic matter pools to changes in temperature. Nature 433, 57–59. doi:10.1038/nature03138
- Farquhar, G.D., Caemmerer, S. von, Berry, J.A., 1980. A biochemical model of photosynthetic CO2 assimilation in leaves of C3 species. Planta 149, 78–90. doi:10.1007/BF00386231
- Fassbinder, J.J., Schultz, N.M., Baker, J.M., Griffis, T.J., 2013. Automated, low-power chamber system for measuring nitrous oxide emissions. Journal of Environmental Quality 42, 1–9. doi:10.2134/jeq2012.0283

- Fidelis, A., Lyra, M.F. di S., Pivello, V.R., 2013. Above- and below-ground biomass and carbon dynamics in Brazilian Cerrado wet grasslands. Journal of Vegetation Science 24, 356–364. doi:10.1111/j.1654-1103.2012.01465.x
- Fierer, N., Allen, A.S., Schimel, J.P., Holden, P.A., 2003a. Controls on microbial CO2 production: a comparison of surface and subsurface soil horizons. Global Change Biology 9, 1322–1332. doi:10.1046/j.1365-2486.2003.00663.x
- Fierer, N., Craine, J.M., McLauchlan, K., Schimel, J.P., 2005. Litter quality and the temperature sensitivity of decomposition. Ecology 86, 320–326. doi:10.1890/04-1254
- Fierer, N., Schimel, J.P., Holden, P.A., 2003b. Variations in microbial community composition through two soil depth profiles. Soil Biology and Biochemistry 35, 167–176. doi:10.1016/S0038-0717(02)00251-1
- Fierer, N., Strickland, M.S., Liptzin, D., Bradford, M.A., Cleveland, C.C., 2009. Global patterns in belowground communities. Ecology Letters 12, 1238–1249. doi:10.1111/j.1461-0248.2009.01360.x
- Figueiredo, C.C., Resck, D.V.S., Carneiro, M.A.C., Ramos, M.L.G., Sá, J.C.M., 2013. Stratification ratio of organic matter pools influenced by management systems in a weathered Oxisol from a tropical agro-ecoregion in Brazil. Soil Research 51, 133–141.
- Follett, R., Vogel, K., Varvel, G., Mitchell, R., Kimble, J., 2012. Soil carbon sequestration by switchgrass and no-till maize grown for bioenergy. BioEnergy Research 866–875. doi:10.1007/s12155-012-9198-y
- Fontaine, S., Barot, S., Barre, P., Bdioui, N., Mary, B., Rumpel, C., 2007. Stability of organic carbon in deep soil layers controlled by fresh carbon supply. Nature 450, 277–280. doi:10.1038/nature06275

- Forster, P., 2007. Changes in atmospheric constituents and in radiative forcing. In: Climate Change 2007: The Physical Science Basis. Contributions of Working Group I to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change [WWW Document].
- Fox, J., Weisberg, S., 2011. An {R} companion to applied regression, second edition. Thousand Oaks CA: Sage.
- Franco, H.C.J., Trivelin, P.C.O., Carlos Eduardo, F., Vitti, A.C., Otto, R., 2010. Stalk yield and technological attributes of planted cane as related to nitrogen fertilization. Scientia Agricola 67, 579–590. doi:10.1590/S0103-90162010000500012
- Frey, S.D., Lee, J., Melillo, J.M., Six, J., 2013. The temperature response of soil microbial efficiency and its feedback to climate. Nature Climate Change 3, 395–398. doi:10.1038/nclimate1796
- Friedlingstein, P., Bopp, L., Ciais, P., Dufresne, J.-L., Fairhead, L., LeTreut, H., Monfray, P., Orr, J., 2001.
   Positive feedback between future climate change and the carbon cycle. Geophysical Research
   Letters 28, 1543–1546. doi:10.1029/2000GL012015
- Friedlingstein, P., Cox, P., Betts, R., Bopp, L., al, et, 2006. Climate-Carbon Cycle Feedback Analysis: Results from the C^sup 4^MIP Model Intercomparison. Journal of Climate 19, 3337–3338,3340– 3353.
- Friedlingstein, P., Prentice, I., 2010. Carbon–climate feedbacks: a review of model and observation
  based estimates. Current Opinion in Environmental Sustainability 2, 251–257.
  doi:10.1016/j.cosust.2010.06.002
- Fröberg, M., Hanson, P.J., Trumbore, S.E., Swanston, C.W., Todd, D.E., 2009. Flux of carbon from 14Cenriched leaf litter throughout a forest soil mesocosm. Geoderma 149, 181–188. doi:10.1016/j.geoderma.2008.11.029

- Fronning, B.E., Thelen, K.D., Min, D.-H., 2008. Use of manure, compost, and cover crops to supplant crop residue carbon in corn stover removed cropping systems. Agronomy Journal 100, 1703–1710. doi:10.2134/agronj2008.0052
- Fujita, Y., Witte, J.-P.M., van Bodegom, P.M., 2014. Incorporating microbial ecology concepts into global soil mineralization models to improve predictions of carbon and nitrogen fluxes. Global Biogeochemical Cycles 28, 2013GB004595. doi:10.1002/2013GB004595
- Fuka, D.R., Walter, M.T., MacAlister, C., Degaetano, A.T., Steenhuis, T.S., Easton, Z.M., 2014. Using the Climate Forecast System Reanalysis as weather input data for watershed models. Hydrological Processes 28, 5613–5623. doi:10.1002/hyp.10073
- Galdos, M.V., Cerri, C.C., Cerri, C.E.P., 2009. Soil carbon stocks under burned and unburned sugarcane in Brazil. Geoderma 153, 347–352. doi:16/j.geoderma.2009.08.025
- Galdos, M.V., Cerri, C.C., Cerri, C.E.P., Paustian, K., Van Antwerpen, R., 2009. Simulation of soil carbon dynamics under sugarcane with the CENTURY model. Soil Science Society of America Journal 73, 802–811. doi:10.2136/sssaj2007.0285
- Galdos, M.V., Cerri, C.E., Cerri, C.C., Paustian, K., van Antwerpen, R., 2009. Simulation of sugarcane residue decomposition and aboveground growth. Plant and Soil 326, 243–259. doi:http://dx.doi.org/10.1007/s11104-009-0004-3
- Gao, J., Thelen, K.D., Hao, X., 2013. Life cycle analysis of corn harvest strategies for bioethanol production. Agronomy Journal 105, 705–712. doi:10.2134/agronj2012.0420
- Gava, G.J. de C., Trivelin, P.C.O., Vitti, A.C., Oliveira, M.W. de, 2005. Urea and sugarcane straw nitrogen balance in a soil-sugarcane crop system. Pesquisa Agropecuária Brasileira 40, 689–695. doi:10.1590/S0100-204X2005000700010
- Gelman, A., Rubin, D.B., 1992. Inference from Iterative Simulation Using Multiple Sequences. Statistical Science 7, 457–472. doi:10.1214/ss/1177011136

- Geremia, C., White, P.J., Hoeting, J.A., Wallen, R.L., Watson, F.G.R., Blanton, D., Hobbs, N.T., 2014. Integrating population- and individual-level information in a movement model of Yellowstone bison. Ecological Applications 24, 346–362. doi:10.1890/13-0137.1
- Geweke, J., 1992. Evaluating the accuracy of sampling-based approaches to calculating posterior moments, in: Bernado, J.M., Berger, J.O., Dawid, A.P., Smith, A.F.M. (Eds.), Bayesian Statistics 4. Clarendon Press, Oxford, UK.
- Giardina, C.P., Litton, C.M., Crow, S.E., Asner, G.P., 2014. Warming-related increases in soil CO2 efflux are explained by increased below-ground carbon flux. Nature Climate Change 4, 822–827. doi:10.1038/nclimate2322
- Giardina, C.P., Ryan, M.G., 2000. Evidence that decomposition rates of organic carbon in mineral soil do not vary with temperature. Nature 404, 858–861. doi:10.1038/35009076
- Gleixner, G., 2013. Soil organic matter dynamics: a biological perspective derived from the use of compound-specific isotopes studies. Ecological Research 28, 683–695. doi:10.1007/s11284-012-1022-9
- Golchin, A., Oades, J., Skjemstad, J., Clarke, P., 1994. Soil structure and carbon cycling. Soil Research 32, 1043–1068.
- Goldemberg, J., Mello, F.F.C., Cerri, C.E.P., Davies, C.A., Cerri, C.C., 2014. Meeting the global demand for biofuels in 2021 through sustainable land use change policy. Energy Policy 69, 14–18. doi:10.1016/j.enpol.2014.02.008
- Gollany, H.T., Novak, J.M., Liang, Y., Albrecht, S.L., Rickman, R.W., Follett, R.F., Wilhelm, W.W., Hunt, P.G., 2010. Simulating soil organic carbon dynamics with residue removal using the CQESTR model. Soil Sci Soc Am J 74, 372–383. doi:10.2136/sssaj2009.0086
- Grant, R.F., Oechel, W.C., Ping, C.-L., 2003. Modelling carbon balances of coastal arctic tundra under changing climate. Global Change Biology 9, 16–36. doi:10.1046/j.1365-2486.2003.00549.x

- Guenet, B., Eglin, T., Vasilyeva, N., Peylin, P., Ciais, P., Chenu, C., 2013. The relative importance of decomposition and transport mechanisms in accounting for soil organic carbon profiles.
   Biogeosciences 10, 2379–2392. doi:10.5194/bg-10-2379-2013
- Gulde, S., Chung, H., Amelung, W., Chang, C., Six, J., 2008. Soil Carbon Saturation Controls Labile and Stable Carbon Pool Dynamics. Soil Science Society of America Journal 72, 605. doi:10.2136/sssaj2007.0251
- Hagedorn, F., Kammer, A., Schmidt, M.W.I., Goodale, C.L., 2012. Nitrogen addition alters mineralization dynamics of 13C-depleted leaf and twig litter and reduces leaching of older DOC from mineral soil. Global Change Biology 18, 1412–1427. doi:10.1111/j.1365-2486.2011.02603.x
- Hagerty, S.B., van Groenigen, K.J., Allison, S.D., Hungate, B.A., Schwartz, E., Koch, G.W., Kolka, R.K., Dijkstra, P., 2014. Accelerated microbial turnover but constant growth efficiency with warming in soil. Nature Climate Change 4, 903–906. doi:10.1038/nclimate2361
- Halvorson, A.D., Jantalia, C.P., 2011. Nitrogen fertilization effects on irrigated no-till corn production and soil carbon and nitrogen. Agronomy Journal 103, 1423–1431. doi:10.2134/agronj2011.0102
- Hamdi, S., Moyano, F., Sall, S., Bernoux, M., Chevallier, T., 2013. Synthesis analysis of the temperature sensitivity of soil respiration from laboratory studies in relation to incubation methods and soil conditions. Soil Biology and Biochemistry 58, 115–126. doi:10.1016/j.soilbio.2012.11.012
- Hammerbeck, A.L., Stetson, S.J., Osborne, S.L., Schumacher, T.E., Pikul, J.L., 2012. Corn residue removal impact on soil aggregates in a no-till corn/soybean rotation. Soil Science Society of America Journal 76, 1390 1398.
- Hararuk, O., Luo, Y., 2014. Improvement of global litter turnover rate predictions using a Bayesian MCMC approach. Ecosphere 5, 1–13.
- Harper, R.J., Tibbett, M., 2013. The hidden organic carbon in deep mineral soils. Plant and Soil 368, 641– 648. doi:10.1007/s11104-013-1600-9

Harrell Jr., F.E., 2012. Hmisc: Harrell Miscellaneous.

- Hartley, I.P., Ineson, P., 2008. Substrate quality and the temperature sensitivity of soil organic matter decomposition. Soil Biology and Biochemistry 40, 1567–1574. doi:10.1016/j.soilbio.2008.01.007
- Hassink, J., Whitmore, A.P., 1997. A model of the physical protection of organic matter in soils. Soil Sci Soc Am J 61, 131–139.
- Hassuani, S.J., Leal, M.R.L.V., de Carvalho Macedo, I. (Eds.), 2005. Biomass power generation: sugar cane bagasse and trash, 1st ed. Programa das Nações Unidas para o Desenvolvimento, Centro de Tecnologie Canavieira, Piracicaba, Brazil.
- Hedges, J.I., Eglinton, G., Hatcher, P.G., Kirchman, D.L., Arnosti, C., Derenne, S., Evershed, R.P., Kögel-Knabner, I., de Leeuw, J.W., Littke, R., Michaelis, W., Rullkötter, J., 2000. The molecularlyuncharacterized component of nonliving organic matter in natural environments. Organic Geochemistry 31, 945–958. doi:10.1016/S0146-6380(00)00096-6
- Heimann, M., Reichstein, M., 2008. Terrestrial ecosystem carbon dynamics and climate feedbacks. Nature 451, 289–292. doi:10.1038/nature06591
- Heitkamp, F., Wendland, M., Offenberger, K., Gerold, G., 2012. Implications of input estimation, residue quality and carbon saturation on the predictive power of the Rothamsted Carbon Model. Geoderma 170, 168–175. doi:10.1016/j.geoderma.2011.11.005
- Henderson-Sellers, A., 2006. Improving land-surface parameterization schemes using stable water isotopes: Introducing the "iPILPS" initiative. Global and Planetary Change, Isotopes in the Project for Intercomparison of Landsurface Parameterisation Schemes (iPILPS) 51, 3–24.

doi:10.1016/j.gloplacha.2005.12.009

Hénin, S., Depuis, M., 1945. Essai de bilan de la materie organique du sol. Annales Agronomiques 15, 17–29.

- Herman, J., Moorhead, D., Berg, B., 2008. The relationship between rates of lignin and cellulose decay in aboveground forest litter. Soil Biology and Biochemistry 40, 2620–2626.
  doi:10.1016/j.soilbio.2008.07.003
- He, Z., Xu, M., Deng, Y., Kang, S., Kellogg, L., Wu, L., Van Nostrand, J.D., Hobbie, S.E., Reich, P.B., Zhou, J.,
   2010. Metagenomic analysis reveals a marked divergence in the structure of belowground
   microbial communities at elevated CO2. Ecology Letters 13, 564–575. doi:10.1111/j.1461 0248.2010.01453.x
- Hilgard, E.W., 1906. Soils, their formation, properties, com- positions, and relations to climate and plant growth in the humid and arid regions. The Macmilan Company, London, England.
- Hobbs, N.T., Hooten, M.B., in press. Simple Bayesian Models, in: Bayesian Models: A Statistical Primer for Ecologists. Princeton University Press, Princeton, NJ, pp. 79–106.
- Hooker, B.A., Morris, T.F., Peters, R., Cardon, Z.G., 2005. Long-term effects of tillage and corn stalk return on soil carbon dynamics. Soil Science Society of America Journal 69, 188–196. doi:10.2136/sssaj2005.0188
- Houghton, R.A., 2007. Balancing the Global Carbon Budget. Annual Review of Earth and Planetary Sciences 35, 313–347. doi:10.1146/annurev.earth.35.031306.140057
- Huggins, D.R., Karow, R.S., Collins, H.P., Ransom, J.K., 2011. Introduction: evaluating long-term impacts of harvesting crop residues on soil quality. Agronomy Journal 103, 230–233.
   doi:10.2134/agronj2010.0382s
- Hunt, H.W., 1977. A Simulation Model for Decomposition in Grasslands. Ecology 58, 469–484. doi:10.2307/1938998

Innis, G.S. (Ed.), 1978. Grassland simulation model., Ecological Studies. Springer-Verlag, New York, NY.

- Ise, T., Moorcroft, P.R., 2006. The global-scale temperature and moisture dependencies of soil organic carbon decomposition: an analysis using a mechanistic decomposition model. Biogeochemistry 80, 217–231. doi:10.1007/s10533-006-9019-5
- Iversen, C.M., 2010. Digging deeper: fine-root responses to rising atmospheric CO2 concentration in forested ecosystems. New Phytologist 186, 346–357. doi:10.1111/j.1469-8137.2009.03122.x
- Iversen, C.M., Keller, J.K., Garten, C.T., Norby, R.J., 2012. Soil carbon and nitrogen cycling and storage throughout the soil profile in a sweetgum plantation after 11 years of CO2-enrichment. Global Change Biology 18, 1684–1697. doi:10.1111/j.1365-2486.2012.02643.x
- Janssens, I.A., Lankreijer, H., Matteucci, G., Kowalski, A.S., Buchmann, N., Epron, D., Pilegaard, K.,
  Kutsch, W., Longdoz, B., Grünwald, T., Montagnani, L., Dore, S., Rebmann, C., Moors, E.J., Grelle,
  A., Rannik, Ü., Morgenstern, K., Oltchev, S., Clement, R., Guðmundsson, J., Minerbi, S., Berbigier,
  P., Ibrom, A., Moncrieff, J., Aubinet, M., Bernhofer, C., Jensen, N.O., Vesala, T., Granier, A.,
  Schulze, E.-D., Lindroth, A., Dolman, A.J., Jarvis, P.G., Ceulemans, R., Valentini, R., 2001.
  Productivity overshadows temperature in determining soil and ecosystem respiration across
  European forests. Global Change Biology 7, 269–278. doi:10.1046/j.1365-2486.2001.00412.x
- Jansson, S.L., 1958. Tracer Studies on Nitrogen Transformations in Soil with Special Attention to Mineralisation-immobilization Relationships. tr. Almqvist & Wiksell.
- Jastrow, J.D., 1996. Soil aggregate formation and the accrual of particulate and mineral-associated organic matter. Soil Biology and Biochemistry 28, 665–676. doi:10.1016/0038-0717(95)00159-X
- Jenkinson, D.S., Adams, D.E., Wild, A., 1991. Model estimates of CO2 emissions from soil in response to global warming. Nature 351, 304–306. doi:10.1038/351304a0
- Jenkinson, D.S., Coleman, K., 2008. The turnover of organic carbon in subsoils. Part 2. Modelling carbon turnover. European Journal of Soil Science 59, 400–413. doi:10.1111/j.1365-2389.2008.01026.x

- Jenkinson, D.S., Hart, P.B.S., Reyner, J.H., Parry, L.C., 1987. Modeling the turnover of organic matter in long-term experiments at Rothamsted. (INTECOL Bulletin No. 15).
- Jenkinson, D.S., Rayner, J.H., 1977. The turnover of soil organic matter in some of the Rothamsted classical experiments. Soil Sci Soc Am J 123, 298–305.
- Jenny, H., 1941. Factors of Soil Formation: A System of Pedology. McGraw-Hill book Company, Incorporated.
- Jenny, H., Gessel, S.P., Bingham, F.T., 1949. Comparative Study of Decomposition Rates of Organic Matter in Temperate and Tropical Regions. Soil Science 68, 419–432. doi:10.1097/00010694-200606001-00017
- Jobbagy, E.G., Jackson, R.B., 2000. The vertical distribution of soil organic carbon and its relation to climate and vegetation. Ecological Applications 10, 423–436.
- Johnson, J.M.F., Wilhelm, W.W., Karlen, D.L., Archer, D.W., Wienhold, B., Lightle, D.T., Laird, D., Baker, J., Ochsner, T.E., Novak, J.M., Halvorson, A.D., Arriaga, F., Barbour, N., 2010. Nutrient removal as a function of corn stover cutting height and cob harvest. BioEnergy Research 3, 342–352. doi:10.1007/s12155-010-9093-3
- Jones, C.D., Cox, P., Huntingford, C., 2003. Uncertainty in climate–carbon-cycle projections associated with the sensitivity of soil respiration to temperature. Tellus B 55, 642–648. doi:10.1034/j.1600-0889.2003.01440.x
- Kaiser, K., Kalbitz, K., 2012. Cycling downwards dissolved organic matter in soils. Soil Biology and Biochemistry 52, 29–32. doi:10.1016/j.soilbio.2012.04.002
- Kalbitz, K., Solinger, S., Park, J.-H., Michalzik, B., Matzner, E., 2000. Controls on the dynamics of dissolved organic matter in soils: a review. Soil Science 165, 277–304.
- Karlen, D.L., 2010. Corn stover feedstock trials to support predictive modeling. GCB Bioenergy 2, 235– 247. doi:10.1111/j.1757-1707.2010.01061.x

- Karlen, D.L., Birell, S.J., Hess, J.R., 2011a. A five-year assessment of corn stover harvest in central Iowa, USA. Soil & Tillage Research 115-116, 47–55.
- Karlen, D.L., Varvel, G.E., Johnson, J.M.F., Baker, J.M., Osborne, S.L., Novak, J.M., Adler, P.R., Roth, G.W., Birrell, S.J., 2011b. Monitoring soil quality to assess the sustainability of harvesting corn stover. Agronomy Journal 103, -295. doi:10.2134/agronj2010.0160s
- Keenan, T.F., Carbone, M.S., Reichstein, M., Richardson, A.D., 2011. The model–data fusion pitfall: assuming certainty in an uncertain world. Oecologia 167, 587–597. doi:10.1007/s00442-011-2106-x
- Kelleher, B.P., Simpson, A.J., 2006. Humic substances in soils: are they really chemically distinct? Environ. Sci. Technol. 40, 4605–4611.
- Kimetu, J.M., Lehmann, J., Kinyangi, J.M., Cheng, C.H., Thies, J., Mugendi, D.N., Pell, A., 2009. Soil organic
   C stabilization and thresholds in C saturation. Soil Biology and Biochemistry 41, 2100–2104.
   doi:10.1016/j.soilbio.2009.07.022
- Kirkham, D., Bartholomew, W.V., 1954. Equations for Following Nutrient Transformations in Soil, Utilizing Tracer Data1. Soil Science Society of America Journal 18, 33.

doi:10.2136/sssaj1954.03615995001800010009x

- Kirschbaum, M.U.F., 2000. Will changes in soil organic carbon act as a positive or negative feedback on global warming? Biogeochemistry 48, 21–51.
- Kirschbaum, M.U.F., 1995. The temperature dependence of soil organic matter decomposition, and the effect of global warming on soil organic C storage. Soil Biology and Biochemistry 27, 753–760. doi:10.1016/0038-0717(94)00242-S

Kleber, M., 2010. What is recalcitrant soil organic matter? Environmental Chemistry 7, 320–332.

Kleber, M., Nico, P.S., Plante, A., Filley, T., Kramer, M., Swanston, C., Sollins, P., 2011. Old and stable soil organic matter is not necessarily chemically recalcitrant: implications for modeling concepts and

temperature sensitivity. Global Change Biology 17, 1097–1107. doi:10.1111/j.1365-2486.2010.02278.x

Kleber, M., Sollins, P., Sutton, R., 2007. A conceptual model of organo-mineral interactions in soils: self-assembly of organic molecular fragments into zonal structures on mineral surfaces.
 Biogeochemistry 85, 9–24. doi:10.1007/s10533-007-9103-5

Koarashi, J., Hockaday, W.C., Masiello, C.A., Trumbore, S.E., 2012. Dynamics of decadally cycling carbon in subsurface soils. Journal of Geophysical Research: Biogeosciences 117, G03033. doi:10.1029/2012JG002034

- Koven, C.D., Riley, W.J., Subin, Z.M., Tang, J.Y., Torn, M.S., Collins, W.D., Bonan, G.B., Lawrence, D.M.,
   Swenson, S.C., 2013. The effect of vertically resolved soil biogeochemistry and alternate soil C
   and N models on C dynamics of CLM4. Biogeosciences 10, 7109–7131. doi:10.5194/bg-10-7109-2013
- Koven, C.D., Ringeval, B., Friedlingstein, P., Ciais, P., Cadule, P., Khvorostyanov, D., Krinner, G., Tarnocai,
   C., 2011. Permafrost carbon-climate feedbacks accelerate global warming. Proceedings of the
   National Academy of Sciences 108, 14769–14774. doi:10.1073/pnas.1103910108
- Kramer, C., Gleixner, G., 2008. Soil organic matter in soil depth profiles: Distinct carbon preferences of microbial groups during carbon transformation. Soil Biology and Biochemistry 40, 425–433. doi:10.1016/j.soilbio.2007.09.016
- Krinner, G., Viovy, N., de Noblet-Ducoudré, N., Ogée, J., Polcher, J., Friedlingstein, P., Ciais, P., Sitch, S., Prentice, I.C., 2005. A dynamic global vegetation model for studies of the coupled atmospherebiosphere system. Global Biogeochemical Cycles 19, GB1015. doi:10.1029/2003GB002199
- Kuzyakov, Y., Domanski, G., 2000. Carbon input by plants into the soil. Review. Journal of Plant Nutrition and Soil Science 163, 421–431. doi:10.1002/1522-2624(200008)163:4<421::AID-JPLN421>3.0.CO;2-R

- Kuzyakov, Y., Friedel, J.K., Stahr, K., 2000. Review of mechanisms and quantification of priming effects. Soil Biology and Biochemistry 32, 1485–1498. doi:10.1016/S0038-0717(00)00084-5
- Langley, J.A., McKinley, D.C., Wolf, A.A., Hungate, B.A., Drake, B.G., Megonigal, J.P., 2009. Priming depletes soil carbon and releases nitrogen in a scrub-oak ecosystem exposed to elevated CO2. Soil Biology and Biochemistry 41, 54–60. doi:10.1016/j.soilbio.2008.09.016
- Lawrence, C.R., Neff, J.C., Schimel, J.P., 2009. Does adding microbial mechanisms of decomposition improve soil organic matter models? A comparison of four models using data from a pulsed rewetting experiment. Soil Biology and Biochemistry 41, 1923–1934. doi:10.1016/j.soilbio.2009.06.016
- Lawrence, D.M., Oleson, K.W., Flanner, M.G., Thornton, P.E., Swenson, S.C., Lawrence, P.J., Zeng, X., Yang, Z.-L., Levis, S., Sakaguchi, K., Bonan, G.B., Slater, A.G., 2011. Parameterization improvements and functional and structural advances in Version 4 of the Community Land Model. Journal of Advances in Modeling Earth Systems 3, M03001. doi:10.1029/2011MS000045
- Leavitt, S.W., Follett, R.F., Paul, E.A., 2006. Estimation of slow- and fast-cycling soil organic carbon pools from 6N HCl hydrolysis. Radiocarbon 38, 231–239. doi:10.2458/azu\_js\_rc.38.1903
- Lee, K., 2004. Life cycle assessment, best practices of ISO 14040 series [WWW Document].
- Lefèvre, R., Barré, P., Moyano, F.E., Christensen, B.T., Bardoux, G., Eglin, T., Girardin, C., Houot, S., Kätterer, T., van Oort, F., Chenu, C., 2014. Higher temperature sensitivity for stable than for labile soil organic carbon – Evidence from incubations of long-term bare fallow soils. Global Change Biology 20, 633–640. doi:10.1111/gcb.12402
- Lehmann, J., Solomon, D., Kinyangi, J., Dathe, L., Wirick, S., Jacobsen, C., 2008. Spatial complexity of soil organic matter forms at nanometre scales. Nature Geoscience 1, 238–242. doi:10.1038/ngeo155
- Leifeld, J., Fuhrer, J., 2005. The Temperature Response of CO₂ Production from Bulk Soils and Soil Fractions Is Related to Soil Organic Matter Quality. Biogeochemistry 75, 433–453.
- Lekkerkerk, L., Lundkvist, H., Ågren, G.I., Ekbohm, G., Bosatta, E., 1990. Decomposition of heterogeneous substrates; An experimental investigation of a hypothesis on substrate and microbial properties. Soil Biology and Biochemistry 22, 161–167. doi:10.1016/0038-0717(90)90081-A
- Levins, R., 1966. The strategy of model building in population biology. American Scientist 54, 421–431.
- Linden, D.R., Clapp, C.E., Dowdy, R.H., 2000. Long-term corn grain and stover yields as a function of tillage and residue removal in east central Minnesota. Soil & Tillage Research 56, 167–174. doi:10.1016/S0167-1987(00)00139-2
- Liski, J., Mäkelä, A., Johan, C., 1999. CO₂ Emissions from Soil in Response to Climatic Warming Are Overestimated: The Decomposition of Old Soil Organic Matter Is Tolerant of Temperature. AMBIO 28, 171–174.
- Lloyd, J., Taylor, J.A., 1994. On the temperature dependence of soil respiration. Functional Ecology 8, 315–323.
- Lopez-Sangil, L., Rovira, P., Casals, P., 2013. Decay and vertical reallocation of organic C, and its incorporation into carbonates, in agricultural soil horizons at two different depths and rewetting frequencies. Soil Biology and Biochemistry 61, 33–44. doi:10.1016/j.soilbio.2013.02.008
- Luo, Y., Ogle, K., Tucker, C., Fei, S., Gao, C., LaDeau, S., Clark, J.S., Schimel, D.S., 2011. Ecological forecasting and data assimilation in a data-rich era. Ecological Applications 21, 1429–1442. doi:10.1890/09-1275.1
- Luo, Y.Q., Randerson, J.T., Abramowitz, G., Bacour, C., Blyth, E., Carvalhais, N., Ciais, P., Dalmonech, D.,
  Fisher, J.B., Fisher, R., Friedlingstein, P., Hibbard, K., Hoffman, F., Huntzinger, D., Jones, C.D.,
  Koven, C., Lawrence, D., Li, D.J., Mahecha, M., Niu, S.L., Norby, R., Piao, S.L., Qi, X., Peylin, P.,
  Prentice, I.C., Riley, W., Reichstein, M., Schwalm, C., Wang, Y.P., Xia, J.Y., Zaehle, S., Zhou, X.H.,
  2012. A framework for benchmarking land models. Biogeosciences 9, 3857–3874.
  doi:10.5194/bg-9-3857-2012

- Luo, Y., Wan, S., Hui, D., Wallace, L.L., 2001. Acclimatization of soil respiration to warming in a tall grass prairie. Nature 413, 622–625. doi:10.1038/35098065
- Lützow, M. v., Kögel-Knabner, I., Ekschmitt, K., Matzner, E., Guggenberger, G., Marschner, B., Flessa, H., 2006. Stabilization of organic matter in temperate soils: mechanisms and their relevance under different soil conditions – a review. European Journal of Soil Science 57, 426–445. doi:10.1111/j.1365-2389.2006.00809.x
- Lynch, J.M., von Lampe, M., 2011. The need for bioenergy policy analysis. Biomass and Bioenergy 35, 2311–2314. doi:10.1016/j.biombioe.2009.10.015
- Maeght, J.-L., Rewald, B., Pierret, A., 2013. How to study deep roots—and why it matters. Functional Plant Ecology 4, 299. doi:10.3389/fpls.2013.00299
- Magill, A.H., Aber, J.D., 2000. Dissolved organic carbon and nitrogen relationships in forest litter as affected by nitrogen deposition. Soil Biology and Biochemistry 32, 603–613. doi:10.1016/S0038-0717(99)00187-X
- Mahecha, M.D., Reichstein, M., Carvalhais, N., Lasslop, G., Lange, H., Seneviratne, S.I., Vargas, R.,
  Ammann, C., Arain, M.A., Cescatti, A., Janssens, I.A., Migliavacca, M., Montagnani, L.,
  Richardson, A.D., 2010. Global Convergence in the Temperature Sensitivity of Respiration at
  Ecosystem Level. Science 329, 838–840. doi:10.1126/science.1189587
- Malamoud, K., McBratney, A.B., Minasny, B., Field, D.J., 2009. Modelling how carbon affects soil structure. Geoderma 149, 19–26. doi:10.1016/j.geoderma.2008.10.018
- Manlay, R.J., Feller, C., Swift, M.J., 2007. Historical evolution of soil organic matter concepts and their relationships with the fertility and sustainability of cropping systems. Agriculture, Ecosystems & Environment 119, 217–233. doi:10.1016/j.agee.2006.07.011
- Manzoni, S., Jackson, R.B., Trofymow, J.A., Porporato, A., 2008. The Global Stoichiometry of Litter Nitrogen Mineralization. Science 321, 684–686. doi:10.1126/science.1159792

- Manzoni, S., Piñeiro, G., Jackson, R.B., Jobbágy, E.G., Kim, J.H., Porporato, A., 2012. Analytical models of soil and litter decomposition: Solutions for mass loss and time-dependent decay rates. Soil Biology and Biochemistry 50, 66–76. doi:10.1016/j.soilbio.2012.02.029
- Manzoni, S., Porporato, A., 2009. Soil carbon and nitrogen mineralization: Theory and models across scales. Soil Biology and Biochemistry 41, 1355–1379. doi:10.1016/j.soilbio.2009.02.031
- Maskina, M.S., Power, J.F., Doran, J.W., Wilhelm, W.W., 1993. Residual effects of no-till crop residues on corn yield and nitrogen uptake. Soil Sci Soc Am J 57, 1555–1560.
- Matus, F., Rumpel, C., Neculman, R., Panichini, M., Mora, M.L., 2014. Soil carbon storage and stabilisation in andic soils: A review. CATENA 120, 102–110. doi:10.1016/j.catena.2014.04.008
- McGill, W.B., Hunt, H.W., Woodmansee, R.G., Reuss, J.O., 1981. Phoenix, a model of the dynamics of carbon and nitrogen in grassland soils, in: Terrestrial Nitrogen Cycles, Ecological Bulletins. Swedish Natural Science Research Council, Stockholm, Sweden, pp. 49–115.
- McGuire, K.L., Treseder, K.K., 2010. Microbial communities and their relevance for ecosystem models: Decomposition as a case study. Soil Biology and Biochemistry 42, 529–535. doi:10.1016/j.soilbio.2009.11.016
- Meehl, G.A., Washington, W.M., Arblaster, J.M., Hu, A., 2004. Factors Affecting Climate Sensitivity in Global Coupled Models. Journal of Climate 17, 1584–1596. doi:10.1175/1520-0442(2004)017<1584:FACSIG>2.0.CO;2
- Melillo, J.M., Steudler, P.A., Aber, J.D., Newkirk, K., Lux, H., Bowles, F.P., Catricala, C., Magill, A., Ahrens,
   T., Morrisseau, S., 2002. Soil Warming and Carbon-Cycle Feedbacks to the Climate System.
   Science 298, 2173–2176. doi:10.1126/science.1074153
- Mello, F.F.C., Cerri, C.E.P., Davies, C.A., Holbrook, N.M., Paustian, K., Maia, S.M.F., Galdos, M.V., Bernoux, M., Cerri, C.C., 2014. Payback time for soil carbon and sugar-cane ethanol. Nature Climate Change 4, 605–609. doi:10.1038/nclimate2239

- Miltner, A., Bombach, P., Schmidt-Brücken, B., Kästner, M., 2011. SOM genesis: microbial biomass as a significant source. Biogeochemistry 111, 41–55. doi:10.1007/s10533-011-9658-z
- Moebius-Clune, B.N., van Es, H.M., Idowu, O.J., Schindelbeck, R.R., Moebius-Clune, D.J., Wolfe, D.W., Abawi, G.S., Thies, J.E., Gugino, B.K., Lucey, R., 2008. Long-term effects of harvesting maize stover and tillage on soil quality. Soil Science Society of America Journal 72, 960–969. doi:10.2136/sssaj2007.0248
- Moorhead, D.L., Lashermes, G., Sinsabaugh, R.L., Weintraub, M.N., 2013. Calculating co-metabolic costs of lignin decay and their impacts on carbon use efficiency. Soil Biology & Biochemistry 66, 17– 19.
- Moorhead, D.L., Sinsabaugh, R.L., 2006. A theoretical model of litter decay and microbial interaction. Ecological Monographs 76, 151–174. doi:10.1890/0012-9615(2006)076[0151:ATMOLD]2.0.CO;2
- Moss, R.H., Edmonds, J.A., Hibbard, K.A., Manning, M.R., Rose, S.K., van Vuuren, D.P., Carter, T.R., Emori,
  S., Kainuma, M., Kram, T., Meehl, G.A., Mitchell, J.F.B., Nakicenovic, N., Riahi, K., Smith, S.J.,
  Stouffer, R.J., Thomson, A.M., Weyant, J.P., Wilbanks, T.J., 2010. The next generation of
  scenarios for climate change research and assessment. Nature 463, 747–756.
  doi:10.1038/nature08823
- Muth, D.J., Bryden, K.M., 2013. An integrated model for assessment of sustainable agricultural residue removal limits for bioenergy systems. Environmental Modelling & Software 39, 50–69. doi:10.1016/j.envsoft.2012.04.006

Oades, J.M., 1988. The retention of organic matter in soils. Biogeochemistry 5, 35–70.

Oades, J.M., 1984. Soil organic matter and structural stability: mechanisms and implications for management. Plant and Soil 76, 319–337. doi:10.1007/BF02205590

- Ogle, K., Pendall, E., 2015. Isotope partitioning of soil respiration: A Bayesian solution to accommodate multiple sources of variability. Journal of Geophysical Research: Biogeosciences 2014JG002794. doi:10.1002/2014JG002794
- Ogle, S.M., Breidt, F.J., Easter, M., Williams, S., Killian, K., Paustian, K., 2010. Scale and uncertainty in modeled soil organic carbon stock changes for US croplands using a process-based model. Global Change Biology 16, 810–822. doi:10.1111/j.1365-2486.2009.01951.x
- Ogle, S.M., Breidt, F.J., Easter, M., Williams, S., Paustian, K., 2007. An empirically based approach for estimating uncertainty associated with modelling carbon sequestration in soils. Ecological Modelling 205, 453–463. doi:10.1016/j.ecolmodel.2007.03.007
- Olson, J.S., 1963. Energy Storage and the Balance of Producers and Decomposers in Ecological Systems. Ecology 44, 322–331. doi:10.2307/1932179
- Oren, A., Chefetz, B., 2012. Successive sorption–desorption cycles of dissolved organic matter in mineral soil matrices. Geoderma 189–190, 108–115. doi:10.1016/j.geoderma.2012.05.004
- Ostle, N.J., Smith, P., Fisher, R., Ian Woodward, F., Fisher, J.B., Smith, J.U., Galbraith, D., Levy, P., Meir, P., McNamara, N.P., Bardgett, R.D., 2009. Integrating plant–soil interactions into global carbon cycle models. Journal of Ecology 97, 851–863. doi:10.1111/j.1365-2745.2009.01547.x
- Ota, M., Nagai, H., Koarashi, J., 2013. Root and dissolved organic carbon controls on subsurface soil carbon dynamics: A model approach. Journal of Geophysical Research: Biogeosciences 118, 1646–1659. doi:10.1002/2013JG002379
- Otto, R., Trivelin, P.C.O., Franco, H.C.J., Faroni, C.E., Vitti, A.C., 2009. Root system distribution of sugar cane as related to nitrogen fertilization, evaluated by two methods: monolith and probes. Revista Brasileira de Ciência Do Solo 33, 601–611. doi:10.1590/S0100-06832009000300013
- Paredes, D. da S., Lessa, A.C. da R., Sant'Anna, S.A.C. de, Boddey, R.M., Urquiaga, S., Alves, B.J.R., 2014. Nitrous oxide emission and ammonia volatilization induced by vinasse and N fertilizer

application in a sugarcane crop at Rio de Janeiro, Brazil. Nutrient Cycling in Agroecosystems 98, 41–55. doi:10.1007/s10705-013-9594-5

Parkin, T.B., 2008. Effect of sampling frequency on estimates of cumulative nitrous oxide emissions. Journal of Environmental Quality 37, 1390–1395. doi:10.2134/jeq2007.0333

Parton, W.J., 1988. Dynamics of C, N, P and S in grassland soils: a model. Biogeochemistry 5, 109 – 131.

- Parton, W.J., 1987. Analysis of factors controlling soil organic matter levels in Great Plains grasslands. Soil Science Society of America 51, 1173 – 1179.
- Parton, W.J., Del Grosso, S.J., Plante, A.F., Adair, E.C., Lutz, S.M., 2015. Modeling the dynamics of soil oragnic matter and nutrient cycling., in: Paul, E.A. (Ed.), Soil Microbiology, Ecology, and
   Biochemistry. Academic Press, 32 Jamestown Road, London, NW1 7BY, UK, pp. 505–537.
- Parton, W.J., Holland, E.A., Del Grosso, S.J., Hartman, M.D., Martin, R.E., Mosier, A.R., Ojima, D.S., Schimel, D.S., 2001. Generalized model for NO x and N2O emissions from soils. Journal of Geophysical Research: Atmospheres 106, 17403–17419. doi:10.1029/2001JD900101
- Parton, W., Silver, W.L., Burke, I.C., Grassens, L., Harmon, M.E., Currie, W.S., King, J.Y., Adair, E.C.,
   Brandt, L.A., Hart, S.C., Fasth, B., 2007. Global-Scale Similarities in Nitrogen Release Patterns
   During Long-Term Decomposition. Science 315, 361–364. doi:10.1126/science.1134853
- Paul, E.A., 1984. Dynamics of organic matter in soils. Plant and Soil 76, 275–285.
- Paul, E.A., Follett, R.F., Leavitt, S.W., Halvorson, A., Peterson, G.A., Lyon, D.J., 1997. Radiocarbon Dating for Determination of Soil Organic Matter Pool Sizes and Dynamics. Soil Science Society of America 61, 1058–1067.
- Paustian, K., 1994. Modeling soil biology and biochemical processes for sustainable agricultural research, in: Soil Biota: Management in Sustainable Farming Systems. CSIRO, Victoria, Australia, pp. 182– 193.

- Paustian, K., Cole, C.V., Sauerbeck, D., Sampson, N., 1998. CO2 Mitigation by Agriculture: An Overview. Climatic Change 40, 135–162. doi:10.1023/A:1005347017157
- Paustian, K., Collins, H.P., Paul, E.A., 1997. Management controls on soil carbon., in: Cole, C.V. (Ed.), Soil Organic Matter in Temperate Agroecosystems: Long-Term Experiments in North America. CRC Press, New York, NY, pp. 15–49.
- Paustian, K., Elliott, E., Collins, H., Cole, C., Paul, E., 1995. Use of a network of long-term experiments for analysis of soil carbon dynamics and global change: the North American model. Australian Journal of Experimental Agriculture 35, 929–939.
- Paustian, K., Levine, E., Post, W.M., Ryzhova, I.M., 1997. The use of models to integrate information and understanding of soil C at the regional scale. Geoderma 79, 227–260. doi:10.1016/S0016-7061(97)00043-8
- Piccolo, A., 2002. The supramolecular structure of humic substances: A novel understanding of humus chemistry and implications in soil science, in: Advances in Agronomy. Academic Press, pp. 57–134.
- Pitman, A.J., 2003. The evolution of, and revolution in, land surface schemes designed for climate models. International Journal of Climatology 23, 479–510. doi:10.1002/joc.893
- Plummer, M., Best, N., Cowles, K., Vines, K., 2006. CODA: Convergence Diagnosis and Output Analysis for MCMC. R News 6, 7–11.
- Poeplau, C., Don, A., 2013. Sensitivity of soil organic carbon stocks and fractions to different land-use changes across Europe. Geoderma 192, 189–201. doi:10.1016/j.geoderma.2012.08.003
- Power, J., Wilhelm, W., Doran, J., 1986. Crop residue effects on soil environment and dryland maize and soya bean production. Soil & Tillage Research 8, 101–111.
- Powlson, D., 2005. Climatology: Will soil amplify climate change? Nature 433, 204–205. doi:10.1038/433204a

- Preston, C.M., Nault, J.R., Trofymow, J.A., 2009. Chemical Changes During 6 Years of Decomposition of 11 Litters in Some Canadian Forest Sites. Part 2. 13C Abundance, Solid-State 13C NMR
  Spectroscopy and the Meaning of "Lignin." Ecosystems 12, 1078–1102. doi:10.1007/s10021-009-9267-z
- Pronk, G.J., Heister, K., Ding, G.-C., Smalla, K., Kögel-Knabner, I., 2012. Development of biogeochemical interfaces in an artificial soil incubation experiment; aggregation and formation of organomineral associations. Geoderma 189–190, 585–594. doi:10.1016/j.geoderma.2012.05.020
- Qian, H., Joseph, R., Zeng, N., 2010. Enhanced terrestrial carbon uptake in the Northern High Latitudes in the 21st century from the Coupled Carbon Cycle Climate Model Intercomparison Project model projections. Global Change Biology 16, 641–656. doi:10.1111/j.1365-2486.2009.01989.x
- Qualls, R.G., Haines, B.L., Swank, W.T., 1991. Fluxes of Dissolved Organic Nutrients and Humic Substances in a Deciduous Forest. Ecology 72, 254–266. doi:10.2307/1938919
- Randall, D.A., Wood, R.A., Bony, S., Colman, R., Fichefet, T., Fyfe, J., Kattsov, V., Pitman, A., Shukla, J.,
  Srinivasan, J., Stouffer, R.J., Sumi, A., Taylor, K.E., 2007. Climate models and their evaluation, in:
  Solomon, S., Qin, D., Manning, M., Chen, Z., Marquis, M., Averyt, K.B., Tignor, M., Miller, H.L.
  (Eds.), Climate Change 2007: The Physical Science Basis. Contribution of Working Group I to the
  Fourth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge
  University Press, United Kingdon & New York, NY USA.
- R Core Team, 2014. R: a language and environment for statistical computing. R Foundation for Statistical Computing, Vienna, Austria.
- R Core Team, 2012. R: a language and environment for statistical computing. R Foundation for Statistical Computing, Vienna, Austria.

- Reichstein, M., Kätterer, T., Andrén, O., Ciais, P., Schulze, E.-D., Cramer, W., Papale, D., Valentini, R., 2005. Temperature sensitivity of decomposition in relation to soil organic matter pools: critique and outlook. Biogeosciences 2, 317–321. doi:10.5194/bg-2-317-2005
- Reicosky, D.C., Evans, S.D., Cambardella, C.A., Allmaras, R.R., Wilts, A.R., Huggins, D.R., 2002. Continuous corn with moldboard tillage: residue and fertility effects on soil carbon. Journal of Soil and Water Conservation 57, 277–284.
- Reijnders, L., 2006. Conditions for the sustainability of biomass based fuel use. Energy Policy 34, 863– 876. doi:10.1016/j.enpol.2004.09.001
- Riley, W.J., Maggi, F., Kleber, M., Torn, M.S., Tang, J.Y., Dwivedi, D., Guerry, N., 2014. Long residence times of rapidly decomposable soil organic matter: application of a multi-phase, multi-component, and vertically resolved model (BAMS1) to soil carbon dynamics. Geosci. Model Dev. 7, 1335–1355. doi:10.5194/gmd-7-1335-2014
- Rillig, M.C., Caldwell, B.A., Wösten, H.A.B., Sollins, P., 2007. Role of proteins in soil carbon and nitrogen storage: controls on persistence. Biogeochemistry 85, 25–44. doi:10.1007/s10533-007-9102-6
- Rumpel, C., Kögel-Knabner, I., 2010. Deep soil organic matter—a key but poorly understood component of terrestrial C cycle. Plant and Soil 338, 143–158. doi:10.1007/s11104-010-0391-5
- Salomé, C., Nunan, N., Pouteau, V., Lerch, T.Z., Chenu, C., 2010. Carbon dynamics in topsoil and in subsoil may be controlled by different regulatory mechanisms. Global Change Biology 16, 416– 426. doi:10.1111/j.1365-2486.2009.01884.x
- Salter, R.M., Green, T.C., 1933. Factors affecting the accumulation and loss of nitrogen andorganic carbon in cropped soils. Journal of the American Society of Agronomy 25, 622–630.
- Sanaullah, M., Chabbi, A., Leifeld, J., Bardoux, G., Billou, D., Rumpel, C., 2011. Decomposition and stabilization of root litter in top- and subsoil horizons: what is the difference? Plant and Soil 338, 127–141. doi:10.1007/s11104-010-0554-4

- Sanderman, J., Baldock, J., Amundson, R., 2008. Dissolved organic carbon chemistry and dynamics in contrasting forest and grassland soils. Biogeochemistry 89, 181–198. doi:10.1007/s10533-008-9211-x
- Scharpenseel, H.W., Becker-Heidmann, P., 2006. Twenty-Five Years of Radiocarbon Dating Soils: Paradigm of Erring and Learning. Radiocarbon 34, 541–549. doi:10.2458/azu\_js\_rc.34.1502
- Scharpenseel, H.W., Becker-Heidmann, P., Neue, H.U., Tsutsuki, K., 1989. Bomb-carbon, 14C-dating and 13C — Measurements as tracers of organic matter dynamics as well as of morphogenetic and turbation processes. Science of The Total Environment, Advances in Humic Substances Research 81–82, 99–110. doi:10.1016/0048-9697(89)90115-0
- Schiff, S.L., Aravena, R., Trumbore, S.E., Hinton, M.J., Elgood, R., Dillon, P.J., 1997. Export of DOC from forested catchments on the Precambrian Shield of Central Ontario: Clues from 13C and 14C.
   Biogeochemistry 36, 43–65. doi:10.1023/A:1005744131385
- Schimel, D.S., 1995. Terrestrial ecosystems and the carbon cycle. Global Change Biology 1, 77–91. doi:10.1111/j.1365-2486.1995.tb00008.x
- Schimel, D.S., Braswell, B.H., Holland, E.A., McKeown, R., Ojima, D.S., Painter, T.H., Parton, W.J., Townsend, A.R., 1994. Climatic, edaphic, and biotic controls over storage and turnover of carbon in soils. Global Biogeochemical Cycles 8, 279–293. doi:10.1029/94GB00993
- Schimel, J., 2013. Soil carbon: Microbes and global carbon. Nature Climate Change 3, 867–868. doi:10.1038/nclimate2015
- Schimel, J.P., Wetterstedt, J.Å.M., Holden, P.A., Trumbore, S.E., 2011. Drying/rewetting cycles mobilize old C from deep soils from a California annual grassland. Soil Biology and Biochemistry 43, 1101– 1103. doi:10.1016/j.soilbio.2011.01.008
- Schmidt, M.W.I., Torn, M.S., Abiven, S., Dittmar, T., Guggenberger, G., Janssens, I.A., Kleber, M., Kögel-Knabner, I., Lehmann, J., Manning, D.A.C., Nannipieri, P., Rasse, D.P., Weiner, S., Trumbore, S.E.,

2011. Persistence of soil organic matter as an ecosystem property. Nature 478, 49–56. doi:10.1038/nature10386

Schnitzer, M., Khan, S., 1972. Humic substances in the environment. Marcel Dekker, Inc., New York, NY.

- Seabra, J.E.A., Tao, L., Chum, H.L., Macedo, I.C., 2010. A techno-economic evaluation of the effects of centralized cellulosic ethanol and co-products refinery options with sugarcane mill clustering.
   Biomass and Bioenergy 34, 1065–1078. doi:10.1016/j.biombioe.2010.01.042
- Searchinger, T., Heimlich, R., Houghton, R.A., Dong, F., Elobeid, A., Fabiosa, J., Tokgoz, S., Hayes, D., Yu, T.-H., 2008. Use of U.S. Croplands for Biofuels Increases Greenhouse Gases Through Emissions from Land-Use Change. Science 319, 1238–1240. doi:10.1126/science.1151861
- Segoli, M., De Gryze, S., Dou, F., Lee, J., Post, W.M., Denef, K., Six, J., 2013. AggModel: A soil organic
  matter model with measurable pools for use in incubation studies. Ecological Modelling 263, 1–
  9. doi:10.1016/j.ecolmodel.2013.04.010
- Serna-Chavez, H.M., Fierer, N., van Bodegom, P.M., 2013. Global drivers and patterns of microbial abundance in soil. Global Ecology and Biogeography 22, 1162–1172. doi:10.1111/geb.12070
- Sharpe, P.J.A., 1990. Forest modeling approaches: compromises between generality and precision., in: Process Modeling of Forest Growth Responses to Environmental Stress. Timber Press, Portland, OR, pp. 180–190.
- Signor, D., Cerri, C.E.P., Conant, R., 2013. N2O emissions due to nitrogen fertilizer applications in two regions of sugarcane cultivation in Brazil. Environmental Research Letters 8, 015013. doi:10.1088/1748-9326/8/1/015013
- Simpson, A.J., Simpson, M.J., Smith, E., Kelleher, B.P., 2007. Microbially Derived Inputs to Soil Organic Matter: Are Current Estimates Too Low? Environmental Science & Technology 41, 8070–8076. doi:10.1021/es071217x

- Simpson, M., Simpson, A., 2012. The chemical ecology of soil organic matter molecular constituents. Journal of Chemical Ecology 38, 768–784. doi:10.1007/s10886-012-0122-x
- Sindelar, A.J., Lamb, J.A., Sheaffer, C.C., Jung, H.G., Rosen, C.J., 2012. Response of corn grain, cellulosic biomass, and ethanol yields to nitrogen fertilization. Agronomy Journal 104, 363–370. doi:10.2134/agronj2011.0279

Sindelar, A.J., Lamb, J.A., Sheaffer, C.C., Rosen, C.J., Jung, H.G., 2013. Fertilizer nitrogen rate effects on nutrient removal by corn stover and cobs. Agronomy Journal 105, 437–445. doi:10.2134/agronj2012.0240

- Sinsabaugh, R.L., Manzoni, S., Moorhead, D.L., Richter, A., 2013. Carbon use efficiency of microbial communities: stoichiometry, methodology and modelling. Ecology Letters 16, 930–939. doi:10.1111/ele.12113
- Sitch, S., Smith, B., Prentice, I.C., Arneth, A., Bondeau, A., Cramer, W., Kaplan, J.O., Levis, S., Lucht, W.,
   Sykes, M.T., Thonicke, K., Venevsky, S., 2003. Evaluation of ecosystem dynamics, plant
   geography and terrestrial carbon cycling in the LPJ dynamic global vegetation model. Global
   Change Biology 9, 161–185. doi:10.1046/j.1365-2486.2003.00569.x
- Six, J., Bossuyt, H., Degryze, S., Denef, K., 2004. A history of research on the link between (micro)aggregates, soil biota, and soil organic matter dynamics. Soil and Tillage Research, Advances in Soil Structure Research 79, 7–31. doi:10.1016/j.still.2004.03.008
- Six, J., Conant, R., Paul, E., Paustian, K., 2002. Stabilization mechanisms of soil organic matter: Implications for C-saturation of soils. Plant and Soil 241, 155–176.

Six, J., Paustian, K., 2014. Aggregate-associated soil organic matter as an ecosystem property and a measurement tool. Soil Biology and Biochemistry 68, A4–A9. doi:10.1016/j.soilbio.2013.06.014

Smith, O.L., 1979. An analytical model of the decomposition of soil organic matter. Soil Biology & Biochemistry 11, 585–606.

- Smith, P., Fang, C., 2010. Carbon cycle: A warm response by soils. Nature 464, 499–500. doi:10.1038/464499a
- Smith, P., Smith, J.U., Powlson, D.S., McGill, W.B., Arah, J.R.M., Chertov, O.G., Coleman, K., Franko, U.,
  Frolking, S., Jenkinson, D.S., Jensen, L.S., Kelly, R.H., Klein-Gunnewiek, H., Komarov, A.S., Li, C.,
  Molina, J.A.E., Mueller, T., Parton, W.J., Thornley, J.H.M., Whitmore, A.P., 1997. A comparison of
  the performance of nine soil organic matter models using datasets from seven long-term
  experiments. Geoderma 81, 153–225. doi:10.1016/S0016-7061(97)00087-6
- Smolander, A., Kitunen, V., 2002. Soil microbial activities and characteristics of dissolved organic C and N in relation to tree species. Soil Biology and Biochemistry 34, 651–660. doi:10.1016/S0038-0717(01)00227-9
- Soetaert, K., Petzoldt, T., Setzer, R.W., 2010. Solving Differential Equations in R: Package deSolve. Journal of Statistical Software 33, 1–25.
- Sollins, P., Homann, P., Caldwell, B.A., 1996. Stabilization and destabilization of soil organic matter: mechanisms and controls. Geoderma 74, 65–105. doi:10.1016/S0016-7061(96)00036-5
- Solomon, B.D., Barnes, J.R., Halvorsen, K.E., 2007. Grain and cellulosic ethanol: History, economics, and energy policy. Biomass and Bioenergy 31, 416–425. doi:16/j.biombioe.2007.01.023
- Soong, J.L., Parton, W.J., Calderon, F., Campbell, E.E., Cotrufo, M.F., 2015. A new conceptual model on the fate and controls of fresh and pyrolized plant litter decomposition. Biogeochemistry 1–18. doi:10.1007/s10533-015-0079-2
- Steinweg, J.M., Dukes, J.S., Wallenstein, M.D., 2012. Modeling the effects of temperature and moisture on soil enzyme activity: Linking laboratory assays to continuous field data. Soil Biology and Biochemistry 55, 85–92. doi:10.1016/j.soilbio.2012.06.015

- Stetson, S.J., Osborne, S.L., Schumacher, T.E., Eynard, A., Chilom, G., Rice, J., Nichols, K.A., Pikul, J.L., 2012. Corn residue removal impact on topsoil organic carbon in a corn–soybean rotation. Soil Science Society of America Journal 76, 1399–1406. doi:10.2136/sssaj2011.0420
- Stewart, C.E., Follett, R.F., Wallace, J., Pruessner, E.G., 2012. Impact of biosolids and tillage on soil organic matter fractions: implications of carbon saturation for conservation management in the Virginia Coastal Plain. Soil Science Society of America Journal 76, 1257–1267.
- Stewart, C.E., Paustian, K., Conant, R.T., Plante, A.F., Six, J., 2009. Soil carbon saturation: Implications for measurable carbon pool dynamics in long-term incubations. Soil Biology and Biochemistry 41, 357–366. doi:10.1016/j.soilbio.2008.11.011
- Stewart, C.E., Paustian, K., Conant, R.T., Plante, A.F., Six, J., 2008a. Soil carbon saturation: Evaluation and corroboration by long-term incubations. Soil Biology and Biochemistry 40, 1741–1750. doi:10.1016/j.soilbio.2008.02.014
- Stewart, C.E., Plante, A.F., Paustian, K., Conant, R.T., Six, J., 2008b. Soil Carbon Saturation: Linking Concept and Measurable Carbon Pools. Soil Science Society of America Journal 72, 379. doi:10.2136/sssaj2007.0104
- Stewart, C., Paustian, K., Conant, R., Plante, A., Six, J., 2007. Soil carbon saturation: concept, evidence and evaluation. Biogeochemistry 86, 19–31. doi:10.1007/s10533-007-9140-0
- Stockmann, U., Adams, M.A., Crawford, J.W., Field, D.J., Henakaarchchi, N., Jenkins, M., Minasny, B.,
  McBratney, A.B., Courcelles, V. de R. de, Singh, K., Wheeler, I., Abbott, L., Angers, D.A., Baldock,
  J., Bird, M., Brookes, P.C., Chenu, C., Jastrow, J.D., Lal, R., Lehmann, J., O'Donnell, A.G., Parton,
  W.J., Whitehead, D., Zimmermann, M., 2013. The knowns, known unknowns and unknowns of
  sequestration of soil organic carbon. Agriculture, Ecosystems & Environment 164, 80–99.
  doi:10.1016/j.agee.2012.10.001

- Subke, J.-A., Bahn, M., 2010. On the "temperature sensitivity" of soil respiration: Can we use the immeasurable to predict the unknown? Soil Biology and Biochemistry 42, 1653–1656. doi:10.1016/j.soilbio.2010.05.026
- Sutton, R., Sposito, G., 2005. Molecular Structure in Soil Humic Substances: The New View. Environmental Science & Technology 39, 9009–9015. doi:10.1021/es050778q
- Swift, M.J., Heal, O.W., Anderson, J.M., 1979. Decomposition in Terrestrial Ecosystems. University of California Press.
- Taghizadeh-Toosi, A., Christensen, B.T., Hutchings, N.J., Vejlin, J., Kätterer, T., Glendining, M., Olesen, J.E., 2014. C-TOOL: A simple model for simulating whole-profile carbon storage in temperate agricultural soils. Ecological Modelling 292, 11–25. doi:10.1016/j.ecolmodel.2014.08.016
- Tang, J., Riley, W.J., 2015. Weaker soil carbon-climate feedbacks resulting from microbial and abiotic interactions. Nature Climate Change 5, 56–60. doi:10.1038/nclimate2438
- Tang, J., Zhuang, Q., 2008. Equifinality in parameterization of process-based biogeochemistry models: A significant uncertainty source to the estimation of regional carbon dynamics. Journal of Geophysical Research 113. doi:10.1029/2008JG000757
- Thornley, J.H.M., Cannell, M.G.R., 2001. Soil Carbon Storage Response to Temperature: an Hypothesis. Annals of Botany 87, 591–598. doi:10.1006/anbo.2001.1372
- Tipping, E., Chamberlain, P., Fröberg, M., Hanson, P., Jardine, P., 2012. Simulation of carbon cycling,
   including dissolved organic carbon transport, in forest soil locally enriched with 14C.
   Biogeochemistry 108, 91–107. doi:10.1007/s10533-011-9575-1
- Tisdall, J.M., Oades, J.M., 1982. Organic matter and water-stable aggregates in soils. Journal of Soil Science 33, 141–163. doi:10.1111/j.1365-2389.1982.tb01755.x

- Todd-Brown, K.E.O., Randerson, J.T., Post, W.M., Hoffman, F.M., Tarnocai, C., Schuur, E.A.G., Allison, S.D., 2013. Causes of variation in soil carbon simulations from CMIP5 Earth system models and comparison with observations. Biogeosciences 10, 1717–1736. doi:10.5194/bg-10-1717-2013
- Todd-Brown, K., Hopkins, F., Kivlin, S., Talbot, J., Allison, S., 2012. A framework for representing microbial decomposition in coupled climate models. Biogeochemistry 109, 19–33. doi:10.1007/s10533-011-9635-6
- Treseder, K.K., Balser, T.C., Bradford, M.A., Brodie, E.L., Dubinsky, E.A., Eviner, V.T., Hofmockel, K.S., Lennon, J.T., Levine, U.Y., MacGregor, B.J., Pett-Ridge, J., Waldrop, M.P., 2012. Integrating microbial ecology into ecosystem models: challenges and priorities. Biogeochemistry 109, 7–18. doi:http://dx.doi.org.ezproxy2.library.colostate.edu:2048/10.1007/s10533-011-9636-5
- Trumbore, S., 2009. Radiocarbon and Soil Carbon Dynamics. Annual Review of Earth and Planetary Sciences 37, 47–66. doi:10.1146/annurev.earth.36.031207.124300
- Trumbore, S., 2000. Age of soil organic matter and soil respiration: radiocarbon constraints on belowground c dynamics. Ecological Applications 10, 399–411. doi:10.1890/1051-0761(2000)010[0399:AOSOMA]2.0.CO;2
- Trumbore, S.E., Chadwick, O.A., Amundson, R., 1996. Rapid exchange between soil carbon and atmospheric carbon dioxide driven by temperature change. Science 272, 393.
- Trumbore, S.E., Davidson, E.A., Barbosa de Camargo, P., Nepstad, D.C., Martinelli, L.A., 1995. Belowground cycling of carbon in forests and pastures of eastern Amazonia. Global Biogeochemical Cycles 9, 515–528. doi:10.1029/95GB02148
- Tucker, C.L., Bell, J., Pendall, E., Ogle, K., 2013. Does declining carbon-use efficiency explain thermal acclimation of soil respiration with warming? Global Change Biology 19, 252–263. doi:10.1111/gcb.12036

- Tuomi, M., Vanhala, P., Karhu, K., Fritze, H., Liski, J., 2008. Heterotrophic soil respiration—Comparison of different models describing its temperature dependence. Ecological Modelling 211, 182–190. doi:10.1016/j.ecolmodel.2007.09.003
- U.S. Department of Energy, 2011. U.S. billion-ton update: biomass supply for a bioenergy and bioproducts industry (No. ORNL/TM-2011/224). Oak Ridge National Laboratory, Oak Ridge, TN.
- U.S. Environmental Protection Agency, 2013. Inventory of U.S. greenhouse gas emissions and sinks: 1990 - 2011 (No. EPA 430-R-13-001). 1200 Pennsylvania Ave., NW, Washington DC.
- Van Soest, P.J., Wine, R.H., 1968. Determination of lignin and cellulose in acid-detergent fiber with permanganate. J Assoc Off Anal Chem 51, 780.
- Varvel, G.E., Vogel, K.P., Mitchell, R.B., Follett, R.F., Kimble, J.M., 2008. Comparison of corn and switchgrass on marginal soils for bioenergy. Biomass and Bioenergy 32, 18–21. doi:10.1016/j.biombioe.2007.07.003
- Vitousek, P.M., 1994. Factors controlling ecosystem structure and function, in: Factors of Soil Formation: A Fiftieth Anniversary Retrospective, SSSA Special Publication. Soil Science Society of America, Madison, WI, pp. 87–97.
- Von Lutzow, M., 2008. Stabilization mechanisms of organic matter in four temperate soils: development and application of a conceptual model. J. Plant Nutr. Soil Sci. 171, 111–124.
- Von Lützow, M., Kögel-Knabner, I., Ekschmitt, K., Flessa, H., Guggenberger, G., Matzner, E., Marschner,
   B., 2007. SOM fractionation methods: Relevance to functional pools and to stabilization
   mechanisms. Soil Biology and Biochemistry 39, 2183–2207. doi:10.1016/j.soilbio.2007.03.007
- Wagai, R., Kishimoto-Mo, A.W., Yonemura, S., Shirato, Y., Hiradate, S., Yagasaki, Y., 2013. Linking temperature sensitivity of soil organic matter decomposition to its molecular structure, accessibility, and microbial physiology. Global Change Biology 19, 1114–1125. doi:10.1111/gcb.12112

- Waksman, S., 1926. The micro-biological complexities of the soil and soil deterioration. Journal of the American Society of Agronomy 18, 137–142.
- Waksman, S.A., 1938. Humus. Origin, chemical composition and importance in nature., 2nd ed. The William and Wilkins Company, Baltimore, London.
- Waksman, S.A., Stevens, K.R., 1930. A critical study of the methods for determining the nature and abundance of soil organic matter. Soil Science 30, 79–143.
- Wander, M., 2004. Soil organic matter fractions and their relevance to soil function., in: Magdoff, F. (Ed.), Soil Organic Matter in Sustainable Agriculture. CRC Press, Boca Raton, Florida.
- Wang, J., Bao, Q., Zeng, N., Liu, Y., Wu, G., Ji, D., 2013. Earth System Model FGOALS-s2: Coupling a dynamic global vegetation and terrestrial carbon model with the physical climate system model. Advances in Atmospheric Sciences 30, 1549–1559.

doi:http://dx.doi.org.ezproxy2.library.colostate.edu:2048/10.1007/s00376-013-2169-1

- Wang, Y.P., Chen, B.C., Wieder, W.R., Leite, M., Medlyn, B.E., Rasmussen, M., Smith, M.J., Agusto, F.B., Hoffman, F., Luo, Y.Q., 2014. Oscillatory behavior of two nonlinear microbial models of soil carbon decomposition. Biogeosciences 11, 1817–1831. doi:10.5194/bg-11-1817-2014
- Wells, P., Faro, T., 2011. Eco-efficiency, self-sufficiency and sustainability in transport: The limits for Brazilian sugarcane ethanol policy. Natural Resources Forum 35, 21–31. doi:10.1111/j.1477-8947.2010.01340.x

- Wickham, H., 2009. ggplot2: elegant graphics for data analysis. Springer, New York.
- Wickham, H., Francois, R., 2015. dplyr: A Grammar of Data Manipulation.
- Wieder, W.R., Boehnert, J., Bonan, G.B., 2014. Evaluating soil biogeochemistry parameterizations in Earth system models with observations. Global Biogeochemical Cycles 28, 211–222. doi:10.1002/2013GB004665

Wickham, H., 2014. tidyr: Easily Tidy Data with spread() and gather() Functions.

- Wieder, W.R., Bonan, G.B., Allison, S.D., 2013. Global soil carbon projections are improved by modelling microbial processes. Nature Climate Change 3, 909–912. doi:10.1038/nclimate1951
- Wieder, W.R., Grandy, A.S., Kallenbach, C.M., Bonan, G.B., 2014. Integrating microbial physiology and physiochemical principles in soils with the MIcrobial-MIneral Carbon Stabilization (MIMICS) model. Biogeosciences Discuss. 11, 1147–1185. doi:10.5194/bgd-11-1147-2014
- Wiggans, D.R., Singer, J.W., Moore, K.J., Lamkey, K.R., 2012. Response of continuous maize with stover removal to living mulches. Agronomy Journal 104, 917–925. doi:10.2134/agronj2011.0395
- Wilhelm, W., Doran, J., Power, J., 1986. Corn and soybean yield response to crop residue management under no-tillage production systems. Agronomy Journal 78, 184–189.
- Wilhelm, W.W., Hess, J.R., Karlen, D.L., Johnson, J.M.F., Muth, D.J., Baker, J.M., Gollany, H.T., Novak,
   J.M., Stott, D.E., Varvel, G.E., 2010. REVIEW: Balancing limiting factors & economic drivers for
   sustainable Midwestern US agricultural residue feedstock supplies. Industrial Biotechnology 6,
   271–287. doi:10.1089/ind.2010.6.271
- Wilhelm, W.W., Johnson, J.M.F., Hatfield, J.L., Voorhees, W.B., Linden, D.R., 2004. Crop and soil productivity response to corn residue removal. Agronomy Journal 96, 1 17.
- Wilts, A.R., Reicosky, D.C., Allmaras, R.R., Clapp, C.E., 2004. Long-term corn residue effects: harvest alternatives, soil carbon turnover, and root-derived carbon. Soil Science Society of America Journal 68, 1342–1351. doi:10.2136/sssaj2004.1342
- Xia, J.Y., Luo, Y.Q., Wang, Y.-P., Weng, E.S., Hararuk, O., 2012. A semi-analytical solution to accelerate spin-up of a coupled carbon and nitrogen land model to steady state. Geosci. Model Dev. 5, 1259–1271. doi:10.5194/gmd-5-1259-2012
- Xu, X., Schimel, J.P., Thornton, P.E., Song, X., Yuan, F., Goswami, S., 2014. Substrate and environmental controls on microbial assimilation of soil organic carbon: a framework for Earth system models.
   Ecology Letters 17, 547–555. doi:10.1111/ele.12254

- Xu, X., Thornton, P.E., Post, W.M., 2013. A global analysis of soil microbial biomass carbon, nitrogen and phosphorus in terrestrial ecosystems. Global Ecology and Biogeography 22, 737–749.
   doi:10.1111/geb.12029
- Yang, X., Thornton, P.E., Ricciuto, D.M., Post, W.M., 2014. The role of phosphorus dynamics in tropical forests a modeling study using CLM-CNP. Biogeosciences 11, 1667–1681. doi:10.5194/bg-11-1667-2014
- Young, I.M., Crawford, J.W., 2004. Interactions and Self-Organization in the Soil-Microbe Complex. Science 304, 1634–1637. doi:10.1126/science.1097394
- Zambrano-Bigiarini, M., 2014. hydroGOF: Goodness-of-fit functions for comparison of simulated and observed hydrological time series.
- Zhang, W.J., Wang, X.J., Xu, M.G., Huang, S.M., Liu, H., Peng, C., 2010. Soil organic carbon dynamics under long-term fertilizations in arable land of northern China. Biogeosciences 7, 409–425. doi:10.5194/bg-7-409-2010
- Zhang, X., Izaurralde, R.C., Manowitz, D., West, T.O., Post, W.M., Thomson, A.M., Bandaru, V.P., Nichols, J., Williams, J.R., 2010. An integrative modeling framework to evaluate the productivity and sustainability of biofuel crop production systems. Global Change Biology Bioenergy 2, 258 – 277.
- Zimmermann, M., Leifeld, J., Schmidt, M.W.I., Smith, P., Fuhrer, J., 2007. Measured soil organic matter fractions can be related to pools in the RothC model. European Journal of Soil Science 58, 658– 667. doi:10.1111/j.1365-2389.2006.00855.x

#### APPENDIX 1

In this section we detail the hierarchical Bayesian analysis used for the initial conditions of the LIDEL model. We describe the steps of the Monte Carlo Markov chain (MCMC) algorithm used to estimate the posterior distributions of model parameters. We define the notation used to express the process models, data models, and parameter models, and then describe the assumptions for the data and parameter models. We then describe the steps of the MCMC algorithm used in this analysis.

The LIDEL model has 7 pools. The microbial ( $C_4$ ), microbial products ( $C_5$ ), and CO<sub>2</sub> ( $C_7$ ) pools each have an initial value of 0. This analysis estimates the posterior distribution for the latent ("true") states of the non-lignin structural ( $C_2$ ) and DOC ( $C_6$ ) pools, as well as initial litter mass ( $L_{iC_M\tau}$ ) based on parameters for the fraction of initial litter C that is soluble ( $f_{s_i}$ ), the fraction of soluble material that bypasses all processes reflected in the LIDEL model when exposed to water ( $f_{DOC}$ ), and the fraction of initial litter C that is lignin ( $f_{LIG_i}$ ). The initial values for the soluble ( $C_1$ ) and lignin ( $C_3$ ) LIDEL pools are then calculated by using the posterior distribution of these parameters and the initial litter mass latent state.

# **A1.1 Definitions**

i = litter type (alfalfa, ash, bluestem, oak pine)

k = measurement replicate

L = latent state, or the 'true' underlying condition of the C mass

z = process model state

s = index for pools in the LIDEL model linked to measured data;  $s = C_2, C_6$ , where  $C_2$  indicates the non-lignin structural (NLS) pool, and  $C_6$  indicates the DOC pool

 $\tau =$  initial time point, where day = 0

 $f_{DOC}$  = fraction of DOC, non-microbially released, from plant soluble C

191

 $f_{s_i}$  = fraction of plant soluble carbon (C) by litter type (i)

 $f_{LIG_i}$  = fraction of lignin C by litter type (*i*)

## A1.2 Data models

The % C fraction measurements were assumed to have beta distributions, which have shape parameters calculated by moment matching the mean of the  $f_{s_i}$  or  $f_{LIG_i}$  parameters and the variance of the given measurement type ( $\sigma_{HWE}^2$ ,  $\sigma_{DIF}^2$ ,  $\sigma_{LIG}^2$ ).

 $y_{HWE_{ik}}$  =extraction measurement of soluble litter C, as % C in hot water extracted material.

These measurements were drawn from Soong et al. (2015) as well as a review of published literature.

$$y_{HWE_{ik}} \sim beta(g(f_{s_i}, \sigma_{HWE}^2))$$

 $y_{DIF_{ik}}$  = mass difference measurement of soluble litter C, as % remaining material after acid detergent fiber digestion. These measurements were drawn from Soong et al. (2015) as well as a review of published literature.

$$y_{DIF_{ik}} = beta(g(f_{s_i}, \sigma_{DIF}^2))$$

 $y_{LIG_{ik}} = \%$  acid unhydrolyzable residue C, with acid detergent fiber digestion method. These measurements were drawn from Soong et al. (2015)

$$y_{LIG_{ik}} = beta(g(f_{LIG_i}, \sigma_{LIG}^2))$$

 $y_{iC_M\tau k}$  = mass of C in initial litter, drawn from Soong et al. (2015). This measurement is assumed to be normally distributed, with a mean and a variance determined by the latent state of the litter mass and the measurement error, where:

 $y_{iC_M\tau k} = normal(L_{iC_M\tau}, \sigma_{yC_M i}^2).$ 

 $y_{iC_2\tau k}$  =mass of the cellulose and hemicellulose C in initial litter mass, using acid detergent fiber digestion, taken from Soong et al. (2015). In the LIDEL model we chose to define the non-lignin structural (NLS) pool ( $C_2$ ) as everything remaining when the lignin pool and the soluble pool had been removed. However this pool is predominantly cellulose and hemicellulose. Therefore the direct measurement of cellulose and hemicellulose was assumed to be a reasonable approximation for the size of the NLS pool. This measurement could not be used for the temporal dynamics analysis because hemicellulose was not measured at subsequent time points. This measurement is assumed to be normally distributed, with a mean and a variance determined by the latent state of the cellulose plus the hemicellulose mass and the measurement error, where;

 $y_{iC_2\tau k} = normal(L_{iC_2\tau}, \sigma_{yC_2i}^2)$ 

 $y_{iC_6\tau k}$  =mass of cumulative DOC as of day 10 of the experiment, taken from Soong et al. (2015). Conceptually, the initial value of the DOC ( $C_6$ ) pool in the LIDEL model is representative of soluble litter C that is released immediately with exposure to water, without any exposure to microbial processes. Thus the initial value of this pool indicates the quantity of litter C that bypasses all processes simulated in the LIDEL model. This is an unknown and unmeasured value that conceptually important but extremely difficult to observe directly. However given changes in DOC chemistry observed in Soong et al. (2015) we made the assumption that the DOC released in the first 10 days of the experiment is a reasonable approximation. These data were assumed to be normally distributed, with a mean and variance determined by the latent state of the initial DOC mass and measurement error, where:

$$y_{iC_6\tau k} = normal(L_{iC_6\tau}, \sigma_{yC_6i}^2)$$

## A1.3 Process models

Models for the initial state of the non-lignin structural () and DOC () LIDEL model pools.

$$z_{iC_{2}\tau} = L_{iC_{M}\tau} \left( 1 - \left( f_{s_{i}} + f_{LIG_{i}} \right) \right) + \varepsilon_{2}, \text{ where } \varepsilon_{2} = normal(0, \sigma_{C_{2}}^{2}), \text{ and}$$
$$z_{iC_{6}\tau} = L_{iC_{M}\tau} * f_{DOC} * f_{s_{i}} + \varepsilon_{6}, \text{ where } \varepsilon_{6} = normal(0, \sigma_{C_{6}}^{2})$$

#### A1.4 Parameter models

We assumed the latent states for the non-lignin structural ( $C_2$ ) and DOC ( $C_6$ ) pools to be normally distributed, with a mean of the process model prediction and a variance based on model error:

$$L_{iC_2\tau} \sim normal(g(f_{S_i}, f_{LIG_i}, L_{iC_M\tau}), \sigma_{C_2}^2)$$
, and

# $L_{iC_6\tau} \sim normal(g(f_{DOC}, f_{S_i}, L_{iC_M\tau}), \sigma_{C_6}^2).$

The % soluble C in the initial litter pool as well as the % lignin C are assumed to have beta distributions, but given uninformative priors due to lack of knowledge. Therefore:

 $f_{s_i} \sim beta(1,1)$ , and

 $f_{LIG_i} \sim beta(1,1).$ 

The % soluble C in the initial DOC pool is assumed to have a beta distribution, but is given an uninformative prior due to the lack of knowledge of this parameter. Therefore:

 $f_{DOC} \sim beta(1,1).$ 

For total litter mass, we assume an uninformative uniform prior distribution on the Latent state, in units of g C as;

$$L_{iC_M\tau} \sim uniform(min = 1, max = 1500).$$

Measured and modeled variance terms for all mass values were assumed to have inverse gamma distributions. Measurement variances for  $\sigma_{yC_M i}^2$ ,  $\sigma_{yC_2 i}^2$ , and  $\sigma_{yC_6 i}^2$  were given informative priors based on known measurement error in the instruments used to collect observations, at 5-10% of the quantity measured. Informative priors were calculated by determining the two variance terms that placed 95% of the normal distribution within either 5 or 10%, respectively, of the mean of all measurements for each litter type. The shape parameters were then calculated assuming that 95% of the inverse gamma distribution for the variance was within these two variance values. Model variances were given uninformative priors. Therefore;

$$\sigma_{yC_{2}i}^{2} \sim inverse \ gamma(\alpha_{2i}, \beta_{2i}) \text{ where};$$
  

$$\alpha_{2} = (8.5, 8.5, 8.5, 8.5, 8.5) \text{ and } \beta_{2} = (337.1, 968.6, 2842.8, 953.1, 1152.6),$$
  

$$\sigma_{yC_{6}i}^{2} \sim inverse \ gamma(\alpha_{6i}, \beta_{6i}) \text{ where};$$
  

$$\alpha_{6} = (8.5, 8.5, 8.5, 8.5, 8.5, 8.5) \text{ and } \beta_{6} = (116.4, 77.4, 10.5, 17.3, 3.9),$$
  

$$\sigma_{yC_{M}i}^{2} \sim inverse \ gamma(\alpha_{Mi}, \beta_{Mi}) \text{ where};$$

 $\pmb{\alpha}_M = (8.5, 8.5, 8.5, 8.5, 8.5)$  and  $\pmb{\beta}_M = (7684.9, 8550.5, 7670.9, 8912.4, 10815.0)$ , and in units of grams carbon as

 $\sigma_c^2 \sim uniform(min = 1, max = 100).$ 

The informative prior alpha and beta terms reported above are for the variance of the mass measurements, in units of grams carbon.

The variance terms for all fractional measurements were assumed to have beta distributions, and were given uninformative priors. Therefore:

$$\sigma^2_{HWE} \sim beta(1,1),$$
  
 $\sigma^2_{DIF} \sim beta(1,1),$  and  
 $\sigma^2_{LIG} \sim beta(1,1).$ 

# A1.5 Monte Carlo Markov chain algorithm

The joint posterior distribution for the latent states and parameters described above, give the observations, can be expressed as:

$$[f, \alpha, \beta, L, \sigma_{z}^{2}, \sigma_{y}^{2}|y] \propto$$

$$\prod_{s=1}^{S} \prod_{i=1}^{I} \prod_{k=1}^{K} [y_{iC_{s}\tau k}|L_{iC_{s}\tau}, \sigma_{yC_{s}i}^{2}] X [y_{HWE_{ik}}|g(f_{s_{i}}, \sigma_{HWE}^{2})] X [y_{DIF_{ik}}|g(f_{s_{i}}, \sigma_{DIF}^{2})]$$

$$X [y_{LIG_{ik}}|g(f_{LIG_{i}}, \sigma_{LIG}^{2})] X [L_{iC_{z}\tau}|g(f_{s_{i}}, f_{LIG_{i}}, L_{iC_{M}\tau}), \sigma_{C_{2}}^{2}] X [L_{iC_{6}\tau}|g(f_{DOC}, f_{s_{i}}, L_{iC_{M}\tau}), \sigma_{C_{6}}^{2}]$$

$$X [L_{iC_{M}\tau}|1,1500] X [f_{s_{i}}|1,1] X [f_{LIG_{i}}|1,1] X [f_{DOC}|1,1] X [\sigma_{yC_{2}i}^{2}|\alpha_{2i}, \beta_{2i}]$$

$$X [\sigma_{yC_{6}i}^{2}|\alpha_{6i}, \beta_{6i}] X [\sigma_{yC_{M}i}^{2}|\alpha_{Mi}, \beta_{Mi}] X [\sigma_{HWE}^{2}|1,1] X [\sigma_{DIF}^{2}|1,1] X [\sigma_{LIG}^{2}|1,1] X [\sigma_{C_{s}}^{2}|1,100]$$

We approximated the marginal posterior distributions for each latent state and parameter using a random-walk Metropolis Hastings Markov Chain-Monte Carlo (MCMC) algorithm. We present the steps of the MCMC algorithm in the following section.

Step 1.  $f, \alpha, \beta, L, \sigma_z^2, \sigma_y^2$  parameters and latent variables are initialized with starting values.

Step 2. The soluble fraction parameter vector  $(f_{s_i})$  is updated in order for each individual litter type, based on the conditional distribution given by

$$[f_{s_i}|\cdot] \propto \prod_{k=1}^{K} \left[ \left[ y_{HWE_{ik}} | g(f_{s_i}, \sigma_{HWE}^2) \right] \left[ y_{DIF_{ik}} | g(f_{s_i}, \sigma_{DIF}^2) \right] \right] \left[ L_{iC_2\tau} | g(f_{s_i}, f_{LIG_i}, L_{iC_M\tau}), \sigma_{C_2}^2 \right]$$
$$[L_{iC_6\tau} | g(f_{DOC}, f_{s_i}, L_{iC_M\tau}), \sigma_{C_6}^2] [f_{s_i}|1,1].$$

Step 3. The lignin fraction parameter vector ( $f_{LIG_i}$ ) is updated in order for each individual litter type, based on the conditional distribution given by

$$[f_{s_i}|\cdot] \propto \prod_{k=1}^{K} [y_{LIG_{ik}}|g(f_{LIG_i},\sigma_{LIG}^2)] [L_{iC_2\tau}|g(f_{s_i},f_{LIG_i},L_{iC_M\tau}),\sigma_{C_2}^2] [f_{LIG_i}|1,1].$$

Step 4. Individual litter value for latent variables for  $L_{iC_M\tau}$  are updated in succession based on its conditional distribution, given by

$$\begin{split} & [L_{iC_M\tau}|\cdot] \propto \prod_{k=1}^{K} [y_{iC_M\tau k}|L_{iC_M\tau},\sigma_{yC_Mi}^2] [L_{iC_2\tau}|g(f_{s_i},f_{LIG_i},L_{iC_M\tau}),\sigma_{C_2}^2] \\ & [L_{iC_6\tau}|g(f_{DOC},f_{s_i},L_{iC_M\tau}),\sigma_{C_6}^2] [L_{iC_M\tau}|1,1500] \end{split}$$

Step 5. Measurement and model variance are then updated in succession, given by the conditional distributions

$$\begin{split} & \left[\sigma_{yC_{M}i}^{2}|\cdot\right] \propto \prod_{k=1}^{K} [y_{iC_{M}\tau k}|L_{iC_{M}\tau},\sigma_{yC_{M}i}^{2}] [\sigma_{yC_{M}i}^{2}|\alpha_{Mi},\beta_{Mi}], \\ & \left[\sigma_{yC_{2}i}^{2}|\cdot\right] \propto \prod_{k=1}^{K} [y_{iC_{2}\tau k}|L_{iC_{2}\tau},\sigma_{yC_{2}i}^{2}] [\sigma_{yC_{2}i}^{2}|\alpha_{2i},\beta_{2i}], \\ & \left[\sigma_{yC_{6}i}^{2}|\cdot\right] \propto \prod_{k=1}^{K} [y_{iC_{6}\tau k}|L_{iC_{6}\tau},\sigma_{yC_{6}i}^{2}] [\sigma_{yC_{6}i}^{2}|\alpha_{6i},\beta_{6i}]. \end{split}$$

Step 6. Individual litter value for latent variables for  $L_{iC_2\tau}$  are updated in succession based on its conditional distribution, given by

$$[L_{iC_{2}\tau}|\cdot] \propto \prod_{k=1}^{K} [y_{iC_{2}\tau k}|L_{iC_{2}\tau}, \sigma_{yC_{2}i}^{2}] [L_{iC_{2}\tau}|g(f_{s_{i}}, f_{LIG_{i}}, L_{iC_{M}\tau}), \sigma_{C_{2}}^{2}],$$

followed by litter values in succession for  $L_{i\mathcal{C}_{6}\tau}$  given by

$$[L_{iC_{6}\tau}|\cdot] \propto \prod_{k=1}^{K} [y_{iC_{6}\tau k}|L_{iC_{6}\tau}, \sigma_{yC_{6}i}^{2}] [L_{iC_{6}\tau}|g(f_{DOC}, f_{s_{i}}, L_{iC_{M}\tau}), \sigma_{C_{6}}^{2}].$$

Step 7. After step 6, parameters not individual to litter type (scalars) are then updated. This first includes the scalar for  $f_{DOC}$ , given by the conditional distribution as

$$[f_{DOC}|\cdot] \propto \left[L_{iC_6\tau}|g\left(f_{DOC},f_{s_i},L_{iC_M\tau}\right),\sigma^2_{C_6}\right][f_{DOC}|1,1].$$

Step 8. Measurement and model variance are then updated in succession, given by the conditional distributions

$$\begin{split} & \left[\sigma_{C_{6}}^{2}|\cdot\right] \propto \prod_{i=1}^{I} \left[L_{iC_{6}\tau}|g(f_{DOC}, f_{S_{i}}, L_{iC_{M}\tau}), \sigma_{C_{6}}^{2}\right] \left[\sigma_{C_{6}}^{2}|1, 100\right], \\ & \left[\sigma_{C_{2}}^{2}|\cdot\right] \propto \prod_{i=1}^{I} \left[L_{iC_{2}\tau}|g(f_{S_{i}}, f_{LIG_{i}}, L_{iC_{M}\tau}), \sigma_{C_{2}}^{2}\right] \left[\sigma_{C_{2}}^{2}|1, 100\right], \\ & \left[\sigma_{HWE}^{2}|\cdot\right] \propto \prod_{k=1}^{K} \left[y_{HWE_{ik}}|g(f_{S_{i}}, \sigma_{HWE}^{2})\right] \left[\sigma_{HWE}^{2}|1, 1\right], \\ & \left[\sigma_{DIF}^{2}|\cdot\right] \propto \prod_{k=1}^{K} \left[y_{DIF_{ik}}|g(f_{S_{i}}, \sigma_{DIF}^{2})\right] \left[\sigma_{DIF}^{2}|1, 1\right], \\ & \left[\sigma_{LIG}^{2}|\cdot\right] \propto \prod_{k=1}^{K} \left[y_{LIG_{ik}}|g(f_{LIG_{i}}, \sigma_{LIG}^{2})\right] \left[\sigma_{LIG}^{2}|1, 1\right]. \end{split}$$

Step 9. Repeat steps 1-8 many times.

To simplify the number of parameters being estimated in this analysis, the initial values for the soluble ( $C_1$ ) and lignin ( $C_3$ ) LIDEL model pools were then calculated based on posterior distributions for  $L_{iC_M\tau}$ ,  $f_{s_i}$ ,  $f_{DOC}$ , and  $f_{LIG_i}$ , since these pools did not have additional data to estimate the underlying latent states.

$$z_{iC_{1}\tau} = (L_{iC_{M}\tau} * f_{s_{i}}) - (L_{iC_{M}\tau} * f_{s_{i}} * f_{DOC})$$
$$z_{iC_{3}\tau} = L_{iC_{M}\tau} * f_{LIG_{i}}$$

#### APPENDIX 2

In this section we describe a hierarchical Bayesian analysis to estimate posterior distributions of parameters in the LIDEL model, as well as the posterior distribution of latent variables and variance components of time series measurement data from Soong et al. (2015). The posterior distributions and analyses are based on initial conditions analysis (Appendix 1) priors to initialize model pools when the LIDEL model is run in the analysis described here. The LIDEL model is presented in the main text and will not be repeated here. We define notation used to express the data and parameter models, and then present the assumptions used to specify the data and parameter models. We then describe the steps of the MCMC algorithm used in this analysis.

## A2.1 Definitions

i = litter types (alfalfa, ash, bluestem, oak pine)

k = measurement replicate

L = latent state, or the 'true' underlying condition of the variable

s = index for states in the LIDEL model linked to measured data;  $s = C_M, C_6, C_7$ , where  $C_M$  indicates the total mass of the 3 litter pools plus the microbes and microbial products pools,  $C_6$  indicates the DOC pool, and  $C_7$  indicates the CO<sub>2</sub> pool

 $\tau =$  initial time point, where day = 0

l = index measures of a given s, from 1-n measurements in increments of 1

 $d_l = day$  day at which l - indexed measurement is taken. The time interval between measurements can therefore be calculated using the notation  $d_l - \Delta d_{l-1}$ .

 $\theta$  = vector of the 5 parameters of interest in this analysis: the decay of the soluble ( $k_1$ ), nonsoluble structural ( $k_2$ ), and microbial biomass pools ( $k_4$ ), the generation of microbial products from microbe biomass ( $\beta_3$ ), and the generation of DOC from the decay of microbe biomass ( $\lambda_2$ ).

## A2.2 Data models

All measures of litter C mass were assumed to be normally distributed, with a mean of the latent variable corresponding to the 'true' state of that measurement and variance based on the measurement error associated with the instrument used to take the observation. Therefore:

$$y_{iC_{M}d_{l}k} \sim normal(L_{iC_{M}d_{l}}, \sigma_{y_{iC_{M}d_{l}}}^{2}),$$
$$y_{iC_{6}d_{l}k} \sim normal(L_{iC_{6}d_{l}}, \sigma_{y_{iC_{6}d_{l}}}^{2}), \text{ and}$$
$$y_{iC_{7}d_{l}k} \sim normal(L_{iC_{7}d_{l}}, \sigma_{y_{iC_{7}d_{l}}}^{2}).$$

## A2.3 Parameter models

We assume a normal distribution for all latent variables, with a mean of the LIDEL model prediction at a given measurement time point and variance based on model error for that time point and model state. Therefore:

$$L_{iC_{M}d_{l}} \sim normal(g(\boldsymbol{\theta}, L_{iC_{M}d_{l}-\Delta d_{l-1}}), \sigma_{C_{M}d_{l}}^{2}),$$

$$L_{iC_{6}d_{l}} \sim normal(g(\boldsymbol{\theta}, L_{iC_{6}d_{l}-\Delta d_{l-1}}), \sigma_{C_{6}d_{l}}^{2}), \text{ and }$$

$$L_{iC_{7}d_{l}} \sim normal(g(\boldsymbol{\theta}, L_{iC_{7}d_{l}-\Delta d_{l-1}}), \sigma_{C_{7}d_{l}}^{2}).$$

All five parameters of interest in this analysis are rates (expressed as a fraction day<sup>-1</sup>). Therefore all five parameters were assumed to have beta distributions and were given uninformative priors, where:

 $\theta_i \sim beta(1,1).$ 

Measured and modeled variance terms were assumed to have inverse gamma distributions.

Measurement variances were given informative priors based on known measurement error in the instruments used to collect observations. Model variances were given uninformative priors. Therefore;

 $\sigma^2_{C_sd_l} \sim inverse \ gamma(0.001, 0.001), \text{ and}$  $\sigma^2_{y_{iC_sd_l}} \sim inverse \ gamma(\alpha_{iC_sd_l}, \beta_{iC_sd_l}).$ 

## A2.4 Monte Carlo Markov chain algorithm

The joint posterior distribution for the latent states and parameters described above, given the observations, can be expressed as:

$$\begin{bmatrix} \boldsymbol{\theta}, \boldsymbol{L}, \boldsymbol{\sigma}_{C}^{2}, \boldsymbol{\sigma}_{y}^{2} | \boldsymbol{y} \end{bmatrix} \propto \prod_{i=1}^{I} \prod_{s=1}^{S} \prod_{l=2}^{L} \prod_{k=1}^{K} \left[ y_{isd_{l}k} | L_{isd_{l}}, \sigma_{y_{isd_{l}}}^{2} \right] X \left[ L_{isd_{l}} | \boldsymbol{g} (\boldsymbol{\theta}, L_{isd_{l}-\Delta d_{l-1}}), \sigma_{C_{s}d_{l}}^{2} \right] X \left[ L_{isd_{1}} | \boldsymbol{g} (\boldsymbol{\theta}, L_{is\tau}), \sigma_{C_{s}d_{l}}^{2} \right] X$$
$$X \left[ \sigma_{C_{s}d_{l}}^{2} | 0.001, 0.001 \right] X \left[ \sigma_{y_{icsd_{l}}}^{2} | \alpha_{iC_{s}d_{l}}, \beta_{iC_{s}d_{l}} \right] X \left[ \boldsymbol{\theta}_{i} | 1, 1 \right]$$

We estimated posterior distributions for each latent state and parameter using a random-walk Metropolis Hastings within a Markov Chain-Monte Carlo (MCMC) algorithm. We present the steps of the MCMC algorithm in the following section.

Step 1.  $\theta$ , L,  $\sigma_c^2$ ,  $\sigma_y^2$  parameters and latent variables are each initialized with an estimated starting value.

Step 2. The latent variables for  $C_M$ ,  $C_6$  and  $C_7$  are updated for all measurement time points and model pools across litter types by iterating across the l indices. For  $C_M$ , the two measured time points (day 95 and day 365) of mass remaining in the incubation experiment correspond to the sum of five LIDEL model pools ( $C_1$ - $C_5$ ). However the litter mass ( $C_M$ ) measurements at these time points are not paired with sufficient data to inform how measures of litter mass are fractionated across these five LIDEL model pools. Therefore the day 95 and day 365  $C_M$  measurements can only be informed by the initial model values, where sufficient fractionation data allowed for the posterior distribution of each individual LIDEL model pool to be estimated. The conditional distribution for the  $C_M$  latent variables are therefore given by

$$[L_{iC_Md_l}|\cdot] \propto [L_{iC_Md_l}|g(\boldsymbol{\theta}, L_{is\tau}), \sigma^2_{C_Md_l}] \prod_{k=1}^K [y_{iC_Md_lk}|L_{iC_Md_l}, \sigma^2_{y_{iC_Md_l}}].$$

The latent variables for  $C_6$  and  $C_7$ , however, are informed by measurements that correspond to these individual pools through time. Therefore, except for the initial and final time measurements, the conditional distribution of a given latent variable at a given point in time must consider both the previous and the subsequent time point.

At l = 1, the conditional distribution is given by

$$[L_{isd_1}|\cdot] \propto [L_{isd_l+\Delta d_{l+1}}|g(\boldsymbol{\theta}, L_{isd_1}), \sigma^2_{C_sd_{1+1}}][L_{isd_1}|g(\boldsymbol{\theta}, L_{is\tau}), \sigma^2_{C_sd_1}] \prod_{k=1}^K \left[y_{isd_1k}|L_{isd_1}, \sigma^2_{y_{iC_sd_1}}\right]$$

When l = 1 is between 2 and L, the conditional distribution is given by

$$\begin{split} & \left[L_{isd_{l}}|\cdot\right] \propto \left[L_{isd_{l}}|g(\boldsymbol{\theta},L_{isd_{l}-\Delta d_{l-1}}),\sigma_{C_{s}d_{l}}^{2}\right]\left[L_{isd_{l}+\Delta d_{l+1}}|g(\boldsymbol{\theta},L_{isd_{1}}),\sigma_{C_{s}d_{1+1}}^{2}\right] \\ & \prod_{k=1}^{K} \left[y_{isd_{l}k}|L_{isd_{l}},\sigma_{y_{i}C_{s}d_{l}}^{2}\right] \end{split}$$

When l = L the conditional distribution is given by

...

$$[L_{isd_L}|\cdot] \propto [L_{isd_L}|g(\boldsymbol{\theta}, L_{isd_l-\Delta d_{l-1}}), \sigma_{C_sd_L}^2] \prod_{k=1}^K [y_{isd_Lk}|L_{isd_L}, \sigma_{y_{iC_sd_L}}^2].$$

Step 3. The each of the five LIDEL parameters is then updated sequentially, each based on the conditional distribution given by

$$\begin{split} & [\theta_i|\cdot] \propto \left[L_{isd_L}|g(\boldsymbol{\theta}, L_{isd_l-\Delta d_{l-1}}), \sigma_{C_sd_L}^2\right] \prod_{i=1}^{I} \prod_{s=1}^{S} \prod_{l=2}^{L} \left[L_{isd_l}|g(\boldsymbol{\theta}, L_{isd_l-\Delta d_{l-1}}), \sigma_{C_sd_l}^2\right] \\ & \prod_{i=1}^{I} \prod_{s=1}^{S} \left[L_{isd_1}|g(\boldsymbol{\theta}, L_{is\tau}), \sigma_{C_sd_1}^2\right] [\theta_i|1,1]. \end{split}$$

Step 4. Measurement and model variance are then updated in succession, given by the conditional distributions

$$\left[\sigma_{y_{iC_{s}d_{l}}}^{2}|\cdot\right] \propto \prod_{k=1}^{K} \left[y_{isd_{l}k}|L_{isd_{l}},\sigma_{y_{iC_{s}d_{l}}}^{2}\right] \left[\sigma_{y_{iC_{s}d_{l}}}^{2}|\alpha_{iC_{s}d_{l}},\beta_{iC_{s}d_{l}}\right]$$

$$\left[\sigma_{C_{s}d_{l}}^{2}|\cdot\right] \propto \prod_{i=1}^{l} \left[L_{isd_{l}}|g(\boldsymbol{\theta}, L_{isd_{l}-\Delta d_{l-1}}), \sigma_{C_{s}d_{l}}^{2}\right] \left[\sigma_{C_{s}d_{l}}^{2}|0.001, 0.001\right].$$

Step 5. Repeat steps 1-4 many times.